

## UNIVERSITÀ DI SIENA 1240

Department of Physical Sciences, Earth and Environment

#### PhD in Experimental Physics

XXXV Cycle

#### Coordinator: Prof. Riccardo Paoletti

## Probing star clusters as cosmic ray factories

Disciplinary Scientific Sector: FIS/05

A thesis submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy

PhD Student Stefano Menchiari University of Siena Via Roma 56

Signature

Stepar Membin.

Supervisor Dott. Elena Amato INAF – Osservatorio Astrofisico di Arcetri

Signature

Pleus Amaro

## \_CONTENTS

i

### Introduction

Young Massive Star Clusters as very-high-energy sources			1	
1.1	What is a stellar cluster?			
1.2	volutionary path of a young massive star cluster	3		
	1.2.1	The birth of a stellar cluster	3	
	1.2.2	Emerging from the cradle: creation of a wind bubble	5	
	1.2.3	Evolution of the wind bubble	6	
1.3	ic Rays and the enigma of their origins: Young Massive Star Clusters			
	as Cos	mic Ray factories?	15	
	1.3.1	Acceleration mechanisms in the core of a YMSCs	21	
	1.3.2	Acceleration through efficient scattering with turbulence	25	
	1.3.3	Acceleration at the cluster wind termination shock	28	
1.4	Probir	ng YMSCs as CRs accelerators	33	
The scientific case of Cygnus OB2				
2.1	The yo	oung massive star cluster Cygnus OB2	37	
	2.1.1	Cygnus OB2 wind luminosity and mass loss rate	38	
	2.1.2	The wind blown bubble of Cygnus OB2	42	
2.2 The distribution of CRs in Cygnus OB2		stribution of CRs in Cygnus OB2	45	
2.3	2.3 Modeling $\gamma$ -ray emission from Cygnus OB2			
	2.3.1	The interstellar medium close to Cygnus OB2	49	
2.4	Comparison between expected and observed $\gamma\text{-ray emission}$		53	
	2.4.1	Kolmogorov case	54	
	2.4.2	Kraichnan case	55	
	2.4.3	Bohm case	57	
	Your 1.1 1.2 1.3 1.4 The 2.1 2.2 2.3 2.4	Yourse Mass         1.1       What         1.2       The exercised         1.2.1       1.2.1         1.2.2       1.2.3         1.3       Cosmit         as Cos       1.3.1         1.3       Cosmit         as Cos       1.3.1         1.3.2       1.3.3         1.4       Probin         The scienti       2.1.1         2.1       The you         2.1.2       2.1.1         2.1.2       The di         2.3       Model         2.3.1       2.4.1         2.4.2       2.4.3	Young Massive Star Clusters as very-high-energy sources1.1What is a stellar cluster?1.2The evolutionary path of a young massive star cluster1.2.1The birth of a stellar cluster1.2.2Emerging from the cradle: creation of a wind bubble1.2.3Evolution of the wind bubble1.2.3Evolution of the enigma of their origins: Young Massive Star Clustersas Cosmic Rays and the enigma of their origins: Young Massive Star Clustersas Cosmic Ray factories?1.3.1Acceleration mechanisms in the core of a YMSCs1.3.2Acceleration through efficient scattering with turbulence1.3.3Acceleration at the cluster wind termination shock1.4Probing YMSCs as CRs accelerators2.1The young massive star cluster Cygnus OB22.1.1Cygnus OB2 wind luminosity and mass loss rate2.1.2The wind blown bubble of Cygnus OB22.3Modeling $\gamma$ -ray emission from Cygnus OB22.4Comparison between expected and observed $\gamma$ -ray emission2.4.1Kolmogorov case2.4.2Kraichnan case2.4.3Bohm case	

	2.5	Discussion			
		2.5.1 Discussion on the spectral analysis results	59		
		2.5.2 Discussion on the morphological analysis results	61		
	2.6	6 Cygnus OB2 as a cosmic ray accelerator?			
3	3 The contribution of YMSCs to the diffuse $\gamma$ -ray emission				
	3.1	Modeling the star population inside YMSCs	68		
		3.1.1 Mass distribution of stars inside clusters	68		
		3.1.2 Modeling stellar parameters	70		
		3.1.3 Modeling stellar winds	75		
	3.2	Generating a synthetic population of YMSCs	77		
		3.2.1 Cluster age and mass distribution	77		
		3.2.2 Cluster spatial distribution	81		
		3.2.3 Wind luminosity of star clusters	87		
		3.2.4 Statistical properties of the YMSC synthetic population	87		
	3.3	3 Cosmic Ray distribution in YMSCs			
	3.4	$\gamma$ -ray emission from single sources	97		
	3.5	Diffuse emission and comparison with data	98		
	3.6	Final remarks			
		3.6.1 Future prospects	108		
4	4 Ionization of molecular clouds close to YMSCs		110		
	4.1	Penetration of low-energy CRs in a molecular cloud	111		
	4.2	Ionization of molecular clouds embedded in the wind-blown bubble of a			
		YMSC	114		
	4.3	Comparison with reality: The case of DR21	118		
	4.4	Preliminary conclusions and future prospectives	121		
5	Con	clusions	124		
Ap	Appendix 13				
	A.1	$\gamma$ -ray emission as a tracer of CRs $\ldots$	131		
	A.2	Ionization of molecular clouds by low energy CRs	134		

### INTRODUCTION

TELLAR clusters are among the most studied celestial objects in the cosmos. They represent crucial laboratories for understanding a wide range of problems in different subtopics of astrophysics. As they include a vast number of stars with masses spanning a broad interval, star clusters are, for instance, essential in the studies of the stellar Initial Mass Function (Bastian et al., 2010a). In general, all cluster members are born simultaneously and share common primordial properties since they are the offspring of the same progenitor molecular cloud. This makes star clusters an excellent subject for studies of stellar evolution (Kalirai & Richer, 2010). Some stellar clusters are also gravitationally bound systems, i. e. are held together by the mutual gravitational attraction of their members. From this point of view, they are the perfect target for studies on stellar dynamics (Vesperini, 2010).

Stellar clusters are fundamental building blocks of galaxies. The analysis of the spatial distribution in the host galaxy of different types of clusters has often been used to determine the galactic structure. One example is the distribution of globular clusters in the Milky Way, which has allowed the estimation of the Milky Way size (Shapley, 1918), the determination of the Galactic Center, and has established the existence of the galactic halo (Bica et al., 2006). Or similarly, the distribution of young clusters in external galaxies, which is employed to trace star forming regions and the spiral structure in galactic disks (Adamo et al., 2020).

Stellar clusters are also vital for understanding the star formation mechanism. Indeed, a relevant fraction of stars is born in clustered environments, as demonstrated by Lada & Lada (2003), who showed that the contribution to the local star formation rate from the population of young embedded clusters is similar to that obtained from field stars. This is even more true for massive stars, as almost ~70% of O-type stars are observed in clusters or associations (Gies, 1987; Parker & Goodwin, 2007), and at least ~50% of the remaining stars are identified as runaways (de Wit et al., 2005). More accurate estimates seem to lower the percentage of massive stars born outside a cluster to ~4% (de Wit et al., 2005).

Last but not least, stellar clusters play a prominent role in high-energy astrophysics. For a long time, the interest in star clusters by the high-energy astrophysics community was mainly due to the unusual abundance of extreme objects, often referred to as lusus naturæ<sup>1</sup> or stellar exotica. Indeed, globular clusters, because of their old age and relatively homogeneous population, and isolation from their parent galaxies, are known to be rich in unusual extreme objects, such as X-rays binaries (Heinke, 2010), pulsars (Ransom, 2008), black holes (Maccarone et al., 2007), etc. However, in recent decades, star clusters, or to be more accurate, young massive star clusters (YMSC), have found themselves at the center of attention not so much as a possible cradle for a future generation of lusis naturæ, but rather as objects capable of producing and accelerating cosmic rays (CRs). Massive OB-type stars are known to launch powerful winds (Abbott, 1979; Cassinelli, 1979; Kudritzki & Puls, 2000), and the presence of tens (if not hundreds or, in the most extreme and rare cases, thousands) of massive stars crammed into a small volume can generate favorable conditions for boosting particles to very high-energies. Several CR production mechanisms have been proposed through the years, such as for example, acceleration at the wind termination shock of single massive stars (Casse & Paul, 1980; Cesarsky & Montmerle, 1983) or acceleration by wind-wind interaction (Klepach et al., 2000; Reimer et al., 2006). In the case of compact clusters, the winds from individual stars may end up combining, somehow creating a collective cluster wind. In this scenario, particle acceleration may occur at the cluster wind termination shock (Morlino et al., 2021). As the shock generated by the winds from the stars or by the collective cluster wind interacts with density inhomogeneities of different scales, broad spectra of magnetohydrodynamic fluctuations are generated. In these systems then, secondorder Fermi acceleration within the turbulent plasma becomes also a possible efficient mechanism of particle acceleration (Bykov et al., 2020). Finally, in the case of YMSCs older than  $\sim 5 - 10$  Myr, the most massive stars gradually leave the main sequence and rapidly move to their final evolutionary stage. At this point, those stars begin to explode as supernovae, leaving behind supernova remnants (SNRs). In such aged systems, CR acceleration is achieved thanks to the multiple interactions between SNR shocks and stellar winds (Bykov & Toptygin, 2001), with a gradual decrease in time of the contribution to the process from the stars' winds (Vieu et al., 2022). The possibility that YMSCs can actually accelerate particles is a fact of extreme relevance in the general panorama of high-energy astrophysics, especially in connection with the problem of the origin of CRs. At present, it is well proven, based both on theoretical arguments (Blandford & Eichler, 1987; Berezhko & Krymskii, 1988) and empirical observations (Koyama et al., 1995; Reynolds, 2008; Helder et al., 2012), that SNRs are a class of sources able to produce CRs. Nevertheless, the population of SNRs alone has difficulties accounting for

<sup>&</sup>lt;sup>1</sup>lusus naturæ is Latin for freaks of nature, mutants, or monsters

all the observed properties of galactic CRs. More precisely, a single population of SNRs struggles to reproduce two distinct things: the observed CR composition at Earth and the so-called *knee feature* in the CR spectrum. Measurements of CR composition seem to point to an excess in the ratio of some isotopes, such as for example, the <sup>22</sup>Ne to <sup>20</sup>Ne ratio, which is found  $5.3 \pm 0.3$  times higher than the solar value (Wiedenbeck & Greiner, 1981; Binns et al., 2008). YMSCs can easily explain this excess, as these anomalous ratios can result from the winds of Wolf-Rayet and massive stars (Gupta et al., 2020).

On a very general ground, the observed CR spectrum at the Earth between  $\sim 10^{12}$ - $-10^{18}$  eV can be adequately described by a broken power law (Workman et al., 2022). The change in the slope, referred to, in the literature, as the *knee*, appears to be located at  $\sim 3 \times 10^{15}$  eV and has been interpreted as the maximum energy reachable by the population of galactic CR accelerators (Blasi, 2013; Amato, 2014). Celestial objects able to produce particles up to these energies are commonly called *PeVatrons*. From both the theoretical and observational (Aharonian et al., 2019) point of view, SNRs struggle to accelerate CRs up to such energies, unless under specific conditions involving extreme energy releases, at least assuming that the needed magnetic field amplification is well understood (Bell et al., 2013; Cardillo et al., 2015). On the contrary, YMSCs seem to be promising PeVatron candidates, as suggested by theoretical considerations (Morlino et al., 2021) and tentative observational hints (Cao et al., 2021). The answer to the century-old enigma about the origin of galactic CRs could therefore lie in having multiple populations of galactic accelerators: SNRs could account for the bulk of the observed CRs, while particle acceleration associated with the powerful winds of YMSCs could provide the highest-energy particles. This would in parallel explain the unusual abundances of some elements. But how to confirm or eventually reject this scenario? It is widely known that it is impossible to probe the properties of a given galactic accelerator by observations of CRs at Earth alone. In fact, the propagation of charged particles below a few  $10^{18}$  eV in the Galaxy is diffusive because of the scattering with the interstellar magnetic field fluctuations (Strong et al., 2007; Amato & Blasi, 2018). The information on the position of the CR sources is totally lost during the propagation process. For this reason, the search for CR sources must rely on somewhat indirect investigation methods based on the interaction between CRs and the interstellar matter (ISM) and radiation field. In practice, one can exploit observations of non-thermal radiation: in this sense  $\gamma$ -ray emission has a privileged role as it directly traces the presence of highenergy and very-high-energy particles (Tibaldo et al., 2021). Another promising strategy is to search for indications of a high ionization rate induced by energetic particles in dense clouds close to the accelerator (Gabici & Montmerle, 2015). In summary, in order to assess whether YMSCs are eventually playing a primary role in the origin of galactic CRs,  $\gamma$ -ray observations and comprehensive studies of the environment close to these objects are of extreme importance.

In this thesis, we investigate the observational properties of YMSCs as very-highenergy sources under the assumption that CR production occurs at the cluster wind termination shock. In particular we will assume CR acceleration and propagation as described by the model of Morlino et al. (2021) and derive different observational signatures that can help constrain the model and test the efficiency of clusters as CR accelerators. The entire work is divided into three distinct parts.

First, we will try to assess whether and to what extent the employed model of particle acceleration can effectively reproduce the observed  $\gamma$ -ray emission from a given YMSC. To this purpose, we will focus on the scientific case of Cygnus OB2. This is one of the most iconic cases when discussing YMSCs as CR sources since several experiments have detected diffuse  $\gamma$ -ray emission in both the high-energy (Ackermann et al., 2011a) and very-high-energy (Bartoli et al., 2014a; Abeysekara et al., 2021a) bands towards its position. Moreover, the detection of a 1.4 PeV photon from the same region (Cao et al., 2021) makes it one of the most promising PeVatron candidates in the Galaxy.

In the second part of the thesis, we will rather focus on YMSCs as a population of  $\gamma$ -ray sources. If YMSCs are indeed particle accelerators, then it is natural to think of them also as  $\gamma$ -ray emitters. In view of the new and upcoming facilities for  $\gamma$ -ray observations such as the Cherenkov Telescope Array (Cherenkov Telescope Array Consortium et al., 2019), the Astri Mini Array (D'Aì et al., 2022; Vercellone et al., 2022), and the Southern Wide-field Gamma-ray Observatory (Bakalová, 2022), simulating the emission from a synthetic population of galactic YMSCs becomes of fundamental importance. In fact, the comparison between the number of expected versus detected YMSCs can be used to discriminate the capabilities of these sources as particle accelerators and their contribution to the galactic CR sea. Moreover, as we will see, the  $\gamma$ -ray emission from most of these objects is predicted to be extended and potentially difficult to disentangle from the diffuse. Consequently, YMSCs could also significantly contribute to the galactic diffuse  $\gamma$ -ray emission, and the estimation of this contribution is of particular importance for all those studies that require a solid modelization of the galactic Center.

In the third and final part of the work, we will focus on the low- energy CRs and their impact on the ISM surrounding the stellar clusters. In addition to highly-energetic particles, YMSCs are expected to produce also a (largely more conspicuous) population of CRs with energies below 1 GeV. Differently from the cluster starlight, these particles are capable of penetrating deep in the core of the dense molecular clouds that are commonly found in the neighborhood of YMSCs, ionizing the cold neutral medium of which the clouds are made. The general aim will then be to evaluate the ionization rate of a molecular cloud located close to a YMSC. This type of investigation is of particular interest for two main reasons. First, the measurement of the ionization rate can be employed as a test to trace the presence of freshly accelerated CRs. This information can be subsequently combined with independent  $\gamma$ -ray observations to have a self-consistent picture of the CR distribution around the stellar cluster. Secondly, depending on how much the ionization rate is different from the Spitzer value (Spitzer & Tomasko, 1968), low-energy CRs could be an additional feedback channel for YMSCs to regulate the star formation process in their environment.

The following manuscript is divided into five main chapters. In the first chapter, we give a general review of YMSCs and their main properties, followed by a discussion of their role as particle accelerators. We will review the main acceleration models in stellar clusters and the indirect techniques to observe CRs at the sources. The bulk of the work is then presented in chapters two, three, and four, where we analyze the scientific case of Cygnus OB2, the emission from a synthetic population of YMSCs, and the estimate of the ionization rate in clouds close to a stellar cluster. Finally in chapter five, we summarize and discuss the conclusions of the work.

## CHAPTER 1

## YOUNG MASSIVE STAR CLUSTERS AS VERY-HIGH-ENERGY SOURCES

s primary players in different fields of astrophysics, from stellar physics to physics and astronomy of star formation, and recently also in the highenergy astrophysics field, the properties of stellar clusters have been widely investigated. In this chapter, we provide an overview of these objects, starting from their classification based on fundamental characteristics, and then briefly reviewing their evolutionary path. Afterward, we describe their capabilities in shaping the ISM around them. We then concentrate on stellar clusters as CR accelerators, giving a comprehensive summary of the most widely considered acceleration mechanism in these systems.

#### **1.1** What is a stellar cluster?

When it comes to defining a star cluster, several criteria can be given (Krumholz et al., 2019), and unfortunately, there is no single definition that can be used universally. It follows that the definition of a star cluster has a certain degree of arbitrariness, and it may change from author to author, directly marking their research case. Among the several proposed formulae, one of the most widespread is the one given by Lada & Lada (2003), who define a stellar cluster as a group of stars with a mass density large enough ( $\rho_{\star} \gtrsim 1 M_{\odot} pc^{-3}$ ) to withstand tidal disruption in Solar Neighborhood conditions and, in parallel, to have enough members to avoid kinematical evaporation for at least 100 Myr. Another possible criterion is the one proposed by Portegies Zwart et al. (2010), who define a star cluster as a group of stars with density  $\rho_{\star}$  and size r is considered

to be gravitationally bound if the velocity dispersion  $\sigma$  is such that  $\sigma \leq r \sqrt{G\rho_{\star}}$ . The prescription suggested by Lada & Lada (2003) is somewhat more general as it also includes unbound clusters, and as far as the present work is concerned, it is perhaps the most suitable. In fact, let us now think for a moment in terms of the involved timescales. As we will justify in the next sections, for the aim of this thesis, we are interested in those clusters that are young enough so that the pollution by supernova (SN) explosions is relatively low. This means that we are interested in clusters younger than  ${\sim}10$  Myr, a timescale that is close to the lifetime of a star with mass 20  $M_{\odot}$ . As the number of stars more massive than 20  $M_{\odot}$  is relatively low<sup>1</sup> on average, clusters younger than ~10 Myr are expected to have witnessed few SN explosions. The time required for an unbound system to disperse is the crossing time, defined as  $t_{cr} = r/\sigma \sim 1/\sqrt{G\rho_{\star}}$ . Considering the threshold in  $\rho_{\star}$  set by Lada & Lada (2003), the crossing time is  $t_{cr} \sim 10 Myr$ . It follows that unbound clusters must also be considered, as they survive long enough to potentially contribute to CR acceleration. In addition to the threshold in the stellar density, the definition of Lada & Lada (2003) includes also a condition on the minimum number of stars in the cluster, due to the requirement of having a kinematic evaporation time less than 100 Myr. For a cluster with N members, the relaxation time is  $\tau_{relax} \approx (0.1N/\ln N)t_{cr}$  (Lada & Lada, 2003), and the evaporation can be calculated as  $\tau_{ev} \approx 100 \tau_{relax}$ . Again, even for clusters with a relatively small number of members, the evaporation timescale is well above our threshold of 10 Myr. To sum up, we can safely use Lada & Lada (2003) definition of star clusters, with the clarification that, during our work, we will focus specifically on the subcategory of young clusters (age less than 10 Myr).

Yet, youth is not the only parameter we require. As the presence of massive stars is a fundamental ingredient for CR acceleration, we also need to consider only those clusters with a significant number of massive stars. Unfortunately, it is not straightforward to formally express this condition. Once the stars initial mass function is known (Salpeter, 1955; Kroupa, 2001), the number of massive stars clearly depends on the cluster mass. In this way, the condition can be shifted to a lower limit cut on the star cluster mass. Let us now, for the moment, fix this threshold to  $10^3 M_{\odot}$ , we will then justify a posteriori in § 3.2.4 that this limit is reasonable.

 $<sup>^{1}</sup>$ This statement is somewhat tricky, as the population of stars is potentially correlated with the mass of the stellar cluster, see § 3.1.1

## **1.2** The evolutionary path of a young massive star cluster

The evolutionary path of a stellar cluster can change drastically from one case to another. Clusters that are not gravitationally bound are expected to disperse in a few crossing times. Even for bound clusters, their long-term fate is not straightforward, as the complex inner stellar dynamics may induce kinematic evaporation on timescales of a few hundred Myr. Despite this, the early evolutionary stages of the various types of star clusters should, on average, be quite similar. On a very general ground, we can divide the initial phases of stellar clusters evolution into two main parts: a first initial stage, where the cluster members are forming as a result of the collapse of dense gas clumps in giant molecular clouds, and a second phase, when the first stars light up, generating stellar feedback on the surrounding gas and stopping the star formation process. In the case of massive star clusters, during the second stage, the presence of a significant number of massive stars with their powerful winds can produce large bubbles filled with hot shocked wind material. In the next two subsections we will describe separately these two evolutionary stages.

#### 1.2.1 The birth of a stellar cluster

Star clusters are known to form within massive large complexes of cold molecular gas (Fig. 1.1a), usually called giant molecular clouds (GMCs). The gas within a GMC is not uniformly distributed. On the contrary, GMCs are characterized by having a self-similar structure over a wide range of scales down to individual protostellar cores, which are aggregated into cluster-forming dense gas clumps (Williams et al., 2000; McKee & Ostriker, 2007) with typical sizes of 0.1 - 2 pc and masses ranging from a few solar masses up to thousands of solar masses (Lada & Lada, 2003). On average, less than  $\sim 10\%$  of the volume and mass of a GMC end up in the form of dense gas (>  $10 \text{ cm}^{-3}$ ). The clumps where clusters are born are highly localized and occupy a small fraction (a few percent) of the volume of a GMC. The mass distribution of these cluster-forming clumps is well described by a power law with index  $\sim -1.7$  (Kramer et al., 1998). Interestingly, with such index, it directly follows that most of the mass of the dense gas component of a GMC is found in its most massive cores. Consequently, as the star formation process is triggered only in dense gas regions, it is not surprising that a significant fraction of stars are born in a clustered environment, coherently with the mass distribution of the GMC clumps.

As not the whole GMC is in form of dense gas, it follows that not the entire mass of a GMC ( $M_{gas}$ ) ends up into stars ( $M_{stars}$ ). The parameter describing the fraction of gas converted into stars is the star formation efficiency (SFE), defined as  $\epsilon_{SFE} = M_{stars}/(M_{gas} + M_{stars})$ , which is  $\epsilon_{SFE} = 1$  if all the gas ends up in the form of stars. For the entire GMC, the  $\epsilon_{SFE}$  is usually of the order of a few percent (Duerr et al., 1982). The value increases up to  $\sim 30\%$  when considering single cluster-forming clumps (Lada & Lada, 2003). The implied low-efficiency at small scales indicates that, at some point, there is something preventing the process of star formation from proceeding (Krumholz et al., 2019). This is thought to be related mainly to feedback processes associated with the lighting-up of the first stars<sup>2</sup>, although some feedback may also arise from the outflows of low-mass protostellar objects (Bally, 2016). As the stars enter the main sequence, stellar feedback processes sweep away the gas, causing the cluster to emerge from its cradle and begin the next evolutionary stage.

Before proceeding further, it is interesting to note the following: since the SFE is generally low, a significant fraction of the mass is still in the form of dense gas, which will be swept away when stellar feedback mechanisms kick in. So from the point of view of stellar clusters, the gas expulsion can potentially be a destructive process<sup>3</sup>, as many systems that were once gravitationally bound may find themselves deprived of most of their mass.

Whether a cluster may survive or not from its birth as a bound system can be assessed by comparing its dynamical crossing time with the timescale of gas expulsion  $(\tau_{ge})$ , which ultimately depends on the dominant feedback process. If  $\tau_{ge} \ll t_{cr}$ , which seems to be the case for massive star clusters, since OB-type massive stars produce the strongest feedback (Krumholz et al., 2014), the cluster should disperse unless the SFE is not at least as high as ~ 50% (Wilking & Lada, 1983). However, this seems to be in tension with observations.

The formation of clusters and the issue of infant mortality is still not fully understood. The process is still under debate (see Lada & Lada (2003) and references therein for a comprehensive review of the problem). Nevertheless, as stated in § 1.1, assuming that unbound systems can last at least a few crossing times, they live long enough to possibly contribute as CRs accelerators.

<sup>&</sup>lt;sup>2</sup>For the sake of completeness, it must be mentioned that a minor contribution to the quenching of the SFE can also be caused by the dynamical properties of the molecular clumps, i.e., if they are not bound (Dobbs et al., 2011) or are highly turbulent (Krumholz & McKee, 2005).

<sup>&</sup>lt;sup>3</sup>As some sort of cosmic joke, the death of unbound star clusters has been proposed as a mechanism for the birth of the loose aggregations of massive stars known as OB associations. However, this picture has lost relevance over the years due to the growing body of opposing evidence (see Wright et al. (2022) and references therein for a well made review on the topic)



Figure 1.1: Three different celestial objects representing different evolutionary stages of a massive stellar cluster. (a) Herschel observations of the giant molecular cloud DR21 localized in the Cygnus-X star-forming complex. (b) Hubble observations of the embedded cluster RCW 38. The star cluster itself is invisible at the optical wavelength as it is still deep within its parental cloud. (c) JWST observation of the YMSC 30 Doradus emerging from its GMC as it blows away the surrounding gas.

#### 1.2.2 Emerging from the cradle: creation of a wind bubble

As soon as stellar feedback kicks in, quickly bringing the star formation process to a halt, the cluster is formally born. At the very beginning, the cluster is still embedded in the dense gas of its parental clump, buried deep within the GMC. Objects in this evolutionary phase are often called embedded clusters (see Fig. 1.1b). The embedded phase is usually short-lived as the feedback from the stars rapidly blows away the dense gaseous envelope.

Stellar feedback may come in different flavors. Massive stars are known to produce a significant flux of ionizing photons. The presence of ionizing radiation directly heats the surrounding gas up to  $10^4$  K. If not trapped, the heated gas will flow out of the cluster in a breeze called *champagne flow*. This process can effectively remove a certain amount of gas (Williams & McKee, 1997). In parallel, the light from the same zero-age population can also produce significant direct radiation pressure on the surrounding gas. The emitted power (the estimated luminosity to mass ratio is  $\sim 1100 \text{ L}_{\odot} \text{ M}_{\odot}^{-1}$ ) is mostly in the UV band, where the ISM is highly opaque. Depending on the density, this mechanism is one of the most efficient gas removal processes (Fall et al., 2010). In the case of a very thick envelope, the indirect radiation pressure provided by multiple cycles of absorption and re-emission from the dust may also play an important role (Thompson et al., 2015).

Last but not least comes the feedback from the powerful winds of massive OB-type stars. Hot stars with surface temperatures higher than a few  $10^4$  K are known to blow fast winds with speeds up to several thousand km s<sup>-1</sup> (Kudritzki & Puls, 2000). The energy injected through this channel can be considerably high, and slightly less or of the

order of supernova explosions (Krause et al., 2013). As the stellar wind impacts the surrounding gas, the wind material gets shocked and can reach temperatures as high as  $10^7$  K, entering a regime where radiation cooling becomes practically inefficient. This leads to the formation of hot, expanding bubbles (Castor et al., 1975; Weaver et al., 1977) that push away all the surrounding cold gas putting formally an end to the embedded phase (see Fig. 1.1c). The physics of these bubble-like structures has been comprehensively studied by Weaver et al. (1977), who have developed a simple analytical model for their evolution. Although the model by Weaver et al. (1977) was originally intended for bubbles generated by isolated massive stars, it can be fairly well applied to the case of YMSCs if the cluster is compact enough (we will better state this condition soon) so that the winds from the central stars may combine together to form a collective cluster wind. Understanding these structures is of vital importance for the scope of our work. In the following, we will review the evolution of YMSC bubbles following the work of Weaver et al. (1977).

#### 1.2.3 Evolution of the wind bubble

In a general, ideal, situation, massive stars within the clusters start to blow fast winds as soon as they enter the main sequence. The winds from the individual stars combine to generate a spherically symmetric cluster wind with a speed  $v_w$ . As a consequence of this wind, the YMSC loses mass at a rate  $\dot{M}$ . The mechanical wind luminosity  $L_w$  is:

$$L_w = \frac{1}{2} \dot{M} v_w^2 \,. \tag{1.1}$$

The collision of the supersonic, cold, cluster wind (region 1 in Fig. 1.2) with the surrounding ism generates a shock that propagates in the latter (*forward shock*) and a reverse shock that propagates backward towards the origin of the wind. This reverse shock becomes the wind *termination shock* (TS). The wind material is slowed down and heated up when crossing it, and a bubble of hot gas is formed (region 2 in Fig. 1.2). A contact discontinuity separates the bubble of shocked wind material from the shocked ISM that piles up on top of it (region 3 in Fig. 1.2).

The dynamical evolution of a wind bubble can be divided into: early, intermediate and late stage, which differ primarily for the importance of losses. The early phase is characterized by a fast adiabatic expansion of the bubble, during which radiative losses do not affect the dynamics of any region. This is not true anymore during the intermediate stage, when radiative losses cause the contraction of the shell of shocked ISM, while the region filled with the hot shocked wind material continues to expand adiabatically. Finally, in the late phase, non adiabatic losses begin to influence the whole hot cavity.



Figure 1.2: Sketch of the bubble structure generated by a YMSC.

Let us now summarize one by one each evolutionary stage.

#### Early stage of a wind bubble

The early stage of evolution is expected to last a short time (a few 10<sup>4</sup> yr), so, in general, it is not particularly relevant in practical terms. Nevertheless, let us now proceed in evaluating the system evolution. The expression of the FS position as a function of time can be easily found through a dimensional analysis, and it is:

$$R_{FS}(t) = A L_w^{1/5} \rho_0^{-1/5} t^{3/5}$$
(1.2)

where A is a dimensionless constant to be determined. To obtain the evolution of the size of the contact discontinuity  $(R_c)$ , one needs to numerically integrate the equations of continuity of mass and momentum for the warm shocked shell of ISM (region 3) under the assumption of adiabatic expansion. The position of  $R_c$  is then set where the shell density profile drops to zero, that is  $R_c \simeq 0.86R_{FS}(t)$ . Obtaining the evolution of the cold wind region boundary  $(R_{TS})$  is not straightforward, as while the outer shell has a self similar evolution, this is not true for the region filled with hot shocked wind material (region 2). One possible way to obtain an analytical expression for  $R_{TS}$  is to consider region 2 as almost isobaric. Under this assumption, which is indeed a good approximation

(Weaver et al., 1977), it is possible to derive from the continuity equation and adiabatic law the profiles of the velocity, density and pressure in region 2. The normalizations of these profiels are obtained using the Rankine-Hugoniot relations for shocks (assuming to known the wind speed). By imposing the regularity conditions at the contact discontinuity, one finds:

$$R_{TS}(t) = 0.90A^{3/2} \dot{M}^{3/10} \rho_0^{-3/10} v_w^{1/10} t^{2/5}$$
(1.3)

with A = 0.88. The early evolutionary stage lasts until the system age becomes comparable to the cooling timescale of the outer shell, at which time the adiabatic approximation cannot be used. The time at which this occurs is (Falle, 1975):

$$\tau_{\text{shell collapse}} = 2.33 \times 10^3 \left(\frac{\dot{M}}{M_{\odot} y r^{-1}}\right)^{0.32} \left(\frac{v_w^2}{k m^2 s^{-2}}\right)^{0.32} \left(\frac{n_0}{c m^{-3}}\right)^{-0.68} yr \tag{1.4}$$

which for  $n_0 = 1 \ cm^{-3}$  and in the case of an extreme YMSC with  $\dot{M} = 10^{-4} \ M_{\odot} yr^{-1}$  and  $v_w = 3000 \ kms^{-1}$  is  $\tau_{\text{shell collapse}} \simeq 2 \times 10^4 \text{ yr.}$ 

#### Intermediate stage of a wind bubble

As the outer shell cools, it contracts to form a thin, cold and dense, isobaric layer. Consequently, in this phase,  $R_{FS} \simeq R_c$ , and it is then reasonable to assume that the position of the shell marks the total extent of the bubble, that we shall name as  $R_b \equiv R_{FS} \simeq R_c$ . To obtain the evolution of the bubble size (the former forward shock), we can consider the momentum equation for the outer shell:

$$\frac{d}{dt} \left[ M_s \frac{dR_b}{dt} \right] = 4\pi R_{FS}^2 P_2 \tag{1.5}$$

where  $P_2$  is the pressure in the hot bubble (region 2) and  $M_s$  is the mass contained in the swept-up shell, which is  $M_s = (4/3)\pi\rho_0 R_b^3$ . Eq. (1.5) tells us that the momentum of the outer shell changes due to the work done by the expansion of the hot shocked wind material. This induces a variation of the energy of the hot gas equal to:

$$\frac{dE_2}{dt} = L_w - 4\pi R_b^2 P_2 \frac{dR_b}{dt}$$
(1.6)

with  $E_2$  as the energy of the hot gas, defined as:

$$E_2 = \frac{4}{3}\pi R_b^3 \frac{P}{\gamma_g - 1} = 2\pi R_b^3 P \tag{1.7}$$

where in the last equality we have assumed the case of a monoatomic gas ( $\gamma_g = 5/3$ ). Finally, assuming that, similarly to the earlier stage, the forward shock radius can be parametrized as  $R_b \propto t^{\alpha}$ , combining Eq. (1.5–1.7) leads to:

$$R_b(t) = \left(\frac{250}{308\pi}\right)^{1/5} L_w^{1/5} \rho_0^{-1/5} t^{3/5}$$
(1.8)

$$P_2(t) = \frac{7}{(3850\pi)^{2/5}} L_w^{2/5} \rho_0^{3/5} t^{-4/5} \,. \tag{1.9}$$

Interestingly, the expression of Eq. 1.8 is equal to that of Eq. 1.2, but with A = 0.76, as a consequence of the contraction of the outer shell. Eq. 1.9 can be used to obtain an estimation of the TS position during this intermediate stage, which can be calculated by imposing the balance between the ram pressure of the wind and  $P_2$ :

$$\frac{Mv_w}{4\pi R_{TS}^2} = \frac{7}{(3850\pi)^{2/5}} L_w^{2/5} \rho_0^{3/5} t^{-4/5} , \qquad (1.10)$$

which returns:

$$R_{TS}(t) = \sqrt{\frac{(3850\pi)^{2/5}}{28\pi}} \dot{M}^{1/2} v_w^{1/2} L_w^{-1/5} \rho_0^{-3/10} t^{2/5} \,. \tag{1.11}$$

Note that in Eq.1.6 the contribution of radiation losses is not included. The intermediate stage is considered to last until radiation losses begin to be relevant, and the assumption of adiabatic expansion ceases to be valid. To estimate the cooling timescale and then the expected duration of the intermediate evolutionary phase, one needs to calculate the shocked wind temperature, which is:

$$T_2 = \frac{P}{k_B n_2} \simeq 25.5 \times 10^6 \left(\frac{L_w}{10^{37} \,\mathrm{erg \, s^{-1}}}\right)^{2/5} \left(\frac{n_0}{10 \,\mathrm{cm^{-3}}}\right)^{3/5} \left(\frac{n_2}{10^{-2} \,\mathrm{cm^{-3}}}\right)^{-1} \left(\frac{t}{1 \,\mathrm{Myr}}\right)^{-4/5} \mathrm{K}$$
(1.12)

where  $n_2$  is the density in the bubble. The expected temperature is of the order of a few  $10^7$  K. However, one needs to account also for the thermal flux due to the heat conductivity between the cold dense shell and the hot shocked plasma, which rapidly cools the bubble temperature to lower values. The heat flux from the hot bubble is:

$$Q = \kappa_S \frac{\partial T}{\partial r} \tag{1.13}$$

where  $\kappa_S$  is the thermal conductivity for a fully ionized plasma (Spitzer, 1962):

$$\kappa_S = CT^{5/2} \simeq 1.2 \times 10^{14} \left(\frac{T}{10^8 \,\mathrm{K}}\right)^{5/2} \mathrm{erg \, s^{-1} cm^{-1} K^{-1}}$$
(1.14)

with  $C = 1.2 \times 10^{-6} \text{ erg cm}^{-1} \text{ s}^{-1} \text{ K}^{-7/2}$ . In this way, the conduction timescale ( $t_{cond}$ ) can be estimated as the ratio between the thermal energy and Q:

$$t_{cond} = \frac{1}{2} n_2 k_B \frac{T_2}{Q} R_b \simeq 2 \times 10^3 \left(\frac{n_2}{10^{-2} \,\mathrm{cm}^{-3}}\right) \left(\frac{T}{10^7 \,\mathrm{K}}\right)^{-5/2} \left(\frac{R_b}{60 \,\mathrm{pc}}\right)^2 \mathrm{yr}$$
(1.15)

where we have approximated  $\partial T/\partial r \approx T_2/R_b$ . This extremely short timescale indicates that the plasma immediately cools to lower temperatures, transferring heat to the cold, dense shell. Consequently, in a steady-state regime, temperatures reach values such that the heat flow has the same order of magnitude as the mechanical energy flow ( $F_m$ ), which can be written as:

$$F_m = \frac{5}{2} P_2 \frac{dR_b}{dt} \,. \tag{1.16}$$

Equating Eq.1.13 and Eq.1.14 with Eq.1.16 leads to the temperature expression:

$$T_2 \simeq \left(\frac{5P_2R_b^2}{2tC}\right)^{2/7} \simeq 2.3 \times 10^6 \left(\frac{L_w}{10^{37} \,\mathrm{erg}\,\mathrm{s}^{-1}}\right)^{8/35} \left(\frac{n_0}{10 \,\mathrm{cm}^{-3}}\right)^{2/35} \left(\frac{t}{1 \,\mathrm{Myr}}\right)^{-6/35} \mathrm{K}$$
(1.17)

where we used again the approximation  $\partial T/\partial r \approx T/R_b$  and  $dR_b/dt \approx R_b/t$ , and we also made the dependencies of  $P_2$  and  $R_b$  explicit using Eq. 1.9 and Eq. 1.8 respectively. Due to heat conductivity, the temperature in the bubble is roughly one order of magnitude less than what is expected from a shocked gas, as given by Eq. 1.12. With this temperature in mind, we can now estimate the radiative cooling timescales. A hot plasma with temperatures ranging between  $\sim 10^4 - 10^7$  K loses energy as a consequence of collisional excitation of bound electrons. In this regime, the power loss can be approximated as (Draine, 2011):

$$\frac{dE}{dVdt} \simeq 1.1 \times 10^{-22} n_2^2 \left(\frac{T}{10^6 \,\mathrm{K}}\right)^{-0.7} \mathrm{erg} \,\mathrm{cm}^{-3} \mathrm{s}^{-1}$$
(1.18)

which leads to a radiative cooling timescale of

$$t_{cool} = \left(\frac{3}{2}n_2k_BT\right) \left(\frac{dE}{dVdt}\right)^{-1} \simeq 6 \left(\frac{n_2}{0.01 \text{ cm}^{-3}}\right)^{-1} \left(\frac{T}{10^6 \text{ K}}\right)^{1.7} \text{ Myr}.$$
(1.19)

Considering an average temperature of a few  $10^6$  K, the radiative cooling should start to be relevant after a few tens of Myrs.

Before proceeding further, it is interesting to point out the following phenomenon. As a direct result of the heat flux from the hot shocked wind region to the swept-up shell, the cold material evaporates into the bubble, carrying a significant amount of mass that increases the gas density. Following the same approach as Castor et al. (1975), a simple way to treat the evaporation process is to assume that the inward gas flux from the shell to the hot cavity is approximated by a stationary plane parallel flow. If so, the mass evaporation rate from the shell ( $\dot{M}_s$ ), assuming a constant pressure flow and neglecting radiation losses, is well described by (Zel'Dovich & Pikel'Ner, 1969):

$$\frac{\dot{M}_s}{4\pi R_b^2} \frac{dH}{d\zeta} = \frac{d}{d\zeta} \left( \kappa_S \frac{dT}{d\zeta} \right)$$
(1.20)

where  $\zeta = R_b - r$ , H is the specific enthalpy  $H = 5k_BT/2\mu$ , with  $\mu = 0.62m_p$  as the mean molecular weight, defined as:

$$\mu = \frac{\sum_{j} n_j A_j}{\sum n_j + n_e} \tag{1.21}$$

where  $n_j$  and  $A_j$  are the numerical density and the mass number of a given j-th ion respectively, while  $n_e$  is the electron density. The mass loss rate can be obtained by integrating Eq. 1.20 from  $\zeta = 0$  to  $\zeta = R_b$ . However, to do so, one needs to know the temperature profile in the bubble. The expression reported in Eq. 1.17 can be roughly considered as an average value in the cavity.

The calculation of the temperature profile is a somewhat involved procedure, and has been carried out by Weaver et al. (1977). The temperature has a non uniform profile, with an overall decreasing trend towards the cold shell:

$$T(\zeta) = T_2 (1 - \zeta)^{2/5}$$
(1.22)

where  $T_2$  is given by Eq. 1.17. Using the latter equation leads to the following expression for the mass evaporation rate:

$$\dot{M}_s = \frac{16\pi\mu}{25k_B}\kappa_S R_b \,, \tag{1.23}$$

which can be further expanded by expressing the dependencies of  $\kappa_S$  and  $R_b$  via Eq. 1.14 and Eq.1.8:

$$\dot{M}_s \simeq 2 \times 10^{-4} \left(\frac{L_w}{10^{37} \,\mathrm{erg \, s^{-1}}}\right)^{27/35} \left(\frac{\rho_0}{10m_p \,\mathrm{cm^{-3}}}\right)^{-2/35} \left(\frac{t}{1 \,\mathrm{Myr}}\right)^{6/35} \mathrm{M}_{\odot} \mathrm{yr^{-1}}$$
 (1.24)

On average, the mass evaporation from the shell is significantly larger than the cluster

mass loss rate induced by stellar winds (see § 2.5.1).

#### Late stage of a wind bubble

As the contribution of radiation losses becomes relevant, the system deviates from the adiabatic expansion solution. To understand the evolution in such a condition, one needs to modify Eqs. 1.5, 1.6 and 1.7 by including the contribution of radiation losses. In addition, one must take into account the resulting decrease in volume of the hot bubble caused by the expansion of the TS as a result of the lower pressure induced by the decrease in temperature. The new sets of equations that must be solved are then:

$$\begin{cases} \frac{d}{dt} \left[ M_s \frac{dR_b}{dt} \right] = 4\pi R_b^2 (P - P_3) \\ \frac{dE}{dt} = L_w - 4\pi R_b^2 P \frac{dR_b}{dt} - L_{rad} \\ E = 2\pi (R_b^3 - R_{TS}^3) P \end{cases}$$
(1.25)

where  $P_3$  is the pressure of the outer shell, and

$$L_{rad} = \int_{R_{TS}}^{R_b} 4\pi n_2^2(r) \Lambda(T(r)) r^2 dr \,.$$
(1.26)

In principle, the system can be solved numerically by calculating  $L_{rad}$  at every instant of time, if the an expression for the temperature is provided (Weaver et al., 1977). However, such approach is computationally expensive. A cheap alternative to find an approximate description of the evolution is to parametrize the radiation losses as a constant fraction  $\xi_{rad}$  of the wind luminosity:

$$L_{rad} = \xi_{rad} L_w \,. \tag{1.27}$$

The new expressions for  $R_b$  and  $R_{TS}$  are similar to the previous ones but with the wind brightness value rescaled by a factor  $(1 - \xi_{rad})$ .

At this point, however, the following should be noted: the wind bubble model developed by Weaver et al. (1977) was designed for structures generated by single massive stars. In our case, in which we are dealing with YMSCs, the evolution of the bubble during the final stage can be significantly modified by the occurrence of several other physical processes that take place on a shorter time scale. Since the late phase occurs after a few tens of Myr, if not gravitationally bound, the star cluster could have dispersed before entering in the last evolutionary phase, given that the crossing time is of the order of  $\sim 10$  Myr. Furthermore, after 10 Myr, a significant number of massive stars should have ended their life cycle, generating a high number of supernova explosions whose feedback can severely alter the dynamic evolution of their surroundings.

#### Limitations of the Weaver's wind bubble model

Despite the simplicity and elegance with which it describes a complex phenomenon, the prescription of Weaver et al. (1977) still remains extremely simplified model with limitations that we are now going to briefly summarize.

First of all, the model was conceived for wind bubbles around massive stars. Hence, there is the underlying assumption that the central source injecting the wind is pointlike compared to the system overall size. This could not be the case for some YMSCs, which means that the model is still valid if the cluster is compact enough compared to the system size.

On a very general ground, the compactness condition can be translated into the requirement for the existence of a collective wind from the cluster. This can be formalized as  $R_{YMSC} \ll R_{TS}$ , where  $R_{YMSC}$  is the size of the star cluster. After a more critical look, instead of the radius of the cluster core, what really matters is the overall spatial distribution of the most massive stars. This is because the most massive stars are the ones that contribute the most to the wind energy budget, and are hence the main pillars sustaining the TS structure.

In this regard, it is important to underline the phenomenon of mass segregation in stellar clusters. Several observational pieces of evidence (Lada & Lada, 1991; Hillenbrand & Hartmann, 1998; Elmegreen et al., 2000; Jiang et al., 2002) have shown that the initial mass function of stars in a cluster is characterized by a spatial dependence described by a power-law with an index that becomes harder towards the center of the cluster. This effect seems to be related to the formation mechanism of stellar clusters (Lada & Lada, 2003; Karam & Sills, 2022), and hence, particularly enhanced in young clusters. As consequence, the most massive stars tend to sprout in a very compact space at the very center of the stellar cluster. So, even if it is not guaranteed that a YMSC can generate a collective wind, the compact spatial distribution of massive stars within a cluster favors this scenario.

As for other limitations of the model of Weaver et al. (1977), it also does not account for other types of feedback, such as for example direct radiation pressure. Indeed it has been estimated that direct radiation pressure feedback can be as efficient as wind feedback (see Krumholz et al. (2019) and references therein). This fact could affect the dynamics of the shell.

An additional, and potentially significant limitation, is the assumption of evolution in a uniformly distributed medium. While this could be true in the case of isolated massive stars, it is likely not in the case of YMSCs, as they are surrounded by the structure of the parent GMC which is highly non-uniform. The presence of low-density regions generated by the porosity of the GMC structure can induce severe leaks of the hot shocked gas from the bubble. This can dramatically reduce the pressure, thus stopping the expansion. Numerical simulations (Rogers & Pittard, 2013) and observations (Lopez et al., 2014) seem to point out that, if the gas is not well confined, gas leaks are a common phenomenon.

In parallel to gas leaks, a further process that can seriously affect the evolution of the bubble is the cooling produced by hot-cold gas mixing at the contact discontinuity (Rosen et al., 2014). The mixing is induced by both turbulent motions at the interface (Lancaster et al., 2021a), directly generated by the turbulent behavior of the GMC gas, and by the rise of several instability modes at the contact discontinuity, such as thin shell (Vishniac, 1983) and Rayleight-Taylor-like instabilities (Bucciantini et al., 2004). As a direct consequence of gas mixing, the density of the bubble increases. Eq. 1.19 shows that the radiative cooling scales with  $n_2^{-1}$ , so a substantial growth in the density may induce catastrophic radiative cooling. A secondary effect is that the instabilities at the contact discontinuity cause the fragmentation of the shell (Lancaster et al., 2021b), which leads to a considerable increase of the contact surface between the hot and cold gas. This makes thermal conduction even more efficient and, simultaneously, boosts the mass evaporation rate, causing the density of the bubble to increase further.

Lastly, the solution for the intermediate evolutionary stage given by Weaver et al. (1977) is not self-consistent. In fact, Weaver et al. (1977) calculates the evolution of the bubble boundary without accounting for the thermal energy losses due to the heat exchange with the cold shell. To obtain the correct solution, one needs to modify Eq. 1.6 in:

$$\frac{dE}{dt}(r) = L_w - 4\pi R_b^2 \left[ P \frac{dR_b}{dt} + \kappa_S \frac{\partial T(R_b)}{\partial r} \right]$$
(1.28)

As the temperature of the bubble decreases so does the pressure. As a consequence, the size of the TS increases. In parallel, the expansion of the forward shock slows down, and the bubble size becomes smaller. Obtaining a full solution in this situation becomes challenging and non-trivial since now the energy equation has a radial dependence. Nevertheless, similar to what has been done for radiative losses in the late evolutionary stage, the heat conduction can be approximated as a fraction of the overall wind luminosity:

$$\frac{dE}{dt} \simeq L_w - 4\pi R_b^2 \left( P \frac{dR_b}{dt} + \kappa_S \frac{T}{R_b} \right) = (1 - \xi_{th}) L_w - 4\pi R_b^2 P \frac{dR_b}{dt}$$
(1.29)

which means that if 50% of the wind luminosity is lost due to thermal conduction, the forward shock and TS sizes vary by  $\sim 15\%$ .

# **1.3 Cosmic Rays and the enigma of their origins: Young Massive Star Clusters as Cosmic Ray factories?**

Discovered more than a century ago by Victor Hess, Cosmic Rays (CRs) are charged particles mainly composed of nuclei (99%, of which 87% are protons, and 12% are alpha particles with a smaller fraction of heavier nuclei) and a minor fraction of electrons and antimatter. CRs are a fundamental component of the Cosmos, affecting the environment of galaxies. Indeed, the energy density associated with CRs in the Milky Way is of the order of  $\sim 1 \text{ eV cm}^{-3}$  (Webber, 1998), which is similar to that of the interstellar starlight radiation field  $\sim 0.5 \text{ eV cm}^{-3}$  (Mathis et al., 1983), to that of the average ( $\sim 3 \mu$ G) galactic magnetic field (Webber, 1998), and to that of the cosmic microwave background  $\sim 0.26 \text{ eV cm}^{-3}$  (Workman et al., 2022). CRs are often investigated from the high-energy astrophysics prospective, but, as a matter of fact, they are also a vital ingredient for the physics of the interstellar medium and of star formation.

Unquestionably, low-energy CRs are one of the main ingredients regulating star formation in galaxies, providing a negative feedback. One can in this regard, try to evaluate what is the star formation rate in the Milky Way in the assumption of the total absence of regulating mechanisms. To do so, we can approximately estimate the star formation rate as:

$$SFR_{MW} \simeq \frac{M_{MW}}{\tau_{ff}}$$
 (1.30)

where  $M_{MW} \approx 8 \times 10^9 \text{ M}_{\odot}$  is the Milky Way gas mass (Nakanishi & Sofue, 2016) and  $\tau_{ff}$  is the free fall time describing the timescale for the collapse of a neutral cloud, which can be written as (Spitzer, 1978):

$$\tau_{ff} = \sqrt{\frac{3\pi}{32Gm_p n_C}},\tag{1.31}$$

where  $n_C$  is the numerical density of the cloud. If we assume an average density of  $n_C = 10 \text{ cm}^{-3}$ , the free fall time is  $\tau_{ff} \approx 16.3 \text{ Myr}$  and the star formation rate calculated using Eq.1.30 is  $\approx 500 \text{ M}_{\odot} \text{ yr}^{-1}$ , which is far from the observed value of  $SFR_{MW} = 2.0 \pm 0.7 \text{ M}_{\odot} \text{ yr}^{-1}$  (Elia et al., 2022).

Accounting for the CR-induced ionization can significantly reduce the expected star formation rate, as partially ionized clouds are more stable against gravitational collapse. This is because during gravitational collapse, the ionized component of a cloud remains frozen to the cloud magnetic field, which opposes resistance to collapse-induced compression. The neutral material is then slowed down thanks to the coupling generated by neutral-ion scattering with the ionized component, which results in a lengthening of the collapse characteristic time scales<sup>4</sup>, making the star formation process less efficient.

The significance of the CR role is due to their ability to penetrate deeply into the cores of dense molecular clouds, providing ionization where other ionizing sources cannot. This is the case, for example, for X-rays that are usually absorbed in the first layers of a molecular cloud (McKee, 1989). Due to their ability to ionize, as we shall see in § A.2, low-energy CRs also prove to be crucial in regulating astrochemical processes, indirectly inducing the formation of complex molecules in the interstellar medium (Dalgarno, 2006).

From a pure observational point of view, the spectrum of CRs observed at the Earth is well described by a broken power law, starting from  $\sim 10$  GeV, with three breaks over almost 12 decades in energy (Fig. 1.3). Below 10 GeV, the CR spectrum is modified by solar modulation, preventing the lowest energy component from penetrating within the Solar System (Gleeson & Axford, 1968; Potgieter, 2013). Precise measurement of the spectrum at low energy was made possible by the Voyager 1 spacecraft only after exiting the heliosphere (Cummings et al., 2016).

At energies above 10 GeV the spectrum is characterized by a spectral index of  $\sim -2.7$ . The first break appears at  $\sim 3 \times 10^{15}$  eV, where the spectrum steepens with the spectral index becoming  $\sim -3.1$ , while the second break occurs instead at  $\sim 10^{18.5}$  eV, inducing a hardening of the spectrum and bringing the index back to  $\sim -2.7$  (Workman et al., 2022). The first and second breaks are respectively known as the *knee* and the *ankle*, given the similarity of the overall spectrum to a human leg.

As far as the ankle is concerned, the general consensus is that this feature is caused by the extragalactic CR population beginning to dominate the Galactic population (Bird et al., 1994; Apel et al., 2013). Undoubtedly, if the highest energy CRs are protons of extragalactic origin, then, at energies above  $\sim 5 \times 10^{19}$  eV, the CR spectrum is expected to be suppressed due to the pion photoproduction mechanism with the cosmic microwave background. This effect is known as the Greisen–Zatsepin–Kuzmin limit (or GZK cutoff) (Greisen, 1966). The existence of this cut-off has been shown by several ultra-high energy experiments (Abbasi et al., 2008; Verzi, 2019). Although it is still unclear whether we are witnessing the GZK effect or rather observing the maximum energy achievable by cosmic accelerators. The answer to this important question depends on the still uncertain composition of CRs at the highest energies.

Of greater relevance to the topic of this thesis is the origin of the knee. The presence of the knee has been attributed to the maximum energy reachable by galactic sources of CRs. This idea arose from the evidence of a gradual change in the chemical composition of CRs at energies above a few PeV. Above the knee, the fraction of heavy nuclei in CRs appears to increase, and this trend seems to persist up to  $10^{17}$  eV (Höorandel, 2006). It follows that, in this scenario, the knee could result from the superposition of cutoffs

<sup>&</sup>lt;sup>4</sup>This process is often named ambipolar diffusion (Draine, 2011)



Figure 1.3: CR energy spectrum as observed at the Earth. The plot shows also for comparison the spectra of the galactic diffuse  $\gamma$ -ray emission, the isotropic diffuse  $\gamma$ -ray background (IGRB), and the neutrino/anti-neutrino flux. See Evoli (2020) and references therein.

in the spectra of different elements as the large majority of acceleration processes are rigidity<sup>5</sup> dependent, i.e. proportional to the particle electric charge (Blasi, 2013). This means that if protons are accelerated in the sources to a maximum energy  $E_{p,max} \approx 3$ - $-5 \times 10^{15}$  eV, then an iron nucleus will be accelerated to  $E_{Fe,max} = 26E_{p,max} \approx 1 \times 10^{17}$  eV (assuming that during acceleration the iron nuclei are fully ionized, therefore the unscreened charge is Z = 26).

Clearly, in order to understand the plausibility of this framework, it is necessary to investigate the properties of the galactic CR sources. Since the first half of the 20th century, supernova remnants (SNR) were suggested as possible candidates as galactic CR accelerators. SNRs are the supersonic ejected material released after the supernova explosion of a massive star<sup>6</sup>. or of a white dwarf exceeding the Chandrasekhar limit. The suggestion of SNRs as Galactic CR sources was first made by Baade & Zwicky (1934), and later motivated by Ginzburg & Syrovatskii (1963). The suggestion it is as robust as it is elegant in its simplicity. Whatever the identity of the galactic accelerators, they must be

<sup>&</sup>lt;sup>5</sup>The rigidity of a particle with charge Ze and momentum p can be defined as  $R_q = pc/Ze$ 

<sup>&</sup>lt;sup>6</sup>Collapse supernovae (sometimes also classified as Type II or Type Ib and Ic following the old spectroscopic classification), which occur for stars with masses higher than 8  $M_{\odot}$  are though to account for 80% of these explosions. In addition there are Type Ia SNe which arise when an accreting white dwarf exceeds Chandrasekhar limit generating a thermonuclear explosion. See Vink (2012, 2020) and references therein for comprehensive reviews on SNRs.

able to sustain the observed CR luminosity, which is:

$$L_{CR} = \frac{U_{CR} V_{MW}}{\tau_{conf}} \approx 2.5 \times 10^{40} \text{erg s}^{-1}$$
(1.32)

where  $U_{CR} \approx 1 \text{ eV cm}^{-3}$  is the galactic CR energy density,  $V_{MW}$  is the Milky Way volume<sup>7</sup>, and  $\tau_{conf} \approx 20$  Myr is the CR confinement timescale in the Galaxy, which can be estimated from measurements of the abundance ratio of unstable isotopes (Connell, 1998; Hams et al., 2004). In the case of SNRs, the kinetic energy of a supernova explosion is of the order of  $E_{SN} \approx 10^{51}$  erg (Carroll & Ostlie, 1996). Assuming that the supernova explosion rate in the Milky Way is  $\tau_{SN} = 1/50 \text{ yr}^{-1}$ , then the power injected by supernova explosion is:

$$L_{SN} = \frac{E_{SN}}{\tau_{SN}} \approx 6.3 \times 10^{41} \text{erg s}^{-1}.$$
 (1.33)

This implies that if a few percent of the power injected by supernova explosions goes in CR acceleration, then, SNRs can easily account for the observed CR luminosity. But can SNRs actually accelerate CRs to the knee energy?

The supersonic motion of the ejecta produces a collisionless shock wave that propagates in the ISM. Particle acceleration in collisionless shocks is believed to occur through the first-order Fermi mechanism (also known as Diffusive Shock Acceleration) (Blandford & Ostriker, 1978; Drury, 1983). We can briefly summarize the process as follows. Let us consider the shock as a discontinuity between two regions with different flow velocities, we define these two velocities in the reference system of the shock as  $u_1$  and  $u_2 < u_1$ . We shall refer to the region of unperturbed plasma with speed  $u_1$  as *upstream*, and the region of shocked gas with speed  $u_2$  as *downstream*. Suppose now that particles, with some velocity v in the rest frame of the shock, enter the upstream, and start to diffuse through scattering with magnetic turbulence. The diffusion causes the particle velocity distribution to rapidly become isotropic in the reference frame of the upstream. After some time, some particles may be able to cross again the shock, entering the downstream. Again, particle will start to diffuse through scattering with magnetic turbulence that is associated with the downstream plasma. On average, particles will see plasma on the other side of the shock moving in their direction with a speed  $\sim |u_1 - u_2|$ . Each time the particles cross the shock, they will experience an electric field  $\sim |u_1 - u_2|/c$  times the magnetic field and gain energy. The process is analogous to that of collision with moving wall, and collisions are all head-on, hence inducing a gain in energy.

It can be shown through that particles will on average increase their energy as  $\Delta E \propto |u_1 - u_2|E/c$ . Eventually, particles will return to the upstream side of the shock. The gain

 $<sup>^7</sup>As$  a spiral galaxy, the Milky Way volume can be easily approximated as a thick disc with radius  ${\sim}15$  kpc and height  ${\sim}300$  pc.

occurs at every shock crossing, and several cycles of diffusion back and forth across the shock will lead to a significant increase in particle energy.

To estimate the maximum energy obtained in SNRs, one has to equate the acceleration timescales ( $\tau_{acc}$ ) with the time for which the source is efficiently accelerating particles<sup>8</sup>. The acceleration time is clearly related to the timescale with which the upstreamdownstream-upstream cycle occurs ( $t_{cycle}$ ), and can be calculated as (Drury, 1983):

$$\tau_{acc} = \frac{t_{cycle}}{\Delta E/E} = \frac{3}{u_1 - u_2} \left( \frac{D_1}{u_1} + \frac{D_2}{u_2} \right) \simeq 8 \frac{D_1}{u_{shock}^2}$$
(1.34)

where  $\Delta E/E$  is the average energy gain per cycle,  $u_{shock}$  is the shock velocity, while  $D_1$ and  $D_2$  are the diffusion coefficients in the upstream and downstream respectively. Generally, the diffusion coefficient can be defined as:

$$D = \frac{1}{3} \lambda_{mfp} v_p \tag{1.35}$$

where  $\lambda_{mfp}$  is the mean free path and  $v_p = \beta_p c$  is the particle speed, parametrized as a fraction  $\beta_p$  of the light speed c. The diffusion is mediated by the scattering with magnetic field irregularities that have length scales comparable with the particle Larmor radius  $(r_L)$ . In the most extreme case, when the magnetic irregularities  $\delta B$  are comparable with the average magnetic field B, one has  $\lambda_{mfp} = r_L$ . This is usually called the *Bohm regime*, and the diffusion coefficient reads:

$$D(E) = \frac{1}{3}r_L(E)\beta_p c = \frac{1}{3}\frac{E}{Ze_0B}\beta_p c$$
(1.36)

where E and Ze are the particle energy and charge respectively, with e as the electron elementary charge.

SNRs can efficiently accelerate particles during the so-called ejecta-dominated phase (sometimes referred to as the free expansion phase), defined as the evolutionary stage for which most of the explosion kinetic energy is confined in the freely expanding ejecta (Vink, 2020)

$$E_{SN} = \frac{1}{2} M_{ej} u_{shock}^2 \tag{1.37}$$

During this phase, the fast shock can easily reach CRs that are diffusing upstream, allowing efficient acceleration. This can be easily seen considering that the shock radius scales as  $R_{shock} \propto t$  during the ejecta dominated phase, while the particle diffusion length is proportional to  $\propto t^{1/2}$ . As the swept-up mass becomes comparable to the ejecta mass

<sup>&</sup>lt;sup>8</sup>This statement is correct if the energy losses of particles are negligible, which is likely the case for hadrons but typically is not for leptons. In the latter case, the comparison must be made with the Inverse Compton or Synchrotron cooling time scales.

 $(M_{ej} = 4\pi \rho_{ISM} R_{shock}^3/3)$ , the SNR evolution enters the Sedov-Taylor phase (Vink, 2020). During this stage, the most energetic particles are not caught anymore by the expanding shock  $(R_{shock} \propto t^{2/5})$  and are thus free to escape in the upstream region, making the acceleration mechanism no longer efficient. The time at which the Sedov-Taylor phase starts can be estimated as:

$$\tau_{ST} = \frac{R_{shock}}{u_{shock}} \simeq 220 \left(\frac{M_{ej}}{1 \,\mathrm{M}_{\odot}}\right)^{5/6} \left(\frac{E_{SN}}{10^{51} \,\mathrm{erg}}\right)^{-1/2} \left(\frac{\rho_{ISM}}{1 m_p \,\mathrm{cm}^{-3}}\right)^{-1/3} \mathrm{yr}.$$
(1.38)

Finally, equating  $\tau_{ST}$  with  $\tau_{acc}$  leads to the estimate of the maximum particle energy in SNRs:

$$E_{max,SNRs} = 2.5 \times 10^{13} Z \left(\frac{B}{1\,\mu\text{G}}\right) \left(\frac{M_{ej}}{1\,\text{M}_{\odot}}\right)^{-1/6} \left(\frac{E_{SN}}{10^{51}\,\text{erg}}\right)^{1/2} \left(\frac{n_{ISM}}{1\,\text{cm}^{-3}}\right)^{-1/3} \text{eV}$$
(1.39)

In a standard situation, the maximum energy for a proton is expected to be about one order of magnitude below the PeV. This means that, in order to consider SNRs as PeVa-trons, a stronger magnetic field is required.

Several magnetic field amplification processes have been proposed in the literature, such as non-resonant (Bell, 2004) and resonant (Skilling, 1975) streaming instabilities, amplification driven by the dynamics of the shock impinging in density fluctuation of turbulent magnetized plasma (Guo et al., 2012), and acoustic instabilities (Drury & Downes, 2012). However, even including the most efficient of these amplification mechanisms, the non-resonant streaming instability, a standard SNR can hardly accelerate particles up to PeV energies, unless specific extreme situations with high velocity shocks expanding in dense environments are considered (Cristofari, 2021). The difficulty of SNRs to reach PeV energies makes the interpretation of these objects as a single population of Galactic accelerators problematic, and prompts us to consider the possible presence of an additional class of particle accelerators.

The maximum energy of CRs in SNRs is not the only issue related to these sources. Measurements of the CR composition at the Earth have shown an excess in some isotopic ratios, such as the <sup>22</sup>Ne to <sup>20</sup>Ne ratio, which is found  $5.3 \pm 0.3$  times higher than the solar value (Wiedenbeck & Greiner, 1981; Binns et al., 2008). The <sup>22</sup>Ne is copiously produced by Wolf-Rayet (WR) stars (Prantzos et al., 1986; Maeder & Meynet, 1993), that are the late evolutionary stage of very massive O-type stars ( $\geq 25M_{\odot}$ ) (Crowther, 2007). This means that at least some fraction of CR sources must be located in environments close to WR stars, which are very common in YMSCs. The presence of such overabundance then can be considered as a strong clue that points to stellar clusters as possible CR factories. Undoubtedly, if CR acceleration were to occur in these objects, some of the

wind material of a WR would end up in the galactic CR population, thus explaining the observed excess. The enigma of the origin of CRs then shifts to the general question of whether and how much YMSCs can actually contribute to the acceleration of CRs in the galaxy. In the past years, different models for particle acceleration in YMSCs have been proposed. We will summarize in the next subsections some of the most popular.

#### 1.3.1 Acceleration mechanisms in the core of a YMSCs

In its core, a YMSC may enclose a significant number of massive stars, and each of them can launch a fast wind that collides with the ambient medium and generates a TS. Particle acceleration occurs at the stellar wind TS following the diffusive shock acceleration process. The final distribution of freshly accelerated CRs from the YMSC will then arise as a combination of injected particles from different stellar TSs: such a scenario has been investigated by Klepach et al. (2000).

Let us consider a system composed of several stellar wind cavities (Fig. 1.4). Each of the stellar winds is assumed to be highly supersonic, with constant radial velocity  $u_1$ . The winds impact with the hot shocked material at a distance  $R_{TS}$ . The time evolution of  $R_{TS}$  is somewhat similar to that given by Eq.1.11, so it is slowly expanding in time and can be considered as stationary. Following the classical theory of shocks, the gas velocity downstream of the TS is:

$$u_2 = \frac{u_1}{\mathcal{R}} \left(\frac{r}{R_{TS}}\right)^{-2} \tag{1.40}$$

where  $\mathcal{R}$  is the TS compression ratio. We will further assume that the overall volume occupied by the wind cavities is much less that the system volume, or equivalently, that  $R_{TS}$  is much smaller than the average distance d between the stars. Formally this condition translates into requiring the filling factor  $\mathcal{W} = (R_{TS}/d)^3$  to be  $\mathcal{W} \ll 1$ .

On a very general ground, a CR distribution function  $f_{CR}(r, t, p)$  must obey the following transport equation<sup>9</sup>

$$\frac{\partial f_{CR}}{\partial t} - \vec{\nabla} D \vec{\nabla} f_{CR} + u \vec{\nabla} f_{CR} - \frac{p}{3} \vec{\nabla} u \frac{\partial f_{CR}}{\partial p} = q$$
(1.41)

where p is the particle momentum and q is the source term, which describes the injection of freshly accelerated CRs that is assumed to be occurring at the wind TSs. Clearly, the normalization of  $f_{CR}$  is determined by the total CR number density  $n_{CR} = 4\pi \int p^2 f_{CR} dp$ . The CR distribution function ( $\langle f_{CR} \rangle$ ) obtained in such a system will be an average over several  $f_{CR}$  calculated from a random distribution of winds. In order to estimate the CR

<sup>&</sup>lt;sup>9</sup>This is not the complete transport equation, as some terms are neglected, such as for example, potential energy losses and momentum diffusion, see 1.3.2.



Figure 1.4: System sketch for the multiple TS acceleration mechanism investigated by Klepach et al. (2000).

distribution, we need to solve for  $\langle f_{CR} \rangle$ . The equation for CR injected at a single spherical wind TS, can be written as:

$$\frac{\partial f_{CR}}{\partial t} - \frac{1}{r^2} \frac{\partial}{\partial r} r^2 D \frac{\partial f_{CR}}{\partial r} + u \frac{\partial f_{CR}}{\partial r} - \frac{r^2}{3} \frac{d(r^2 u)}{dr} p \frac{\partial f_{CR}}{\partial p} = Q\delta(r - R_{TS})$$
(1.42)

where Q(p) is the injection term for a single wind TS. The diffusion coefficient is likely to be different in the upstream  $(D_1)$  and the downstream  $(D_2)$ . Klepach et al. (2000) assume the upstream diffusion coefficient to scale as  $D_1r/R_{TS}$ , and that  $D_2$  is spatially constant. Klepach et al. (2000) additionally assume that both  $D_1$  and  $D_2$  are energy independent. Note that since  $\mathcal{W} \ll 1$ , one will have  $f_{CR} \simeq \langle f_{CR} \rangle$  for  $r \gg R_{TS}$ . To proceed, it is useful to define the following transformations:

$$\Phi(s,r,t) = \int_0^\infty p^{s-1} f_{CR}(p,r,t) dp$$
 (1.43)

$$\Psi(s,t) = \int_0^\infty p^{s-1} Q(p,t) dp$$
 (1.44)

$$X(s,r,t) = \int_0^\infty p^{s-1} q(p,r,t) dp,$$
 (1.45)

and to rewrite Eq. 1.41 accordingly:

$$\frac{\partial \Psi}{\partial t} - \nabla D \nabla \Psi + u \nabla \Psi + s \frac{\nabla u}{3} \Psi = X.$$
(1.46)

In principle, the equation for  $\langle f_{CR} \rangle$  can be obtained by averaging Eq. 1.46 over the entire system volume and then perform the inverse transformation. Both the averaging and inverse transform are lengthy, nontrivial calculations, and the derivation of  $f_{CR}$  may result cumbersome. We here summarize the final outcome under the following assumptions: (1) The gas motion is incompressible downstream of the wind TSs, and in addition, the shocked gas is able to diffuse far away from the TS and to fill the cluster volume such that  $\langle \nabla \cdot u \rangle = 3\mathcal{W}u/\mathcal{R}R_{TS}$ . (2) The solution is calculated assuming a steady state regime  $(\partial f_{CR}/\partial t = 0)$ . (3) Repeated acceleration cycles due to crossing of different TSs may occur. This requires that the particles mean free path must be larger than the TS size:  $g_1 = R_{TS}u_1/D_1 \ll 1$  and  $g_2 = R_{TS}u_1/\mathcal{R}D_2 \ll 1$ . (4) CRs escape from the cluster boundary with a leakage timescale  $T_l$ . Under these assumptions, the average CR distribution function is:

$$\langle f_{CR}(p) \rangle = \frac{3\mathcal{R}Q_0}{u_1(\mathcal{R}-1)[1+(t_{acc}/T_l)(1+G^{-1})^{-1}]} \left(\frac{p}{p_0}\right)^{-S}$$
 (1.47)

where  $Q_0$  is the injection rate at momentum  $p_0$ ,  $t_{acc} = \mathcal{R}R_{TS}/3\mathcal{W}u_1$  is the acceleration timescale, and

$$S = \frac{3\mathcal{R}}{\mathcal{R} - 1} \left\{ 1 + \frac{t_{acc}}{\mathcal{R}T_l \left[ 1 + G \left( 1 + \frac{t_{acc}}{T_l} \right) \right]} \right\}$$
(1.48)

$$G = e^{g_2} - 1 \tag{1.49}$$

The obtained spectrum is an approximation of the real solution, as in general, the diffusion coefficients depend on energy. Note that for  $t_{acc} \ll T_l$ , the powerlaw index Sbecomes  $S \simeq 3\mathcal{R}/(\mathcal{R}-1)$  that is the standard result for acceleration at a plane parallel shock (Bell, 1978). Klepach et al. (2000) also estimate the maximum energy that can be reached with this acceleration mechanism, that is:

$$E_{max} \simeq 2 \times 10^{15} Z \left(\frac{D_2}{D_B}\right)^{-1} \,\mathrm{eV}$$
 (1.50)

where  $D_B$  is the diffusion coefficient in the Bohm regime. For standard values of winds from massive stars, the maximum energy is as high as a few  $10^{15}eV$  (Klepach et al., 2000), which is encouraging in terms of viewing YMSCs as a possible counterpart for SNRs as PeVatrons.

We may now wonder whether this scenario could actually develop in a YMCs, given that clusters tend to emerge from small clumps with sizes of a few pc and that the entire acceleration process is based on the assumption that  $W \ll 1$ . To do so, we can estimate the size of  $R_{TS}$  for an isolated star embedded in a hot bubble of shocked wind gas with pressure P given by Eq.1.9. The procedure is identical to that used to obtain the TS radius of the collective wind from a cluster (see Eq:1.10), with the difference that now the ram pressure is calculated for the wind of a single star:

$$\frac{\dot{M}_{\star}v_{w,\star}}{4\pi R_{TS}^2} = \frac{7}{(3850\pi)^{2/5}} L_w^{2/5} \rho_0^{3/5} t^{-4/5}$$
(1.51)

where  $M_{\star}$  and  $v_{w,\star}$  are the mass loss rate and the wind velocity for a single star. Note that  $L_w$  is the wind luminosity of the entire cluster, as the gas heating is provided by the overall stellar population. The size of the stellar wind TS is then:

$$R_{TS} \simeq 1.7 \left(\frac{\dot{M}_{\star}}{10^{-6} M_{\odot} \text{yr}^{-1}}\right)^{1/2} \left(\frac{v_{w,\star}}{2000 \text{ km s}^{-1}}\right)^{1/2} \left(\frac{L_w}{10^{37} \text{erg s}^{-1}}\right)^{-1/5}$$

$$\left(\frac{\rho_0}{10 m_p \text{cm}^{-3}}\right)^{-3/10} \left(\frac{t}{1 \text{ Myr}}\right)^{2/5} \text{pc.}$$
(1.52)

For typical values of the parameters  $R_{TS} \sim 2$  pc. In general<sup>10</sup>, given the mass of a YMSC ( $M_{YMSC}$ ), the cluster size can be estimated by the following empirical cluster mass-radius relation (Pfalzner et al., 2016):

$$R_{YMSC} \simeq 1.82 \left(\frac{M_{YMSC}}{10^3 \,\mathrm{M}_{\odot}}\right)^{1/\delta} \,\mathrm{pc} \tag{1.53}$$

whit  $\delta = 1.71 \pm 0.07$ . Note that by cluster size we mean the half-mass radius (the radius within which half of the stellar mass is enclosed). Reasonably, the average stars spacing will be less than, or of the order of, the cluster size. Thus, considering the typical properties of a massive star in terms of wind luminosity, mass loss rate and wind speed (see Eq. 1.52),  $R_{TS}$  has a size similar to the average stellar distance expected in the case of a YMSC.

It is then clear that this model is likely not a good representation for these objects. Plus, in Eq.1.51, we are not taking into account the effect of pressure reduction due to the cooling induced by the heat flow towards the surrounding ISM, so the  $R_{TS}$  calculated in Eq.1.52 is possibly underestimated. Nonetheless, this framework could be valid for older clusters, especially after a few tens of Myr in the case of gravitationally unbound systems. Assuming that the stars velocity dispersion is  $\sigma = R_{YMSC}\sqrt{3GM_{YMSC}/4\pi R_{YMSC}^3}$ , after

<sup>&</sup>lt;sup>10</sup>From a theoretical point of view, the cluster mass-size relation is significantly affected by the dominant feedback mechanism operating during the emergence of the YMSC from its dense gas cradle (Krumholz et al., 2019).

some time t the cluster has expanded by an extent  $\Delta R$  of:

$$\Delta R = \sigma t \simeq 7.68 \left(\frac{M_{YMSC}}{10^3 \,\mathrm{M}_{\odot}}\right)^{(\delta-1)/2\delta} \left(\frac{t}{10 \,\mathrm{Myr}}\right) \mathrm{pc.}$$
(1.54)

After a few tens of Myr stars should be sparse enough to validate the mechanism proposed by Klepach et al. (2000). Eventually, the model could also be well describing the acceleration process in OB associations, which are structures with sizes spanning over a few tens of pc.

#### **1.3.2** Acceleration through efficient scattering with turbulence

The powerful massive star winds, or alternatively, the strong collective cluster wind, interacting with the ISM irregularities generate chaotic magnetohydrodynamic fluctuation with frozen in magnetic field, which end up permeating the entire hot bubble. The scattering with these fluctuations induces both spatial and momentum diffusion if the correlation lengthscales of the turbulent magnetic field are comparable with the particle Larmor radius (Blasi, 2013). The process of momentum diffusion is often called *second-order Fermi acceleration*, as it statistically induces an energy gain (Fermi, 1949). However, this mechanism is not particularly efficient as the energy gain scales as  $v_a^2/c^2$ , where  $v_a = B/\sqrt{4\pi\rho}$  is the plasma Alfvén speed, with  $\rho$  the plasma mass density. The momentum diffusion term can be included in the CR transport equation:

$$\frac{\partial f_{CR}}{\partial t} - \nabla D_{xx} \nabla f_{CR} + u \nabla f_{CR} - \frac{p}{3} \nabla u \frac{\partial f_{CR}}{\partial p} - \frac{1}{p^2} \frac{\partial}{\partial p} \left( p^2 D_{pp} \frac{\partial f_{CR}}{\partial p} \right) = Q \qquad (1.55)$$

where, to avoid confusion, we have renamed the spatial diffusion coefficient  $D_{xx}$ , while  $D_{pp}$  is the momentum diffusion coefficient. Although not generally relevant, secondorder Fermi acceleration may play a significant role in the case of strong turbulence, potentially boosting particle energies up to relatively high values (Bykov et al., 2020). One can estimate the maximum energy achieved in such a way by comparing the acceleration and the propagation time scales. For second order Fermi acceleration, the characteristic time scale is:

$$\tau_{acc} = \frac{p^2}{D_{pp}} \,, \tag{1.56}$$

where the momentum diffusion coefficient can be written as (Thornbury & Drury, 2014):

$$D_{pp} \simeq \frac{p^2 v_a^2}{9 D_{xx}} \,.$$
 (1.57)

In general, the spatial diffusion coefficient is directly linked to the type of plasma turbulence spectrum:

$$D_{xx} = \frac{1}{3} \frac{\beta c r_L}{\mathcal{F}(1/r_L)} \tag{1.58}$$

with

$$\mathcal{F}(k) = \frac{k\mathcal{P}(k)}{B_0^2/8\pi} \tag{1.59}$$

where k is the wave number and  $\mathcal{P}(k) \propto k^{-\alpha}$  is the turbulent magnetic field power spectrum. If the power is injected at a characteristic scale  $L_{inj}$ , then Eq.1.59 can be normalized to the injection scale and rewritten in terms of the magnetic field correlation length scale l = 1/k:

$$\mathcal{F}(l) = \mathcal{A}\left(\frac{l}{L_{inj}}\right)^{(\alpha-1)}$$
 (1.60)

In Eq.1.60 the index  $\alpha$  is correlated to the type of plasma turbulence cascade, and reads  $\alpha = 5/3$  for Kolmogorov-like turbulence (Kolmogorov, 1941),  $\alpha = 3/2$  for a Kraichnanlike cascade (Kraichnan, 1965), and  $\alpha = 1$  for a flat turbulence spectrum (Bohm diffusion regime). The constant  $\mathcal{A}$  is related to the total power in magnetic fluctuation  $\delta B_{tot}$ , through the condition  $\delta B_{tot}^2/B_0^2 = \Lambda_B \mathcal{A}$ , where  $B_0$  is the unperturbed magnetic field. We consider here the scenario where  $\delta B_{tot} = B_0$ . The parameter  $\Lambda_B$  depends on the turbulence spectral index  $\alpha$ , and is defined as:

$$\Lambda_B \equiv \int_{k_{inj}}^{k_{max}} \left(\frac{k}{k_{inj}}\right)^{1-\alpha} \frac{dk}{k} = \begin{cases} \frac{1}{\alpha-1} \left[1 - \left(\frac{k_{inj}}{k_{max}}\right)^{\alpha-1}\right] & \text{for } \alpha \neq 1\\ \ln\left(\frac{k_{max}}{k_{inj}}\right) & \text{for } \alpha = 1 \end{cases}$$
(1.61)

where  $k_{inj} = 1/L_{inj}$ . Generally, the parameter  $k_{max}$  corresponds to the inverse of the length scale at which turbulence thermal dissipation occurs. Note that, if this is the case, then one has typically  $k_{inj} \ll k_{max}$ , and for  $\alpha \neq 1$  we have  $\Lambda_B \simeq (\alpha - 1)^{-1}$ . In the case  $\alpha = 1$ , however, it appears unrealistic to extend the assumption of equal turbulent power per decade down to the dissipation scale. We consider this description only appropriate to wave modes within an interval  $[k_{inj}, k_{max}]$  in which it is likely that power injection occurs at all scales. For example, a reasonable guess for such range could be given by the length scale associated with the average distance between the stars, down to the characteristic length scale of wind irregularities.

In § 2.2 we will show that CR propagation in an expanding hot bubble is in general dominated by advection for particles with energies  $\lesssim 1$  TeV, so the propagation time scale can be estimated as:

$$t_{adv} \simeq \frac{R_b}{u_2} \simeq t_{age} \tag{1.62}$$

where  $t_{age}$  is the age of the wind-blown bubble (or equivalently the cluster age). Equating Eq.1.56 with Eq.1.62, leads to

$$\frac{9D_{xx}}{v_a^2} = t_{age} \Rightarrow \frac{3\beta c L_{inj}}{v_a^2} \Lambda_B \left(\frac{r_L}{L_{inj}}\right)^{(2-\alpha)} = t_{age} , \qquad (1.63)$$

and the particle maximum energy for the second order Fermi acceleration is readily obtained by considering  $r_L = E/eB_0$ :

$$E_{max}^{Turb} \simeq \left(\frac{e_0^{2-\alpha}}{12\pi\rho c}\right)^{1/(2-\alpha)} \Lambda_B^{-1/(2-\alpha)} L_{inj}^{(1-\alpha)/(2-\alpha)} B^{(4-\alpha)/(2-\alpha)} t_{age}^{1/(2-\alpha)}$$
(1.64)

where we have assumed that  $\beta \approx 1$ . Fixing  $\alpha$  to the three previously mentioned values, the maximum energies are:

$$E_{max,K41}^{Turb} \simeq 29 \left(\frac{B_0}{1\,\mu\text{G}}\right)^7 \left(\frac{L_{inj}}{1\,\text{pc}}\right)^{-2} \left(\frac{t_{age}}{10\,\text{Myr}}\right)^3 \left(\frac{\rho_2}{10^{-2}m_p\,\text{cm}^{-3}}\right)^{-3} \text{MeV}$$
(1.65)

$$E_{max,Kra}^{Turb} \simeq 6 \left(\frac{B_0}{1\,\mu\text{G}}\right)^5 \left(\frac{L_{inj}}{1\,\text{pc}}\right)^{-1} \left(\frac{t_{age}}{10\,\text{Myr}}\right)^2 \left(\frac{\rho_2}{10^{-2}m_p\,\text{cm}^{-3}}\right)^2 \text{GeV}$$
(1.66)

$$E_{max,Bohm}^{Turb} \simeq 5 \frac{1}{\ln(k_{inj}/k_{max})} \left(\frac{B_0}{1\,\mu\text{G}}\right)^3 \left(\frac{t_{age}}{10\,\text{Myr}}\right) \left(\frac{\rho_2}{10^{-2}m_p\,\text{cm}^{-3}}\right)^{-1} \text{TeV.}$$
(1.67)

The values obtained are far from PeV energies but remain considerably high, especially in the case of Bohm diffusion<sup>11</sup>.

Nevertheless, several caveats may limit the realistic efficiency of such an acceleration process. Acceleration is achieved under the core assumption of particle scattering with resonant fluctuations, and the underlying calculations are performed in the context of quasi-linear theory. However, recent simulations have demonstrated that, by using modern anisotropic magnetohydrodynamic turbulence theories, the wave-particle resonances may be strongly suppressed, severely reducing the efficiency of second order Fermi acceleration (see Lemoine (2021) and references therein).

In this section, we have considered the case where acceleration is provided by subsonic turbulence. In older clusters, the combined presence of fast winds and supernova shocks may induce the presence of supersonic turbulence. See the work of Bykov et al. (2020) for a comprehensive review of particle acceleration in such a framework.

 $<sup>^{11}</sup>$  This significantly depends on the value of  $k_{max}$  , which can potentially be several orders of magnitude above  $k_{inj}$  .

#### **1.3.3** Acceleration at the cluster wind termination shock

YMSCs are probably compact enough to allow single star winds to combine so as to create a collective cluster wind. As we showed in § 1.2.3, the fast collective cluster wind impacting on the surrounding hot bubble material produces a strong TS at which particle acceleration may occur. This specific case was recently studied by Morlino et al. (2021).

Let us now consider a YMSC that has developed a bubble structure and is now in the intermediate evolutionary stage. From the CR point of view, the system can be briefly described as follows: particles are accelerated at the TS via the diffusive shock acceleration mechanism, and subsequently escape from the acceleration site experiencing a combination of advection and diffusion in the hot bubble until they reach the forward shock. From there, CRs are free to leave the system by diffusing in the unperturbed ISM. Given the slow expansion rate of the cavity, the system can be considered as stationary, and if one assumes radial symmetry, and neglects second order Fermi acceleration and particles energy losses, the distribution  $f_{CR}(r, E)$  of CRs can be found by solving the following steady-state transport equation<sup>12</sup>

$$\frac{\partial}{\partial r} \left[ r^2 D(r,p) \frac{\partial f}{\partial r} \right] - r^2 u(r) \frac{\partial f}{\partial r} + \frac{d[r^2 u(r)]}{dr} \frac{p}{3} \frac{\partial f}{\partial p} + r^2 Q(r,p) = 0$$
(1.68)

where u(r) is the plasma speed, and D(r, p) is the spatial diffusion coefficient. The source term Q(r, p) describes the particle injection taking place at the TS:

$$Q(r,p) = \frac{\eta_{inj}n_1u_1}{4\pi p_{inj}^2} \delta(p - p_{inj})\delta(r - R_{TS}), \qquad (1.69)$$

where  $n_1$  is the density immediately upstream of the termination shock,  $u_1$  is the speed of the cold wind, and  $\eta_{inj}$  is the fraction of particle that are injected in the acceleration process with momentum  $p_{inj}$ . The global solution of Eq. 1.68 can be found in three main steps: first, the equation must be solved separately in the unperturbed ISM ( $r > R_b$ ) and upstream (in the cold cluster wind,  $r < R_{TS}$ ) and downstream of the TS (in the hot shocked wind bubble,  $R_{TS} < r < R_b$ ). Secondly, the solutions in these three zones are joined together using flux continuity at  $r = R_{TS}$  and  $r = R_b$ . Finally, one needs to specify two boundary conditions. This can be done by assuming no net flux at r = 0, and requiring that  $f_{CR}(r, E)$  at infinity matches the distribution of the galactic CR sea  $(f_{gal})$ .

Following the above-mentioned procedure, the CR radial distribution in the three

<sup>&</sup>lt;sup>12</sup>Note that the system is very similar to that described in §1.3.1 for the acceleration at the wind TS for a single star. The differences are limited to the boundary conditions considered.
zones is:

$$f_1(r,p) \simeq f_{TS}(p) \cdot exp\left[-\int_r^{R_{TS}} \frac{u_1}{D_1(r',p)} dr'\right]$$
 (1.70a)

$$f_2(r,p) = f_{TS}(p)e^{\alpha} \frac{1 + \beta(e^{\alpha_B - \alpha} - 1)}{1 + \beta(e^{\alpha_B} - 1)} + f_{gal}(p) \frac{\beta(e^{\alpha} - 1)}{1 + \beta(e^{\alpha_B} - 1)}$$
(1.70b)

$$f_{ism}(r,p) = f_2(R_b,p)\frac{R_b}{r} + f_{gal}(p)\left(1 - \frac{R_{TS}}{r}\right)$$
(1.70c)

with

$$\alpha = \alpha(r, p) = \frac{u_2 R_{TS}}{D_2(p)} \left(1 - \frac{R_{TS}}{r}\right)$$
(1.71a)

$$\alpha_B = \alpha(r = R_b, p) \tag{1.71b}$$

$$\beta = \beta(p) = \frac{D_{ism}(p)R_b}{u_2 R_{TS}^2}$$
(1.71c)

where the subscripts 1, 2 and *ism* refers orderly to values assumed by the variables in the upstream, downstream, and interstellar medium regions. Notice that Eq. 1.70a is a first-order approximation of the full solution presented by Morlino et al. (2021), which in principle should be formally obtained by iteratively solving Eq. 1.68 in the upstream.  $f_{gal}$  is the average spectrum of the Galactic CR sea, e.g. as inferred from AMS-02 data (Aguilar et al., 2015). Finally,  $f_{TS}$  is the distribution of injected particles at the TS.

The formal solution for  $f_{TS}$  can be written in the following form

$$f_{TS}(p) = s \frac{\eta_{inj} n_1}{4\pi p_{inj}^3} \left(\frac{p}{p_{inj}}\right)^{-s} e^{-\Gamma_1(p)} e^{-\Gamma_2(p)}.$$
(1.72)

The function is composed of three terms: the first one is the standard power-law spectrum resulting from particle acceleration in plane shocks. The second term contains the function  $\Gamma_1(p)$ , which depends itself on  $f_{TS}$ , implying a non-linear nature of the solution. From a physical point of view, the suppression term  $e^{-\Gamma_1(p)}$  can be seen as a modification, induced by the spherical geometry of the system, to the usual energy gain obtained in parallel shocks. The last term finally,  $e^{-\Gamma_2(p)}$ , describes the cut-off caused by the escape of particles at the bubble boundary.

The final form of  $f_{TS}$  is non-analytical. Nevertheless, it can be approximated with good accuracy using a modified power-law with an exponential cut-off, whose expression slightly changes depending on the model for particle diffusion around the TS:

$$f_{TS}(p) \simeq s \frac{\eta_{inj} n_1}{4\pi p_{inj}^3} \left(\frac{p}{p_{inj}}\right)^{-s} \left[1 + a_1 \left(\frac{p}{p_{max}}\right)^{a_2}\right] e^{-a_3(p/p_{max})^{a_4}},$$
(1.73)

Models	$a_1$	$a_2$	$a_3$	$a_4$	
Kolmogorov	10	0.308653	22.0241	0.43112	
Kraichnan	5	0.448549	12.52	0.642666	
Bohm	8.94	1.29597	5.31019	1.13245	

Table 1.1: Parameter values used to calculate the distribution of injected particles.

where  $p_{max}$  is related to the maximum achievable momentum in the system, and the parameters  $a_i$  depend on the type of magnetohydrodynamic turbulence in the plasma (see Tab. 1.1). The normalization of  $f_{TS}$  is determined by  $\eta_{inj}$  and  $p_{inj}$ , which can be usefully expressed in terms of  $\epsilon_{CR}$ , the fraction of cluster wind luminosity  $L_w$  converted into accelerated particles:

$$\epsilon_{CR}L_w = L_{CR} = 4\pi R_{TS}^2 u_2 \int f_{TS}(p) E_k(p) d^3p, \qquad (1.74)$$

where  $L_{CR}$  is the CR luminosity and  $E_k = E(p) - m_p c^2$  is the particles kinetic energy. Knowing that  $L_w = 4\pi m_p n_1 R_{TS}^2 u_1^3$ , from Eq.1.74 one can obtain an expression for  $\eta_{inj}$ , which can be used in Eq. 1.73 to obtain:

$$f_{TS}(p) \simeq \frac{3n_1 u_1^2 \epsilon_{CR}}{4\pi \Lambda_p (m_p c)^3 c^2} \left(\frac{p}{m_p c}\right)^{-s} \left[1 + a_1 \left(\frac{p}{p_{max}}\right)^{a_2}\right] e^{-a_3 (p/p_{max})^{a_4}}$$
(1.75)

where:

$$\Lambda_p = \int_{x_{inj}}^{\infty} x^2 f_{TS}(x) \left(\sqrt{1+x^2} - 1\right) dx,$$
(1.76)

with  $x = p/m_p c$ .

Interestingly, once the parameters of the star cluster are fixed,  $f_{TS}$  is fully described by only two parameters, namely the efficiency of CR production ( $\epsilon_{CR}$ ) and the spectral index of injected particles (s). Indeed, it is possible to fix  $p_{max}$ , or equivalently, the maximum energy of accelerated particles ( $E_{max}$ ), to the intrinsic properties of the YMSC. The maximum momentum of particles can be calculated by equating the particle diffusion length to the size of the TS<sup>13</sup>:

$$\frac{D_1(E_{max})}{u_1} = R_{TS} \,. \tag{1.77}$$

<sup>&</sup>lt;sup>13</sup>From a formal point of view, this approach to estimating  $E_{max}$  is somewhat approximate. In fact, one would have to consider the confinement of the particle in the downstream, since if the particle escapes the bubble it will immediately diffuse into the ISM, effectively terminating the acceleration process. This mechanism is formally included in the term  $e^{-\Gamma(p)_2}$  in Eq. 1.72.

This is because the probability of crossing the TS and being further accelerated decreases significantly when the particles have diffusion lengths greater than or comparable to  $R_{TS}$ . If we consider again the three turbulent cascade models mentioned above, given by the Kolmogorov, Kraichnan and flat (Bohm-like) spectrum, the corresponding diffusion coefficients upstream are easily obtained from Eq. 1.58, and are respectively:

$$D_{K41}(E) = \frac{1}{3} \Lambda_B \beta c r_L^{1/3} L_{inj}^{2/3}$$
(1.78)

$$D_{Kra}(E) = \frac{1}{3} \Lambda_B \beta c r_L^{1/2} L_{inj}^{1/2}$$
 (1.79)

$$D_{Bohm}(E) = \frac{1}{3} \Lambda_B \beta c r_L.$$
(1.80)

where  $r_L = E/e\delta B_{tot}$ .

The diffusion coefficients are directly linked to the intensity of the magnetic field fluctuations upstream of the TS. We assume that the total power in magnetic fluctuation  $\delta B_{tot}/8\pi = \int \mathcal{P}(k)dk$  is a fraction  $\eta_B$  of the wind luminosity, such that:

$$\frac{\delta B_{tot}^2}{4\pi} = \eta_B \frac{L_w}{4\pi u_1 R_{TS}^2}$$
(1.81)

we furthermore consider the scenario of strong turbulence, so that  $\delta B_{tot}/B_1 = 1$ , with  $B_1$  the magnetic field upstream of the TS, and we shall rename for simplicity  $\delta B_{tot} = \delta B_1$ . Solving Eq. 1.77 for  $E_{max}$  and rewriting the expression for  $R_{TS}$  using Eq. 1.11 and Eq. 1.1 leads to the following maximum energies:

$$E_{max}^{K41} \simeq 10^{14} \Lambda_B^{-3} \eta_B^{1/2} \left(\frac{\dot{M}}{10^{-4} M_{\odot} \text{yr}^{-1}}\right)^{11/10} \left(\frac{u_1}{10^3 \text{ kms}^{-1}}\right)^{37/10} \left(\frac{\rho_0}{m_p \text{ cm}^{-3}}\right)^{-3/5} \left(\frac{t_{age}}{10 \text{ Myr}}\right)^{4/5} \left(\frac{L_{inj}}{2 \text{ pc}}\right)^{-2} \text{eV}$$
(1.82)

$$E_{max}^{Kra} \simeq 4 \times 10^{14} \Lambda_B^{-2} \eta_B^{1/2} \left( \frac{\dot{M}}{10^{-4} M_{\odot} \text{yr}^{-1}} \right)^{4/5} \left( \frac{u_1}{10^3 \text{ kms}^{-1}} \right)^{13/5} \left( \frac{\rho_0}{m_p \text{ cm}^{-3}} \right)^{-3/10} \left( \frac{t_{age}}{10 \text{ Myr}} \right)^{2/5} \left( \frac{L_{inj}}{2 \text{ pc}} \right)^{-1} \text{eV}$$
(1.83)

$$E_{max}^{Bohm} \simeq 7.53 \times 10^{15} \Lambda_B^{-1} \eta_B^{1/2} \left(\frac{\dot{M}}{10^{-4} M_{\odot} \text{yr}^{-1}}\right)^{1/2} \left(\frac{u_1}{10^3 \text{ kms}^{-1}}\right)^{3/2} \text{eV}$$
(1.84)

As we will show in § 3.1.2,  $u_1$  is expected to be of the order of the speed of the winds from the most massive stars, which ranges between 2000–3000 km s<sup>-1</sup>. This implies that energies of a few PeV are easily reached in the case of protons. For the sake of completeness, it must be noted that  $E_{max}$  depends on two parameters, namely  $\eta_B$  and  $L_{inj}$ that are observationally hard to estimate and are currently largely unknown. For example,  $L_{inj}$  can vary by one order of magnitude, depending on whether the turbulence is injected at a characteristic scale of the average distance between stars (a few pc) or at the typical length scale of the TS size ( $\sim 10 - 20$  pc), and this has a significant impact in the cases of Kraichnan and Kolmogorov-like cascades. For the Bohm case, the parameter  $L_{inj}$  is instead substituted by  $\Lambda_B$ , which accounts for the interval in length scales for which the power spectrum is flat.

Finally, knowing the relation between  $u_1$  and  $L_w$  given by Eq. 1.1, one can rewrite the equations for the maximum energy in a more handy form, which directly depends on  $L_w$ . Assuming that  $\eta_B \ll 1$ , Eqs. 1.82 – 1.83 – 1.84 becomes:

$$E_{max}^{K41} \simeq 1.2\Lambda_B^{-3} \left(\frac{\eta_B}{0.1}\right)^{1/2} \left(\frac{\dot{M}}{10^{-4} M_{\odot} \text{yr}^{-1}}\right)^{-3/4} \left(\frac{L_w}{10^{39} \text{ ergs}^{-1}}\right)^{37/20} \left(\frac{\rho_0}{20m_p \text{ cm}^{-3}}\right)^{-3/5} \\ \left(\frac{t_{age}}{3 \text{ Myr}}\right)^{4/5} \left(\frac{L_{inj}}{2 \text{ pc}}\right)^{-2} \text{PeV}$$
(1.85)

$$E_{max}^{Kra} \simeq 2.8\Lambda_B^{-2} \left(\frac{\eta_B}{0.1}\right)^{1/2} \left(\frac{\dot{M}}{10^{-4} M_{\odot} \text{yr}^{-1}}\right)^{-5/10} \left(\frac{L_w}{10^{39} \text{ ergs}^{-1}}\right)^{13/10} \left(\frac{\rho_0}{20m_p \text{ cm}^{-3}}\right)^{-3/10} \left(\frac{t_{age}}{3 \text{ Myr}}\right)^{2/5} \left(\frac{L_{inj}}{2 \text{ pc}}\right)^{-1} \text{PeV}$$
(1.86)

$$E_{max}^{Bohm} \simeq 10.07 \Lambda_B^{-1} \left(\frac{\eta_B}{0.1}\right)^{1/2} \left(\frac{\dot{M}}{10^{-4} M_{\odot} \text{yr}^{-1}}\right)^{-1/4} \left(\frac{L_w}{10^{39} \text{ ergs}^{-1}}\right)^{3/4} \text{PeV}$$
(1.87)

## 1.4 Probing YMSCs as CRs accelerators

In § 1.3 we acknowledged that, from the the theory point of view, YMSCs are able to accelerate CRs in different ways. Moreover, the maximum energies achieved in these systems can be as high as a few PeV, making YMSCs conceivable as galactic PeVatron. The general question that one may ask is how to empirically confirm or reject the possibility that of YMSCs actually produce a sizable amount of CRs and accelerate particle up to  $10^{15}$  eV.

Unfortunately, it is impossible to probe CR sources directly from the reconstruction of the trajectories of particles arriving at the Earth, since CR propagation in the Galaxy is fully diffusive at 1 PeV. The validity of this statement is readily proven by computing the Larmor radius of a proton in the average Galactic magnetic field ( $\sim 3 \mu$ G):

$$r_L \simeq 0.36 \left(\frac{E}{1 \,\mathrm{PeV}}\right) \left(\frac{B}{3 \,\mu\mathrm{G}}\right)^{-1} \,\mathrm{pc},$$
 (1.88)

which means that after a few pc from the source, a CR has deviated significantly from its original escape trajectory.

The study of CR accelerators must then rely on observational techniques that can probe the presence of accelerated particles by means of their interaction with the environment close to the CR source. One possibility is to consider the radiation emitted by CRs. High-energy photons within the  $\gamma$ -ray range can serve as a direct indicator of the presence of CRs. This is because only energetic particles, via non-thermal processes, can emit radiation with energies higher than a few tens MeV.

Indeed, during the last decades, several YMSCs have been observed in coincidence with large diffuse  $\gamma$ -ray emission, both in the high-energy ( $\gtrsim 1$  GeV) and very highenergy ( $\gtrsim 1$  TeV) bands. Examples are Westerlund 1 (Abramowski et al., 2012; Aharonian et al., 2022), Westerlund 2 (Yang et al., 2018), Cygnus OB2(Ackermann et al., 2011b; Bartoli et al., 2014b; Abeysekara et al., 2021b), and NGC 3603 (Saha et al., 2020). The presence of  $\gamma$ -ray emission has largely strengthened the hypothesis of YMSC as CRs factories. This is even more true considering that in all the detected YMSCs, the observed  $\gamma$ -ray luminosity, if interpreted as hadronic, is easily explained assuming that a fraction of a few percent of the cluster wind power ends up in accelerated particles.

Let us consider, for instance, the specific case of the YMSC Cygnus OB2. A haronian et al. (2019) measured a total  $\gamma$ -ray luminosity towards Cygnus OB2 of  $L_{\gamma} \approx 3.02 \times 10^{34}$  erg s<sup>-1</sup> for photon energies above 10 GeV. The total energy  $W_p$  in terms of hadronic particles to explain such emission can be estimated as (Aharonian et al., 2019):

$$W_p \simeq 2.55 \times 10^{50} \left(\frac{L_{\gamma}}{10^{34} \,\mathrm{ergs}^{-1}}\right) \left(\frac{n_2}{0.1 \,\mathrm{cm}^{-3}}\right)^{-1} \,\mathrm{erg},$$
 (1.89)

which is  $\approx 0.7 \times 10^{51}$  erg for Cygnus OB2. Given the wind power and the age of Cygnus OB2 of  $L_w \approx 2 \times 10^{38}$  erg s<sup>-1</sup> and  $t_{age} \approx 3$  Myr respectively (see § 2.1), the total injected energy by the stellar cluster is  $E_{cluster} = L_w t_{age} \approx 2 \times 10^{52}$  erg, that implies a reasonable efficiency of CR production of the order of  $\sim 10\%$  to account for the observed emission. For the sake of completeness, it must be underlined that this estimation has been carried out assuming that the observed  $\gamma$ -ray emission was of hadronic nature. In principle, also leptons may generate high-energy radiation, and, often, discriminating the nature of the emission is a challenging task. Nevertheless, a comprehensive study of the morphology and spectrum of the  $\gamma$ -ray emission from a YMSC can provide essential insight into the properties of the freshly accelerated CRs, revealing the main characteristics of the ongoing particle acceleration processes in such systems. In § A.1 a general overview of the principal leptonic and hadronic  $\gamma$ -ray emission processes is reported.

Another way to probe the presence of energetic particles is to observe the enhanced ionization degree induced by the low-energy tail of the CR population in dense regions of the ISM close to a CR source. At the beginning of § 1.3, we emphasized the importance of low-energy CRs as regulators of the molecular cloud dynamics and in general of the star formation process.

A diffuse HI cloud which is embedded in the Galactic CR sea is foreseen to have a ionization rate of  $\zeta_{HI} \gtrsim 6.8 \times 10^{-18} \text{ s}^{-1}$  (Spitzer & Tomasko, 1968), while for dense molecular cloud the ionization rate is instead expected to be  $\zeta_{H2} \gtrsim 1 \times 10^{-17} \text{ s}^{-1}$  (Glassgold & Langer, 1974). Close to a CR source, the flux of ionizing CRs is likely to be higher, and one could consider searching for an enhanced ionization rate in cloud close to the acceleration site to confirm the presence of freshly accelerated particles.

This technique has been successfully used in the case of SNRs. A well known case is that of W28, where the presence of an increased ionization rate in nearby molecular clouds was found, in agreement with the idea of SNRs as CR accelerators (Vaupré et al., 2014). Interestingly, the information on the ionization rate can be further combined with  $\gamma$ -ray observations, to eventually help to discriminate the hadronic nature of the emission (Gabici & Montmerle, 2015).

So far, no attempt to extend this approach to YMSC has been made, in spite of the fact that the environment is foreseen to be particularly promising given the large amount of clumped molecular gas that is expected close to a YMSC as a result of the fragmentation of the swept-up shell. See § A.2, for a review on how to calculate the ionization rate from a population of CRs and how to assess its value from observations based on the detection

of specific molecular lines.

## CHAPTER 2

## THE SCIENTIFIC CASE OF CYGNUS OB2

HE understanding of YMSCs as particle accelerators must necessarily pass, to a large extent through  $\gamma$ -ray observations, as their investigation through direct detection of CRs is severely limited by the diffusive behavior of the accelerated particles, as is the case for all galactic sources. Currently, a handful of YMSCs have been found in coincidence with extended  $\gamma$ -ray emission, including Cygnus OB2.

Cygnus OB2 represents, perhaps, one of the most intriguing cases for 3 main reasons. First, the detected diffuse  $\gamma$ -ray emission has been widely investigated from both the spectral and morphological point of view in both the high-energy ( $1 < E_{\gamma} < 100 \text{ GeV}$ ) and very-high-energy bands ( $0.1 < E_{\gamma} < 100 \text{ TeV}$ ), a fact that allows a robust modelization of the underlying CR distribution, potentially leading to an exhaustive knowledge of the acceleration properties. Secondly, the stellar population of Cygnus OB2 has been extensively studied over the decades, paving the ground for a realistic estimate of fundamental stellar cluster parameters such as the total wind luminosity and the mass loss rate. Last but not least, the recent detection by the LHAASO experiment (Cao et al., 2021) of a 1.4 PeV  $\gamma$ -ray in coincidence with the star cluster could indicate the presence of CRs with energies of at least 10 PeV, marking Cygnus OB2 an excellent PeVatron candidate.

In this chapter, we aim to interpret the observed  $\gamma$ -ray emission assuming an underlying distribution of CRs described by the model of particle acceleration at the cluster wind TS developed by Morlino et al. (2021) summarized in § 1.3.3. The information given by the combination of the morphological and spectral shapes of the extended  $\gamma$  emission can be used to constrain, at some level, the propagation mechanism of CRs. This is particularly critical as CR propagation close to the acceleration site is directly related to the type of plasma turbulence in the system, which in turn affects the maximum particle energy achievable by the accelerator. The chapter is structured as follows: in the first part, we will review the main properties of the YMSC Cygnus OB2, and we will estimate the mass loss rate and wind luminosity considering the population of stars in the cluster. In the second part, knowing the properties of the star cluster, we comment on the morphological properties of the freshly accelerated particle distribution. We furthermore describe how to evaluate the  $\gamma$ -ray flux under the assumption of pure hadronic emission. In the last part, we compare the spectro-morphological properties of the expected  $\gamma$ -ray emission with available observations from different experiments. Finally, we comment on the obtained results, discussing on the model validity and its limitations.

## 2.1 The young massive star cluster Cygnus OB2

Cygnus OB2 (Cyg OB2) is one of the most massive and compact OB associations in the Milky Way, located towards the center of the Cygnus-X star-forming complex ( $l \approx 80.22^{\circ}$ ,  $b\approx 0.79^{\circ}$ ), an extended (~10°) radio structure hosting numerous molecular clouds (Schneider et al., 2006), HII regions (Dickel et al., 1969) and several other OB associations (Uyanıker et al., 2001). Cyg OB2 harbor hundreds, possibly thousands of massive stars. The first study of its population has been carried out by Reddish et al. (1966), who inferred with large uncertainties a total of 400-3000 OB stars, based on star counts on the Palomar Sky Survey plates. Similarly, Knödlseder (2000) found a compatible result using star counts in the near-infrared, estimating a total population of  $2600\pm400$  OB stars, with  $120\pm20$ being O-type stars. However, because of the problematic background subtraction and highly patchy extinction pattern towards the association, the amount of stars in Cyg OB2 is possibly lower, as noted by Wright et al. (2010), who estimated a total star content of  $\sim$ 1200 OB stars, with  $\sim$ 75 O-type stars. The radial stellar distribution from observations seems to follow a compact and peaked profile, with a high stellar density in the core of the association, similar to the YMSCs observed in the Large Magellanic Cloud (Knödlseder, 2000). Due to the peaked morphology of Cyg OB2, a large fraction of the stars is enclosed in the central part of the association, in a region with a radius of  $\sim 14$  pc. A recent census of this central core has revealed the presence of 169 OB stars, of which 52 are O-type, and 3 are Wolf-Rayet stars (Wright et al., 2015).

Several estimations of the age of Cyg OB2 have been made through the years. The presence of O-type dwarf stars and high-luminosity blue supergiants in the sample of 85 OB stars selected by Hanson (2003) suggests that Cyg OB2 should not be older than a few Myr, with a likely value of  $2\pm 1$  Myr. An investigation of the population of A-type stars in Cyg OB2 indicated the presence of a group of 5–7 Myr old stars, located mainly in the southern part of the association (Drew et al., 2008). In parallel, X-ray analysis of low-

mass stars seems to point to an age of 3–5 Myr (Wright et al., 2010). Wright et al. (2015) found a typical age of 2–3 Myr and 4–5 Myr by applying respectively non-rotating and rotating evolutionary stellar models to the selected sample of 169 OB stars. The latter results seem to agree with a scenario describing an overall continuous star formation activity, starting  $\sim$ 7 Myr ago and going on until  $\sim$ 1 Myr ago, with a possible peak of star formation around 4–5 Myr. This is compatible with the results found by Comerón & Pasquali (2012), that were also suggestive of a continuous star-forming activity in the region for the last 10 Myr.

The distance of Cyg OB2 is a subject still under debate in the community. Right after the discovery of the association, Johnson & Morgan (1954) measured the distance of Cyg OB2 using spectroscopic observations of 11 stars, finding a value of  $\sim$ 1500 pc. In the first comprehensive investigation of the Cyg OB2 population, Reddish et al. (1966) estimated a distance of 2100 pc. Independent studies in the early 90s based on the method of spectroscopic parallax resulted in a distance of  $\sim$ 1700 pc (Torres-Dodgen et al., 1991; Massey & Thompson, 1991). Perhaps the most commonly adopted value, at present, is the one measured by Hanson (2003), who inferred a distance of  $1400\pm80$  pc after analyzing the absolute magnitude and extinction of 14 OB stars. This measure is reasonably compatible with the position of some molecular clouds in the Cygnus-X region, whose distance has been calculated using maser parallaxes (Rygl et al., 2012). Moreover, this value is also in agreement with the results of a recent work exploiting parallax based distances calculated using eclipsing binaries, from which a distance of  $1330 \pm 60$  pc was evaluated (Kiminki et al., 2015). Finally, a comprehensive study using parallaxes from the second data release by Gaia seems to point out that the association is actually composed of two main subgroups, the first located at a distance of  $\sim$ 1350 pc and the second at  $\sim$ 1755 pc (Berlanas et al., 2019).

### 2.1.1 Cygnus OB2 wind luminosity and mass loss rate

As we will detail later in this chapter, two fundamental parameters regulating multiple aspects of the CR distribution properties are the cluster mass loss rate and the cluster wind luminosity. In order to calculate these parameters for Cyg OB2, we need to compute the mass loss rate  $\dot{M}_i$  for every i-th member of Cyg OB2. We then consider the stars at the core of the association belonging to the sample studied by Wright et al. (2015). We use two different recipes to calculate  $\dot{M}_i$ . The first one is a theoretical formula given by Yungelson et al. (2008)

$$\dot{M}_{i} = \frac{L_{i}}{v_{\infty,i}c} \frac{1}{(1-\Gamma)^{(\alpha-0.5)}}$$
(2.1)

where  $L_i$  is the bolometric stellar luminosity of the i-th star of the sample, c is the speed of light,  $\Gamma = L_i/L_{\text{Edd}}$  with  $L_{\text{Edd}}$  the Eddington luminosity,  $\alpha$ =0.25 (Yungelson et al., 2008), and  $v_{\infty,i}$  is the wind terminal velocity of the i-th star. The latter is defined as (Kudritzki & Puls, 2000):

$$v_{\infty,i} = C(T_{\text{eff}}) \left[ \frac{2GM_i(1-\Gamma)}{R_{\star,i}} \right]^{1/2}$$
 (2.2)

with  $R_{\star,i} = 0.85 (M_i/M_{\odot})^{0.67} R_{\odot}$  (Demircan & Kahraman, 1991) the stellar radius, G the gravitational constant,  $M_i$  the stellar mass, and  $C(T_{\text{eff}})$  a parameter that depends on the star temperature  $T_{\text{eff}}$ , which is  $C(T_{\text{eff}}) = 2.65$  for  $T_{eff} > 21000$  K (Kudritzki & Puls, 2000).

The second equation we use is an empirical relation valid for stars with temperature<sup>1</sup> 27500 K <  $T_{\text{eff}}^i$  < 50000 K given by Vink et al. (2000):

$$\log_{10} \left( \frac{\dot{M}_i}{M_{\odot} y r^{-1}} \right) = -6.668(80) + 2.210(31) \log_{10} \left( \frac{L_i}{10^5 L_{\odot}} \right)$$
  
-1.339(68)  $\log_{10} \left( \frac{M_i}{30 M_{\odot}} \right) - 1.601(55) \log_{10} \left( \frac{v_{\infty,i}}{2v_{esc}} \right)$   
+1.07(10)  $\log_{10} \left( \frac{T_{\text{eff},i}}{40000 \text{ K}} \right) + 0.85(10) \log_{10} \left( \frac{Z_i}{Z_{\odot}} \right)$  (2.3)

where  $Z_i$  is the stellar metallicity and  $v_{\text{esc}} = \sqrt{2GM_i(1-\Gamma)/R_{\star,i}}$  is the escape velocity. When using equation 2.3 we will assume solar metallicity. For every star in their sample, Wright et al. (2015) provide an estimation of the stellar parameters  $\tilde{L}_i$ ,  $\tilde{T}_{\text{eff},i}$ , and  $\tilde{M}_i$  (we will refer to the quantities estimated by Wright et al. (2015) using the ~ diacritic symbol) together with the associated parameter uncertainty ( $\delta \tilde{L}_i$ ,  $\delta \tilde{T}_{\text{eff},i}$ , and  $\delta \tilde{M}_i$ ).

We compute four diverse estimation of  $\dot{M}$ , using different combinations of the measured parameters  $\tilde{L}_i$ ,  $\tilde{T}_{\text{eff},i}$ , and  $\tilde{M}_i$ . More precisely,

- as a first trial, we calculate the total mass loss rate using all the measured parameters.
- as a second trial, we use only  $\tilde{L}_i$ , while  $T_{\text{eff},i}$  and  $M_i$  are calculated using respectively the Stefan-Boltzmann law and the inverted Luminosity-Mass scaling relation given by Yungelson et al. (2008):

$$T_{\rm eff,i} = \left(\frac{\tilde{L}_i}{4\pi R_{\star,i}^2 \sigma_b}\right)^{1/4} \tag{2.4}$$

<sup>&</sup>lt;sup>1</sup>Note that this condition is respected for all stars in the sample of Wright et al. (2015), whose masses are greater than 20  $M_{\odot}$ 

$$\frac{M_i}{M_{\odot}} = \left[10^{-3.48} \left(\frac{\tilde{L}_i}{L_{\odot}}\right)\right]^{0.75}$$
(2.5)

where  $\sigma_b$  is the Stefan-Boltzmann constant.

- as third trial, we use  $\tilde{L}_i$  and  $\tilde{T}_{\text{eff},i}$ , while  $M_i$  is obtained using equation 2.5.
- lastly, we use  $\tilde{T}_{\text{eff},i}$  and  $\tilde{M}_i$  while  $L_i$  is calculated inverting equation 2.5 (Yungelson et al., 2008).

In each trial, we account for the uncertainties in the measuraments of  $\tilde{L}_i$ ,  $\tilde{T}_{\rm eff,i}$ , and  $\tilde{M}_i$  by utilizing a Montecarlo method. This involves creating  $10^4$  different samples of stars where the measured parameters ( $\tilde{L}_i$ ,  $\tilde{T}_{\rm eff}$ , i, and  $\tilde{M}_i$ ) are randomly fluctuated following a Gaussian distribution with a width equal to the associated error ( $\delta \tilde{L}_i$ ,  $\delta \tilde{T}_{\rm eff,i}$ , and  $\delta \tilde{M}_i$ ). Fig. 2.1a shows the result for the  $\dot{M}$  calculation after considering only single stars (no binary systems) with  $M_i > 20M_{\odot}$  and without the contribution of Wolf-Rayet stars. The value of  $\dot{M}$  lies between  $\sim 0.2 - 0.7 \times 10^{-4} \,\mathrm{M}_{\odot} \,\mathrm{yr}^{-1}$ . To account for the contribution of the three Wolf-Rayet (WR) stars in the sample, we consider an ad-hoc empirical relation reported by Renzo et al. (2017):

$$\log_{10} \left( \frac{\dot{M}_{\text{WR}, i}}{M_{\odot} y r^{-1}} \right) = -11.0 + 1.29(14) \log_{10} \left( \frac{L_{\text{WR}, i}}{10^5 L_{\odot}} \right) + 1.73(42) \log_{10} \left( \frac{Y_{\text{WR}, i}}{Y_{\odot}} \right) + 0.47(09) \log_{10} \left( \frac{Z_{\text{WR}, i}}{Z_{\odot}} \right).$$
(2.6)

Adopting for  $Y_{\text{WR},i}$  (helium fraction) and  $Z_{\text{WR},i}$  (metallicity) the solar values, we find a contribution to the mass loss rate from WR stars of  $\dot{M}_{\text{WR}}(\tilde{L}_{\text{WR},i}) \sim 0.3 \times 10^{-4} \text{ M}_{\odot} \text{yr}^{-1}$ .

The star sample of Wright et al. (2015) includes also 9 known binary systems, whose contribution to  $\dot{M}$  is not straightforward to quantify. However, we can make a rough estimation assuming that the observed luminosity  $\tilde{L}_i$  of the system is equally distributed between the two companions. By doing so, and by following the same approach as for single star systems for the calculation of  $\dot{M}_i$  using only  $\tilde{L}_i$ , we find  $\dot{M}_{\text{Binary}} \sim 0.22 - 0.3 \times 10^{-4} \text{ M}_{\odot} \text{yr}^{-1}$ . By summing up all the contributions, Cyg OB2 mass loss rate should lie in a conservative range of  $\dot{M} \simeq 0.7 - 1.3 \times 10^{-4} \text{ M}_{\odot} \text{yr}^{-1}$ . Once  $\dot{M}_i$  and  $v_{\infty,i}$  are known, it is then possible then to estimate the wind luminosity  $(L_i^w)$  for each star as:

$$L_{i}^{w} = \frac{1}{2} \dot{M}_{i} v_{\infty,i}^{2}.$$
 (2.7)



Figure 2.1: (a) Mass loss rate distribution for Cygnus OB2 considering the sample of stars investigated by Wright et al. (2015). The mass loss rate is estimated using two different recipes, a theoretical one (Yungelson et al., 2008) and an empirical one (Vink et al., 2000), using diverse stars measured quantities (see text). The plot only shows the contribution from single star systems and does not include WR stars. (b) Wind luminosity of Cygnus OB2 inferred using the estimated mass loss rates.

Consequently, the cluster wind luminosity is easily found as:

$$L_w = \sum_i L_i^w \tag{2.8}$$

The contribution of single, non Wolf-Rayet stars, accounting for both the values of  $\dot{M}_i$  inferred using the theoretical and empirical recipes of Yungelson et al. (2008) and Renzo et al. (2017), ranges in the range  $L_w^{NBS} \simeq 0.4 - 1.6 \times 10^{38} \text{ erg s}^{-1}$  (see Fig. 2.1b). In addition to this, we also estimate the wind luminosity contribution from the three Wolf-Rayet stars included in the sample of Wright et al. (2015) assuming an average wind speed of 2500 km s<sup>-1</sup>: this turns out to be  $L_w^{WR} \simeq 0.6 \times 10^{38} \text{ erg s}^{-1}$ . Finally, the contribution of the 9 binary systems using the previously computed mass loss rate, is  $L_w^{BS} \simeq 0.55 - 0.7 \times 10^{38} \text{ erg s}^{-1}$ . Accounting for all contributions, Cyg OB2 wind luminosity should lie between  $L_w \simeq 1.55 - 2.9 \times 10^{38} \text{ erg s}^{-1}$ . It is worth noticing that the obtained value of  $L_w$  is compatible with other estimations done by different authors. For example, Ackermann et al. (2011b) calculate a wind luminosity of  $L_w \simeq 2 - 3 \times 10^{38} \text{ erg/s}$  considering a different sample of stars that include the presence of 17 stars with  $M_\star > 35M_{\odot}$  and 5 Wolf-Rayet stars.

One last point worth mentioning is that the values obtained for the theoretical and empirical prescription for  $\dot{M}$  are approximately in good agreement within a factor of a few, with the trial based on the measured bolometric luminosity becoming the one with the better agreement. There are several possible recipes given in the literature (Renzo et al., 2017), and a discrepancy of a factor 2–3 is reasonably expected.

### 2.1.2 The wind blown bubble of Cygnus OB2

As a young (<7 Myr) and massive stellar cluster, it is reasonable to expect Cyg OB2 to be surrounded by a wind-blown bubble. Over the years, several attempts were made to find this structure: one of the first suggestions was to identify the Cyg OB2 wind bubble with the extended X-ray source known as Cygnus Superbubble (Cash et al., 1980; Uyanıker et al., 2001). However, this hypothesis encountered several problems, such as the fact that Cyg OB2 is significantly offset from the geometrical center of the superbubble. Moreover, various portions of the superbubble itself appeared to be uncorrelated, prompting the idea that the Cygnus Superbubble is actually a combination of several structures along the line of sight (Uyanıker et al., 2001). Another attempt to search for the wind bubble was performed through direct search of the cold shell of dense swept-up material surrounding the expanding hot shocked gas (Lozinskaya et al., 2002). Unfortunately, the unlucky position of Cyg OB2 at Galactic longitudes close to 90° prevents a robust determination of the gas location by using kinematic distances. The detection of

the Cyg OB2 bubble still remains nowadays a matter of debate.

For the sake of curiosity, we can try to estimate the expected dimension of the wind bubble considering the theory of Weaver et al. (1977), introduced in § 1.2.3. If we assume the following reasonable values for the parameters of Cyg OB2:  $L_w = 2 \times 10^{38} \text{ erg s}^{-1}$ (compatible to what has been calculated in § 2.1.1), cluster age of 3 Myr, and  $\rho_0 = 20m_p$ cm<sup>-3</sup>, the size of the forward shock position tracing the location of the swept-up shell of dense material is (Eq. 1.8)  $R_b \approx 86$  pc. Considering a distance of 1.4 kpc, the projected size of the forward shock is of the order of a few degrees in radius. Interestingly, this is slightly larger, but in good agreement with the dimension of the diffuse, filamentary, 21 cm continuum radio emission of the Cygnus-X star-forming region (Fig. 2.2).

Knowing that a significant fraction of the continuum emission is of thermal nature (Xu et al., 2013), the overall filamentary structure observed at 21 cm could be tracing what is left of the former (now fragmented) shell of dense material. Thermal emission could then be produced in this scenario by the ionized HII region trapped within the fragmented dense shell.

Currently, no evidence for a collective wind TS has ever been searched for. It is therefore fair to ask whether Cyg OB2 fulfills the conditions to develop a collective wind. Using the same parameters for the estimation of  $R_b$ , and considering a star cluster mass loss rate  $\dot{M} = 10^{-4} M_{\odot} \text{ yr}^{-1}$ , the TS shock radius can be calculated using Eq. 1.11, from which we obtain  $R_{TS} \approx 13 \text{ pc}$ . This value is similar to the overall size of the Cyg OB2 association core studied by Wright et al. (2015). The half-mass radius is, however, smaller by a factor ~ 0.4 ( $R_{hm} \approx 5.2 \text{ pc}$ ) (Pfalzner, 2009), which is compatible with the half-luminosity radius  $R_{hl} \approx 2.5 \text{ pc}$  obtained from the sample of Wright et al. (2015) (see Fig. 2.3). Since the most massive stars are also the brightest,  $R_{hl}$  should roughly delimit the region containing the most massive stars, i.e., those that contribute most to the creation of the collective cluster wind. Reasonably assuming that the average distance between the stars is a fraction of  $R_{hl}$ , the hypothesis that a collective wind exists becomes rather concrete (see § 1.2.3).





44



Figure 2.3: Cumulative radial distribution of Cyg OB2 stellar luminosity calculated using the star sample of Wright et al. (2015).

## 2.2 The distribution of CRs in Cygnus OB2

Given the compactness of Cyg OB2 and the possibility of developing a wind TS, it is likely that the acceleration mechanism in this system, and the resulting distribution of CRs  $(f_{CR})$ , is the one described in § 1.3.3, appropriate for particles acceleration at the wind TS. In such a scenario, Eq. 1.70 describes the distribution of CRs. Note that  $f_{CR}$  is composed of a contribution of two populations, the freshly accelerated particles escaping from the acceleration site and the population of Galactic CRs that may end up penetrating in the system. Let us now rewrite Eq. 1.70 accounting only for the contribution of freshly accelerated CRs, i. e., considering  $f_{qal} = 0$ :

$$f_{CR}(r,p) = \begin{cases} f_{TS}(p) \cdot exp \left[ -\int_{r}^{R_{TS}} \frac{u_{1}}{D_{1}(r',p)} dr' \right] & \text{for } r \leq R_{TS} \\ f_{TS}(p) e^{\alpha} \frac{1+\beta(e^{\alpha}B^{-\alpha}-1)}{1+\beta(e^{\alpha}B^{-1})} & \text{for } R_{TS} \leq r \leq R_{b} \\ f_{TS}(p) \frac{e^{\alpha}B}{1+\beta(e^{\alpha}B-1)} \frac{R_{b}}{r} & \text{for } r \geq R_{b} \end{cases}$$
(2.9)

where  $\alpha$ ,  $\alpha_B$  and  $\beta$  are defined in Eq. 1.71, and  $f_{TS}$  is given by Eq. 1.75.

Clearly, both the spectral and morphological characteristics of  $f_{CR}$  depend more or less directly on the properties of Cyg OB2, but also on the type of diffusion in the system, which is related to the (unknown) turbulence spectrum in the bubble. We therefore proceed to evaluate Eq. 2.9 in the specific case of Cyg OB2, considering three different cases of turbulence spectrum: Kolmogorov-like, Kraichnan-like and flat spectrum. The following values are used for Cyg OB2:  $L_w = 2 \times 10^{38}$  erg s<sup>-1</sup>,  $\dot{M} = 10^{-4}$  M<sub> $\odot$ </sub> yr<sup>-1</sup>, cluster age of 3 Myr, and  $\rho_0 = 20m_p$  cm<sup>-3</sup> (related to the mean density of the GMC from which Cyg OB2 was formed, unknown in fact and totally hypothetical). We model the plasma turbulence in the three cases considering that the total power in magnetic turbulence is a fraction  $\eta_B = 0.1$  of  $L_w$ , and that the power injection scale of the turbulence is  $L_{inj} = 2$ pc, as it is reasonable to assume it to be of the order of the cluster size  $R_{hl}$ . For the Bohm case, we also assume that the turbulence is injected down to scales of  $k_{min} = 10^{-5}$  pc<sup>-1</sup>. Finally, we additionally assume that a fraction  $\epsilon_{CR} = 0.1$  of  $L_w$  goes into acceleration of CRs, and that the slope of the injected particle spectrum at the TS is s = 4. The results obtained evaluating Eq. 2.9 under these assumptions are shown in Fig. 2.4 and Fig. 2.5, which show respectively the radial shape of  $f_{CR}$  at different energies and the spectra of injected particles at the TS for the three cases under analysis.

By looking at the radial shape of  $f_{CR}$ , one can readily see how the three cases correspond to different spatial profiles: the Kolmogorov turbulence produces a more peaked shape, while a Bohm diffusion induces a flat profile. The Kraichnan case instead, generates a profile that is intermediate between Kolmogorov and Bohm, with a flat distribution at low energies and a peaked profile at energies higher than ~100 TeV. Actually, the morphology is energy-dependent in all the tree models, with an increasingly peaked profile at higher energies. This change in shape is caused by the onset of dominance of diffusive propagation over advection at high energies.

It is possible to find at what energy the two transport processes are equally important by equating the advection time scale  $t_{adv}$ , with the diffusion time  $\tau_d$ , defined as:

$$t_{adv} = \int_{R_{TS}}^{R_b} \frac{dr}{u_2(r)}$$
(2.10)

$$\tau_d(E) = \frac{(R_b - R_{TS})^2}{2D_2(E)},\tag{2.11}$$

where the advection timescale is obtained under the assumption of a strong shock considering a velocity in the downstream of  $u_2(r) = (u_1/4)(r/R_{TS})^{-2}$ . Using the standard parameters for Cygnus OB2, we find that  $t_{adv} \approx 1.8$  Myr. In Eq. 2.11, the diffusion coefficients for the three cases under analysis ( $D_2$ ) are given by Eqs. 1.78 – 1.79 – 1.80. We calculate them considering the turbulent magnetic field in the downstream, that is  $\delta B_2 = \sqrt{11} \delta B_1(R_{TS})$ , with  $\delta B_1$  given by Eq. 1.81. Fig. 2.6 shows the energy at which  $t_{adv} = \tau_d(E)$ . This energy is ~20 GeV, ~2 TeV, and ~100 TeV for Kolmogorov, Kraichnan, and Bohm respectively. It is crucial to underline that these numbers are average values. If one takes into account the velocity profile, the effect of advection should be stronger (i.e., shifting to higher energies the former values) at distances closer to the TS. This can



Figure 2.4: Radial CR distribution normalized to its value at the termination shock in the case of Kolmogorov (top panel), Kraichnan (central panel), and Bohm (bottom panel) turbulence. In these plots, the contribution of the galactic CR sea to  $f_{CR}$  has not been considered.



Figure 2.5: Injected particles spectrum at the TS with same cluster and turbulence injection properties but with different diffusion coefficient in the system.

be clearly seen in Fig. 2.4, where the profile becomes flatter close to the TS.

In addition to the propagation properties, the diffusion coefficient also regulates the confinement of particles around the acceleration site; hence, it directly affects the maximum energies that CRs can reach, and the spectral shape of injected particles. Both effects are noticeable in Fig. 2.5: in spite of begin calculated with the same parameters for Cyg OB2, the particle spectra at the TS present very different shapes of the cut-off, and the position itself of the cut-off is different for the three considered diffusion regimes. Kolmogorov turbulence, for example, is not very efficient in confining particles close to the wind TS. As a consequence, the maximum energy will be less if compared to a harder turbulent cascade spectrum, as for example the flat spectrum behind Bohm-like diffusion.

## 2.3 Modeling $\gamma$ -ray emission from Cygnus OB2

During the escape from the acceleration site, the hadronic component of CRs is expected to interact with the surrounding medium causing the creation of neutral pions. As described in § A.1, the  $\pi^0$ s subsequently decay with the emission of  $\gamma$ -rays. The observed spectrum and morphology will strongly depend on both the distribution of CRs and the target medium. The  $\gamma$ -ray flux from  $\pi^0$  production is described by Eq. 2.

In our specific case,  $f_{CR}(E_p, r)$  is the radial distribution of CRs given by Eq. 2.9, while n(r) represents the number density distribution of the target medium in the vicinity of



Figure 2.6: Comparison between diffusion and advection time scales as a function of the particle energy. For showing purpose, we also included the diffusion time scale obtained considering the standard diffusion coefficient in the ISM (dotted line) (Strong et al., 2007).

Cyg OB2, which is largely unknown and must be assumed a priori (see § 2.3.1). Having in mind the spherical geometry of the system, Eq.2 can be simplified by considering the volume integral in terms of cylindrical coordinates. Knowing that  $r = \sqrt{\ell'^2 + z^2}$ , where *z* is the direction along the line of sight and  $\ell'$  the projected distance on the sky, Eq. 2 can be rewritten as

$$\phi_{\gamma}(\ell, \Delta \ell, E_{\gamma}) = \int_{\ell}^{\ell+\Delta \ell} \overline{n}(\ell')\xi(\ell', E_{\gamma})\ell'd\ell', \qquad (2.12)$$

where  $\overline{n}(\ell')$  is the average ISM density profile that can be inferred from observations (see § 2.3.1) and  $\xi(\ell', E_{\gamma})$  defined as

$$\xi(\ell', E_{\gamma}) = \frac{1}{d^2} \iint cf_{CR}(\ell', z, E_p) \frac{d\sigma(E_p, E_{\gamma})}{dE_p} dE_p dz.$$
(2.13)

By varying the limits of integration, Eq. 2.12 can be used to estimate the total  $\gamma$ -ray flux from a particular area of the sky (for spectral analysis) and also to obtain the  $\gamma$ -ray radial profile (for morphological analysis).

### 2.3.1 The interstellar medium close to Cygnus OB2

The distribution of the ISM is a crucial parameter that directly affects the morphology of the observed  $\gamma$ -ray emission. On a very general ground, the ISM can be divided into three

main gas phases: ionized, neutral atomic gas, and molecular gas. The latter two phases are typically the densest and the most massive, and hence are the ones that contribute the most to the production of  $\gamma$ -ray emission. The neutral atomic gas is usually observed using the emission of the 21 cm line of HI, while molecular H<sub>2</sub> must be traced indirectly using <sup>12</sup>CO, since direct observation of H<sub>2</sub> is not possible due to the symmetry of the molecular structure which forbids dipole emission.

Clearly, due to the projection effect, not all the molecular or atomic gas observed in the vicinity of a given source is actually in its vicinity. The location of the gas along the line of sight can be estimated using the method of kinematic distances (Roman-Duval et al., 2009). Kinematic distances are based on the Doppler shift of a certain transition (atomic or molecular) induced by Galactic differential rotation. Knowing the Galactic rotation curve, a specific Doppler shift value identifies three points along a given line of sight with Galactic longitude  $\theta$ . These three points correspond to three different positions returning the same projected radial velocity  $V_r$  (see Fig. 2.7), and are known as Near distance, Tangent points, and Far distance. The distance (*d*) of these point can be calculated as:

$$d = R_0 \cos(\theta) \pm \sqrt{r_{gal}^2 - R_0^2 \sin(\theta)}$$
 (2.14)

where  $R_0$  is the Galactocentric distance of the Sun,  $V_0$  is the Sun orbital velocity, and  $r_{gal}$  is the Galactocentric distance of the gas:

$$r_{gal} = R_0 \sin(\theta) \frac{V(r_{gal})}{V_r + V_0 \sin(\theta)}$$
(2.15)

with  $V(r_{gal})$  as the Galactic rotation curve. In Eq. 2.14, the tangent point is obtained for  $r_{gal} = R_0 \sin(\theta)$ , and so  $d = R_0 \cos(\theta)$ . It is impossible to discern between near, far and tangent points based on kinematics alone. This issue is generally known as *kinematic distance ambiguity*. In general, additional pieces of information are required to remove the ambiguity, as for example, dust absorption or parallax measures. Note that this problem exists only for  $r_{gal} < R_0$ , as in the outer Galaxy the radial velocity along the line of sight decreases monotonically.

In principle, to model the gas distribution near Cyg OB2, one could use specific kinematic cut on the gas velocity. Unfortunately, the Cygnus-X region is located at Galactic longitudes where the differential galactic rotation up to  $\sim$ 4 kpc results in low radial velocities, with values compatible with the typical gas motion dispersion, thus preventing a robust 3D modelization of the ISM profile along the line of sight. Even if the kinematic ambiguity prevents a small-scale spatial modelization of the gas, we consider anyway a velocity cut between -20 km s<sup>-1</sup> and 20 km s<sup>-1</sup>, selecting in such a way the gas associated with the Cygnus-X star forming complex (Schneider et al., 2006) while removing



Figure 2.7: Scheme for calculating the kinematic gas distance. The Doppler shift of an atomic or molecular transition depends on the component of the gas velocity due to the galactic revolution  $(v_c)$  projected along the line of sight  $(v_r)$ . This produces an ambiguity in position (near and far position, N and F points respectively). The case where the radial velocity of the gas is totally aligned with the direction of the line of sight defines the tangent point (T point).

the gas contribution of both the Perseus and Outer arms. Once the gas is kinematically selected, we assume the most straightforward case where all the observed gas is uniformly distributed along the line of sight in a range  $\Delta z = \pm 400$  pc around Cyg OB2 position. The choice of  $\Delta z$  stems from the fact that the total extent of 800 pc is inferred from the distribution of dust towards the Cygnus-X star forming complex (Green et al., 2019). Note that this specific gas model is perfectly consistent with the expression for the  $\gamma$ -ray emission in Eq.2.12, which was derived by implicitly assuming a constant gas profile along the line of sight.

To quantify the amount of neutral hydrogen, we use 21 cm line data from the Canadian Galactic Plane Survey (CGPS) (Taylor et al., 2003). For the molecular component, we use high-resolution observations of  $^{12}$ CO J(1–0) spectral line from the Nobeyama radio telescope (Takekoshi et al., 2019) in combination with the data from the composite galactic survey of Dame et al. (2001). The neutral hydrogen column density is estimated using the approach described by Wilson et al. (2009):

$$\left[\frac{N_{\rm HI}}{\rm cm^{-2}}\right] = -1.8224 \times 10^{18} \left[\frac{T_s}{\rm K}\right] \int_{-20}^{20} \log\left(1 - \frac{T_B^{\rm HI}(v)}{T_s - T_{\rm BG}}\right) \left[\frac{dv}{\rm km/s}\right],\tag{2.16}$$

where  $T_B^{\text{HI}}$  is the observed line brightness temperature,  $T_s$  is the spin temperature, assumed to be 150 K, and  $T_{BG} = 2.66$  K is the brightness temperature of the cosmic mi-



Figure 2.8: Total target column density considering the interstellar medium in the neutral and molecular phases. The column density of HI has been evaluated using the 21 cm line observation from the CGPS, while for the molecular gas, we used a combination of <sup>12</sup>CO observation from the Nobeyama radio telescope and the data from the <sup>12</sup>CO galactic plane survey by Dame et al. (2001). All the observations are kinematically cut selecting only the gas between -20 and 20 km/s. White rings represent the regions used by Aharonian et al. (2019) and Abeysekara et al. (2021b) to build the  $\gamma$ -ray radial profile. The circle radii correspond to angular sizes of 0.61°, 1.19°, 1.8° and 2.21°.

crowave background at 21 cm. For the molecular hydrogen, we calculate the column density using the standard  $X_{CO}$  conversion factor:

$$N_{\rm H_2} = X_{\rm CO} \int_{-20}^{20} T_B^{\rm CO}(v) \left[\frac{dv}{\rm km/s}\right],$$
(2.17)

with  $X_{\rm CO} = 1.68 \times 10^{20}$  mol. cm<sup>-2</sup> km<sup>-1</sup> s K<sup>-1</sup> as found by Ackermann et al. (2011b). Finally, we can write the total target column density as  $n(r) = (N_{\rm HI} + 2N_{\rm H_2})/\Delta z$  (see fig. 2.8).

# 2.4 Comparison between expected and observed $\gamma$ -ray emission

To test the validity of the model described in § 1.3.3, we compare the spectral and morphological properties of the expected  $\gamma$ -ray emission obtained in § 2.3 with currently available observations by several experiments. The general idea is to find the best values of the parameters in terms of  $L_w$ , s, and  $\epsilon_{CR}$  that can adequately describe the observed  $\gamma$ -ray spectrum. Then, a posteriori, we check if the best values found are reasonable, by comparing for example the obtained wind luminosity with the estimates in § 2.1.1. For the sake of simplicity, when not specified otherwise, we keep all other parameters describing Cyg OB2 (i.e. age,  $\dot{M}$ , distance of Cyg OB2,  $\rho_0$ ,  $\eta_B$ ,  $L_{inj}$ ,  $k_{min}$ ) fixed to the values used in § 2.2.

In the search for the best parameters, we use the following approach. We fit through  $\chi^2$  minimization the observed spectral energy distribution extracted from a region of 2.2° centered on the stellar cluster (corresponding to a projected radius of ~ 54 pc). We define the  $\chi^2$  as:

$$\chi^{2} = \sum_{k} \frac{[(E_{k}^{2}\phi_{\gamma,k})_{data} - (E_{k}^{2}\phi_{\gamma}(E_{k}))_{model}]^{2}}{\sigma_{k}^{2}}$$
(2.18)

where k is the index associated to the k-th spectral point, and  $\sigma_k$  is the error of the k-th spectral energy distribution point. During the fit procedure, the parameters are left free to vary in a range of  $10^{37} - 5 \times 10^{39}$  erg s<sup>-1</sup> for  $L_w$ , 1.8 - 2.6 for s and  $10^{-3} - 10^{-1}$  for  $\epsilon_{CR}$ . In terms of datasets, we consider the Cygnus Cocoon flux points measured by Fermi-LAT in the 4FGL (4FGL J2028.6+4110e) (Abdollahi et al., 2020), and the very-high-energy observations carried out by ARGO (ARGO J2031+4157) (Bartoli et al., 2014b) and HAWC (HAWC J2030+409) (Abeysekara et al., 2021b).

All the employed spectral points are rescaled in order to account only for the observed flux coming from a region of 2.2°. This is done by considering that in all the cases the emission towards Cyg OB2 is modeled using a 2D symmetric Gaussian profile with different sizes: 2.0° for 4FGL J2028.6+4110e, 1.8° for ARGO J2031+4157 and 2.13° for HAWC J2030+409, which leads to rescaling factors of 0.45, 0.53 and 0.41 respectively. In addition, for the calculation of the  $\chi^2$ , we do not account for the highest energy datapoint by HAWC, as it differs, by almost one order of magnitude from the flux measured by LHAASO (LHAASO J2032+4102) at 100 TeV (Cao et al., 2021). The flux measured by LHAASO J2032+4102 is also not considered as the size of the source is not provided, thus the flux cannot be rescaled accordingly to the analyzed sky region size. Finally, note that the expected  $\gamma$ -ray emission is calculated using Eq. 2.9, hence, without accounting for the contribution of  $f_{gal}$ . This is because in principle, all the employed flux points should

be background subtracted. In general, the background emission by Galactic CRs is nonnegligible in the Fermi-LAT band. However, the Fermi-LAT data does not include this contribution. This is because, in the analysis procedure, a template model that follows the gas distribution is fitted along with the source model to account for the background emission.

Before proceeding further, a note of caution is mandatory. The implemented analysis approach is relatively naive, so the final best fit value should be handled with care. This is even more true for the associated confidence intervals. Clearly, to obtain a precise estimation of the parameters, a robust multi-instrument joint analysis accounting for the systematics between different experiments is required. However, even with this simplified approach, we can still obtain a rough estimation of our parameters of interest. We do not however return confidence intervals for the parameters, as they could be significantly affected by the systematics between the experiments.

Once we obtain the CR distribution that best describes the observed spectrum, we investigate the corresponding expected  $\gamma$ -ray radial profile to understand if one specific type of propagation model among the ones implemented can best reproduce the observed morphology. For this purpose, we calculate the total  $\gamma$ -ray luminosity in four different rings centered on Cyg OB2, with projected sizes of 0–15 pc, 15–29 pc, 29–44 pc, and 44–54 pc (see Fig. 2.8). The  $\gamma$ -ray luminosity is defined as:

$$L_{\gamma} = 4\pi d_{OB2}^2 \int_{E_-}^{E_+} E_{\gamma} \phi_{\gamma}(E_{\gamma}) dE_{\gamma}$$
(2.19)

where  $d_{OB2}$  is the distance of Cyg OB2 and  $\phi_{\gamma}(E_{\gamma})$  is the spatial integrated flux from eq. 2.12. we then divide the obtained value of the luminosity by the rings surfaces, and compare the result with the value estimated for the same sky regions by Aharonian et al. (2019) using Fermi-LAT data and by HAWC (Abeysekara et al., 2021b). Consequently, we set the limit of integration in Eq. 2.19 to account only for the luminosity in the energy ranges for which the measurements refer, that are respectively  $E_{-} = 10$  and  $E_{+} = 300$ GeV for the Fermi-LAT band and  $E_{-} = 1$  and  $E_{+} = 250$  TeV for the HAWC observations.

In the following subsections, we will separately discuss the analysis outcomes for the three different transport models implemented, then, we will proceed to discuss the implications of the obtained results in the next section.

### 2.4.1 Kolmogorov case

We start by studying the case in which the CR distribution is calculated assuming a Kolmogorov like diffusion. We found that  $\chi^2$  as a function of  $L_w$  has an overall decreasing trend with increasing  $L_w$  in the considered wind luminosity interval, with no clear sign of a minimum. This trivially translates into a global  $\chi^2$  minimum for  $L_w = 5 \times 10^{39}$  erg s<sup>-1</sup>. With this value of  $L_w$ , the best fit values for the spectral index of injected particles and the efficiency of CR production are  $s \simeq 4.17$  and  $\epsilon_{CR} \simeq 4 \times 10^{-3}$ . Although the position of the global  $\chi^2$  minimum does not fall in the considered luminosity interval, the value of  $L_w = 5 \times 10^{39} \text{ erg s}^{-1}$  should not be far from the true minimum position, which is probably a factor of a few higher. This can be seen in the top panel of Fig. 2.9, where the spectrum corresponding to  $L_w = 5 \times 10^{39}$  erg s<sup>-1</sup> describes fairly well the observations. It follows that to adequately reproduce the observed spectrum, a wind luminosity equal or higher than  $5 \times 10^{39}$  erg s<sup>-1</sup> is needed, which is more than one order of magnitude above the range of values of  $L_w$  estimated in § 2.1.1. The reason for such a high luminosity is to be found in a combination of two factors: first, the low efficiency of Kolmogorov turbulence in confining particles at the acceleration site, causes low maximum energies even for high wind luminosities; second, the large diffusion coefficient produced by Kolmogorov turbulence in the bubble is inefficient in trapping particles at high energy, hence suppressing the emission at very-high energy. One can notice these two effects by considering the  $\gamma$ -ray spectrum obtained by fixing the wind luminosity to the value expected from the star population of Cyg OB2,  $L_w = 2 \times 10^{38}$  erg s<sup>-1</sup>, and by fitting s and  $\epsilon_{CR}$ . In this case, the maximum energy of particles is too low, and the expected spectrum fails to reproduce the observed flux at very high-energies.

Even taking in to account the large uncertainties on the estimated wind luminosities in § 2.1.1, such a high  $L_w$  is largely in disagreement with our expectations, a fact that strongly disfavors the model with diffusion resulting from Kolmogorov turbulence.

As a final consideration, it is worth noticing that the maximum reachable energy for  $L_w = 5 \times 10^{39}$  erg/s is  $E_{max} \simeq 23$  PeV. Since the problem is, in part, linked to the value of  $E_{max}$ , one could consider decreasing the required  $L_w$  to obtain such value by considering a higher value for  $\eta_B$ , as  $E_{max}$  is directly proportional to the square root of  $\eta_B$  (see Eq. 1.85). However, even by increasing  $\eta_B$  to an unrealistic value of 50%, the required luminosity only decreases by ~40%, which is still in strong disagreement with the expected values. Given the difficulties of this model at reproducing the observed spectrum, we do not investigate further the properties of the  $\gamma$ -ray emission for the Kolmogorov case.

### 2.4.2 Kraichnan case

Let us now consider the results for the case of Kraichnan turbulence. After running the analysis procedure, we found the global minimum for the  $\chi^2$  for a wind luminosity of  $L_w = 1.28 \times 10^{39}$  erg s<sup>-1</sup>. The corresponding best fit values for the spectral index and the CR efficiency are  $s \simeq 4.23$  and  $\epsilon_{CR} \simeq 7 \times 10^{-3}$ . The required  $L_w$  is lower compared to the Kolmogorov case, but still a factor of ~4 higher than the maximum estimated lumi-



Figure 2.9: Spectral energy distribution of the expected hadronic  $\gamma$ -ray emission extracted from a region of 2.2° centered on Cyg OB2. Results are shown for the Kolmogorov (top panel), Kraichnan (middle panel), and Bohm diffusion (bottom panel). See text for more details.

nosity of  $2.9 \times 10^{38}$  erg s<sup>-1</sup>. Although  $L_w$  is at odds with the estimates made for the wind luminosity, let us anyway proceed with our analysis without discarding the Kraichnan case, and compare the predicted  $\gamma$ -ray radial profile with current observations. The top panel of Fig. 2.10 shows this comparison for both the high-energy and very high-energy bands. Starting from the high-energy band, one can distinctly see that, excluding the inner ring, the normalization of the expected  $\gamma$ -ray luminosity per ring is, on average, a factor of ~2 higher than the one estimated by Aharonian et al. (2019). This discrepancy is caused by the different flux normalization between the 4FGL datapoints (used in our spectral analysis) and the spectrum measured by Aharonian et al. (2019), as shown in the middle panel of Fig. 2.9. Moreover, even if we account for this difference in normalization by rescaling the observed luminosity by a factor of 2, we can see that the predicted profile shape is still not in agreement with the observations. In fact, we expect a flat profile, contrary to observations that seem to show a peaked morphology. The situation is different at very high energy, where the morphology observed by HAWC is in good agreement with our model, which still predicts a flat profile.

#### 2.4.3 Bohm case

Bohm-like turbulence is highly efficient in confining particles around the acceleration site, producing higher maximum energies with small values of  $L_w$ . If we, in fact, fix the wind luminosity value to the usual  $L_w = 2 \times 10^{38} \text{ erg s}^{-1}$ , and we perform the fit varying s and  $\epsilon_{CR}$ , the resulting spectrum is close to the observed one, with the exception of the very high energy part (see bottom panel of Fig. 2.9). Since this wind luminosity provides a reasonable fit to the spectrum, we decided to use a different approach, consisting in fixing the wind luminosity to  $L_w = 2 \times 10^{38} \text{ erg s}^{-1}$  and performing the fit in s,  $\epsilon_{CR}$  and  $\eta_B$ . Following this method, we found  $\eta_B = 0.35$ , with an associated spectral index and CR efficiency of  $s \simeq 4.27$  and  $\epsilon_{CR} \simeq 0.022$ . The corresponding maximum energy for particle is  $E_{max} \simeq 466$  TeV.

The bottom panel of Fig. 2.10 shows the corresponding predicted  $\gamma$ -ray radial profile. The result is analogous to that obtained for the Krainchnan case, with an overall flat trend in both energy bands.

### 2.5 Discussion

Table 2.1 summarizes the results obtained in the analysis for all the considered cases. We now proceed to a deeper analysis of the results related to the Kraichnan and Bohm cases, while the Kolmogorov model was ruled out because of the high wind luminosity needed to reproduce the  $\gamma$ -ray spectrum.



Figure 2.10: Comparison between the expected  $\gamma$ -ray surface brightness radial profile assuming Kraichnan (upper panel) or Bohm (bottom panel) diffusion. Black squares show the measurements by Fermi-LAT (Aharonian et al., 2019), while purple circles indicate the surface brightness observed by HAWC (Abeysekara et al., 2021b). Grey squares represent instead Fermi-LAT measurements rescaled by a factor of 2 (see text). Diamonds and asterisks indicate the expected surface brightness values from the models at energies of 10 GeV<  $E_{\gamma} < 316$  GeV and 1 TeV<  $E_{\gamma} < 251$  TeV respectively.

Models	$L_w$	S	$\epsilon_{CR}$	$\eta_B$	$E_{\rm max}$	$R_{TS}$	$R_b$	$ar{\chi}^2_{ m min}$
	$[\mathrm{erg}\mathrm{s}^{-1}]$		[%]		[PeV]	[pc]	[pc]	
Kolmogorov	$5 \cdot 10^{39}$	4.17	0.4	0.1 (fixed)	23	16	163	0.66
Kraichnan	$1.28\cdot 10^{39}$	4.23	0.7	0.1 (fixed)	3.97	14	124	0.39
Bohm	$2 \cdot 10^{38}$ (fixed)	4.27	2.2	0.35	0.47	13	86	0.25

Table 2.1: Best fit parameters values and main system properties for three different models. For the Kolmogorov and Kraichnan cases, the parameters varied during the fit are  $L_w$ , s, and  $\epsilon_{CR}$ . For the Bohm case we fix the wind luminosity and vary s,  $\epsilon_{CR}$  and  $\eta_B$ .

Note that in principle, the Kraichnan case, similarly to the Kolmogorov model, also requires a wind luminosity value that is inconsistent with the estimates calculated in § 2.1.1. However we keep the Kraichnan case for the purpose of discussion. In fact, the value of  $L_w$  required to reproduce the spectrum decreases as we consider turbulence models with gradually harder spectra, at the cost of increasing by a factor of a few the fraction of power in magnetic turbulence. We may then conclude that the true turbulence spectral index lies between the cases of Bohm and Kraichnan-like cascades. In this scenario, we can keep the two models as limiting cases, bearing in mind that the most realistic scenario, assuming that CR acceleration occurs at the wind TS, will have intermediate properties between the two diffusion models.

The discussion of the results is organized as follows: first, we will comment on the results obtained from the spectral analysis, and later concentrate on the morphological analysis. Finally, we will discuss which, among the considered cases, is the one that can best reproduce the observed emission properties.

### 2.5.1 Discussion on the spectral analysis results

Starting from the spectral analysis results, one can notice how both Kraichnan and Bohm cases require a particle injection slope at the termination shock s > 4. This is softer than the standard spectral index predicted from diffusive shock acceleration at strong shocks, as one would expect the TS of a compact cluster to be. A possible reason for this softening could be related to the plasma cooling in the bubble caused by the heat transmission to the cold shell of swept material (see § 1.2.3). The temperature drop in the downstream can potentially affect the TS, resulting in an effective weakening of the shock itself and causing a softening in the spectra of accelerated particles.

A significant difference between the two turbulence models is the maximum energy

reached by the particles. For the Kraichnan case  $E_{max}^{Kra} \simeq 4$  PeV, which is almost an order of magnitude higher than in the Bohm case,  $E_{max}^{Bohm}\simeq 0.5$  PeV. Interestingly, despite the different maximum energies, the  $\gamma$ -ray spectra of the two cases are remarkably similar, with a cutoff feature that in both cases is located in the range 10-100 TeV. As the maximum energy in the Kraichnan case is higher, one could naively expect the cutoff in the  $\gamma$ ray spectrum to be positioned at energies slightly higher than 100 TeV. This does not happen because of two distinct factors. The first one is related to the different spectral cutoff shapes of the emitting particle spectrum for the two considered cases. As explained in the paper by Morlino et al. (2021), the maximum energy calculated in § 1.3.3 does not represent the exact location of the cutoff of the injected particles spectrum. This can be readily seen by computing  $E_{max}$  for the standard parameters of Cyg OB2 given in § 2.2 and by comparing the results with the cutoff positions shown in Fig. 2.5. The deviation from a power law and the beginning of the cutoff region usually starts at energies lower than  $E_{max}$ . On a general ground, the energy shift is related to the spherical symmetry of the TS, which affects primarily the most energetic particles, whose diffusion length is of the order of the TS radius. The magnitude of the shift directly depends on the diffusion coefficient in the upstream region: the hardest (the softest) the momentum dependence of the diffusion coefficient, the smaller (the larger) the energy shift. The second factor is related to the different propagation properties in the downstream region. Particles propagating in Kraichnan turbulence are not as effectively trapped inside the bubble as they are in Bohm-like turbulence, and as a result, they enter the diffusion regime at lower energies. This leads to a reduction in the number of particles at high energies, causing the  $\gamma$ -ray cutoff to shift towards lower energies.

Noticeably, both the Kraichnan and Bohm cases require a low fraction, of order a few percent, of the wind luminosity to be converted into CR production. One might then think that the process is inefficient in accelerating particles. This is, however, not true: the best fit value found for  $\epsilon_{CR}$  should be considered as a sort of lower limit, with true values potentially higher by a factor of ~ 10. The reason for this lies in the method implemented in § 2.3.1 to model the distribution of the ISM around Cyg OB2, which consists of a uniform density profile along the line of sight. Under the assumption of a constant distribution, the average numerical particle density in the region is ~ 8 cm<sup>-3</sup>. However, the density in the downstream, where most of the  $\gamma$ -ray emission is produced, is expected to be lower.

An estimate of the density can be obtained assuming that the entire bubble is only filled with the material provided by the cluster wind, that is  $n_2 \simeq \dot{M}t/(4/3\pi R_b^3) \approx 0.005$  cm<sup>-3</sup>. This value is extremely low, and in fact, not realistic. As explained in § 1.2.3, due to the heating of the cold swept-up shell, part of the shell material evaporates into the downstream region, causing an increment of the density.

Considering the parameters of Cyg OB2, using Eq. 1.24, we can estimate the density in the bubble as:  $n_2 \simeq \dot{M}_s t/(4/3\pi R_b^3) \approx 0.1 \text{ cm}^{-3}$ . The latter value could be even higher by a factor of a few, if one accounts for the shell fragmentation, that potentially boosts the evaporation rate (Lancaster et al., 2021b). In the end, we can reasonably expect  $n_2$  to lie between 1–0.1 cm<sup>-3</sup>. If these are the density values in the downstream, the observed spectrum would require  $\epsilon_{CR}$  a factor of ~10 larger.

### 2.5.2 Discussion on the morphological analysis results

As briefly mentioned in the previous section, both cases considered show a flat morphology of the  $\gamma$ -ray emission profile, regardless of the energy band. The uniformity of the profile is a result of the advection dominated transport of CRs, making their distribution uniform in space. While such result is not unexpected in the case of Bohm-like turbulence, in the Kraichnan case, one could expect that the diffusion would dominate in the HAWC energy band. This does not happen because we are probing the morphology of the  $\gamma$ -ray emission in a projected sky area of about 54 pc. This region is very small compared to the size of the forward shock, which for our best-fit, in the Kraichnan case, is  $R_b \simeq 126$  pc. As described in § 2.2, since the advection velocity scales as  $u_2 \propto r^{-2}$ , the vicinity of the TS will be characterised by high advection velocities, making it the main propagation mechanism. This can be easily understood by looking at Fig. 2.11, where the timescales of advection and diffusion are compared taking into account the size of the region under analysis. For the Kraichnan case, in a region of 54 pc, advection dominates over diffusion up to energies of almost a hundred TeV. The same is also true for Bohm-like turbulence.

Actually, to be accurate, one must also consider the effect of projection along the line of sight of the  $\gamma$ -ray emission, so that the observed emission does not come only from a spherical volume of radius 54 pc. However, during the calculation of the radial profile shown in Fig. 2.10, this effect is already taken into account. Nevertheless, a flat trend is consistent with the HAWC observation. Yet this is not the case for the radial morphology observed by Fermi-LAT, which shows a peaked profile in the  $\gamma$ -ray luminosity. This is clearly in tension with the prediction of our model, where at lower energy, the effect of advection is expected to become even stronger.

According to Aharonian et al. (2019), the observed  $\gamma$ -ray morphology is consistent with the expected emission generated by a pure diffusive distribution of CRs continuously injected by Cyg OB2. However, this type of profile cannot be achieved in the picture where YMSCs are surrounded by an expanding bubble of hot gas. In order to reproduce the size of the observed emission, Aharonian et al. (2019) require a diffusion coefficient at 10 TeV which is  $\sim 5 \times 10^{25}$  cm<sup>2</sup> s<sup>-1</sup>. With this normalization of the diffusion coefficient,



Figure 2.11: Same as Fig. 2.6 but now both diffusion and advection timescales are calculated only considering a region of 54 pc, corresponding to the projected size of 2.2° that is the extent of the outermost ring used for the morphological analysis. Note that Kraichnan and Bohm advection times (dot dashed lines) are different as the best fit  $L_w$  are not equal. The dotted orange line represents the diffusion timescale considering the standard interstellar medium diffusion coefficient suppressed by two orders of magnitude.

if the energy dependence is  $\propto E^{\delta}$ , with  $\delta > 0$ , the diffusion time scale will always higher than the advection time at energies below 10 TeV. Aharonian et al. (2019) also provide an upper limit to the diffusion coefficient based on the efficiency of particle acceleration at 10 TeV. The upper limit is found to be two orders of magnitude below the Galactic diffusion coefficient. Assuming that the diffusion coefficient scales at low energy as  $D \propto E^{1/3}$ (hence following a Kolmogorov-like diffusion), even in this case the diffusion time is always larger or comparable with the advection timescale (see Fig. 2.11). Thus, the CR distribution should be flatter than a pure diffusive 1/r profile.

## 2.6 Cygnus OB2 as a cosmic ray accelerator?

Throughout this chapter, we tried to interpret the extended  $\gamma$ -ray radiation detected in coincidence with Cyg OB2 in terms of hadronic emission from a population of freshly accelerated CR under the assumption of particle acceleration at the wind TS. We tested three different CR distributions, considering different models of particle transport based on three specific diffusion coefficients calculated in quasi linear theory for the following plasma turbulence spectra: Kolmogorov-like, Kraichnan-like, and flat spectrum. Before

drawing conclusions and arguing about the capability of Cyg OB2 to accelerate particles, let us first summarize the main results obtained for the three different models:

- We have rejected the hypothesis that the acceleration and propagation of CRs are governed by Kolmogorov-like turbulence, since the resulting  $\gamma$ -ray emission would disagree with observations unless invoking unreasonable values of the cluster wind luminosity compared to the one estimated from the stellar population (see § 2.1.1).
- The Kraichnan model is disfavored as it also requires wind luminosities at least a factor of a few higher than the estimates based on the cluster star population.
- In contrast to the two previous cases, if one considers Bohm-like diffusion, the resulting CR distribution is able to account for the  $\gamma$ -ray spectrum with a reasonable  $L_w$  and an efficiency of the ordrer of  $\sim 30\%$  of turbulent magnetic field production.

Taking into account the results obtained for Kraichnan and Bohm models, it is plausible to consider the existence of a turbulence spectrum midway between the two, which is capable of reproducing the  $\gamma$ -ray spectrum with reasonable values of both  $L_w$  and  $\eta_B$ . In this scenario, the analyzed Kraichnan and Bohm models can be treated as limit cases, with the actual solution leading to spectral and morphological properties intermediate between the two. However, since both models considered are able to reproduce the spectrum equally well (as testified by the almost equivalent  $\chi^2$  value), and both return a predominantly flat morphology, it is natural to expect that also the true solution will adequately reproduce the spectrum as well, and will be likely characterized by a flat radial profile of  $\gamma$ -ray emission.

Having this in mind, we expect that the maximum energy of accelerated particles will be around 1 PeV. A model characterized by a flat morphology, both in high-energy and very high energy  $\gamma$ -ray, is at odds with the peaked profile observed in the Fermi-LAT energy band by Aharonian et al. (2019). One possible solution for this problem consists in trying to explain the increasing  $\gamma$ -ray luminosity towards the cluster center through the leptonic inverse Compton emission from an electron population located in a thin shell around the TS. This scenario will be explored in a future work.

To understand what the actual turbulence spectrum in the system is, a more detailed study would have to be carried out considering a multi-instrument analysis of the  $\gamma$ -ray emission. In a not so far future, this might be possible with the use of new-generation  $\gamma$ -ray telescopes, such as the Cherenkov Telescope Array and the ASTRI Mini Array. In this regard, additional information can be used during the analysis to better constrain the diffusion mechanism in the bubble. First, one can consider the  $\gamma$ -ray spectrum originating from molecular clouds that are possibly found inside the forward shock. Thanks to

precise maser parallaxes measurements, we know the existence of a few massive molecular clouds (i.e. DR21, W75N, and DR20) in the Cygnus-X region, located in close proximity to Cyg OB2. Unfortunately, the exact distance between the clouds and the star cluster is not easy to estimate, as the position of Cyg OB2 is still not well constrained. Fig. 2.12 shows the expected  $\gamma$ -ray spectrum from the molecular cloud DR21, considering different possible distances from Cyg OB2. It is interesting to note that, while below  $\sim$ 100 GeV the spectra obtained from the Kraichnan and Bohm models are similar, except for a normalization factor, in the very-high-energy range, the spectral shape in the two cases can be significantly different. The emission at very-high-energy can be hence used to constrain the diffusion coefficient.

A second possible way to constrain the diffusion coefficient is to analyze the  $\gamma$ -ray spectrum at different projected distances from Cyg OB2. Fig. 2.13 shows the spectra extracted from different regions of the system. The main difference between the two limit cases is observed in the cutoff part of the spectrum. Similarly to the Bohm case, a hard turbulence spectrum will produce sharper cutoff shapes. In the vicinity of the forward shock, the discrepancy between the two limit cases starts to become appreciable even at lower energies (>10 GeV), with the Kraichnan case producing softer emission. Indeed, modeling of spatially resolved  $\gamma$ -ray spectra appears as a promising way to constrain particle transport and contribute to unveil the mechanism of particle acceleration in YMSCs.


Figure 2.12: Expected hadronic  $\gamma$ -ray spectra from the molecular cloud DR21 considering the two possible CR distributions given by the Kraichnan (thin lines) and Bohm (thick lines) cases. The spectra are calculated considering different relative distances from Cyg OB2: 40 pc (continuous line), 60 pc (dashed line), and 80 pc (dot-dashed line). The dotted line shows the CTA North point source sensitivity for 50h of exposure considering the Alpha Configuration (CTA Collaboration, 2023), while black solid line is the Astri sensitivity for a point-like source for 200h of exposure (Lombardi et al., 2022).



Figure 2.13: Expected hadronic  $\gamma$ -ray spectra extracted from rings at projected distances of  $0 < r < \frac{1}{5}R_b$  (continuos lines),  $\frac{2}{5}R_b < r < \frac{3}{5}R_b$  (dash-dotted lines), and  $\frac{4}{5}R_b < r < R_b$  (dotted lines), for the Kraichnan (thin lines) and Bohm (thick lines) cases.

## CHAPTER 3\_\_\_\_\_

# LTHE CONTRIBUTION OF YMSCS TO THE DIFFUSE $\gamma$ -RAY EMISSION

IVEN the ability of YMSCs to produce CRs, it is natural to expect them to also be  $\gamma$ -ray emitters. In addition to the case of Cygnus OB2, a dozen of other clusters have been associated with diffuse  $\gamma$ -ray emission detected by different telescopes (see Tab. 3.1), namely: Westerlund 1 (Abramowski et al., 2012; Aharonian et al., 2022) (observed by HESS and Fermi-LAT), Westerlund 2 (Yang et al., 2018) (observed with HESS and Fermi-LAT), NGC 3603 (Saha et al., 2020) and NGC 6618 (Liu et al., 2022) (observed with Fermi-LAT).

At first glance, one might think that the number of cases is rather low compared to the expected population of galactic YMSCs. The reason for such a low number of observed YMSCs is probably due to a combination of two factors. First, most YMSCs are less massive than the objects listed above, which should be considered somewhat extreme cases. Therefore, a large fraction of Galactic YMSCs is expected to have lower  $L_w$ . This has a direct impact on the luminosity in terms of CRs, which affects the  $\gamma$ -ray emission. Secondly, as we saw in Ch. 2, most of the  $\gamma$ -ray emission comes from particles propagating within the wind blown bubble. Consequently, the projected size of the emission region can be considerably large, especially if compared to the resolution of current  $\gamma$ -ray telescopes. As we will show in § 3.2.4, the typical size of a bubble is of the order of tens of parsecs (10–100 pc), which corresponds to an angular size of:

$$\theta \simeq 0.57 \left(\frac{R_b}{10 \,\mathrm{pc}}\right) \left(\frac{d}{1 \,\mathrm{kpc}}\right)^{-1} \,\mathrm{deg}$$
(3.1)

with *d* the distance of the cluster from the Sun. The fact that they are potentially extended sources makes detection even more difficult, especially given the problems associated

Name	$\log M/{ m M}_{\odot}$	<i>r</i> <sub>c</sub> [pc]	D [kpc]	Age [Myr]	$L_w$ [erg s <sup>-1</sup> ]
Westerlund 1	$4.6\pm0.045$	1.5	4	4 - 6	10
Westerlund 2	$4.56\pm0.035$	1.1	$2.8\pm0.4$	1.5 - 2.5	2
Cygnus OB2	$4.7\pm0.3$	5.2	1.4	2 - 7	2
NGC 3603	$4.1\pm0.1$	1.1	6.9	2 - 3	-
BDS 2003	4.39	0.2	4	1	-
W40	2.5	0.44	0.44	1.5	-
RSGC 1	4.48	1.5	6.6	10 - 14	-
MC 20	$\sim 3$	1.3	3.8 - 5.1	3 - 8	$\sim 4$
NGC 6618	-	3.3	$\sim 2$	< 3	-
30 Dor (LMC)	4.8 - 5.7	multiple	50	1	_
NGC 2070 / RCM 136	4.34 - 5	subcluster	50	5	-

Table 3.1: List of YMSCs for which a diffuse  $\gamma$ -ray emission has been detected in their coincidence. References of each cluster: Westerlund 1 (Abramowski et al., 2012; Aharonian et al., 2022), Westerlund 2 (Yang et al., 2018), Cygnus OB2 (Bartoli et al., 2014b; Abeysekara et al., 2021b; Astiasarain et al., 2023), NGC 3603 (Saha et al., 2020), BDS 2003 (Albert et al., 2021), W40 (Sun et al., 2020b), RSGC 1 (Sun et al., 2020a), MC 20 (Sun et al., 2022), NGC 6618 (Liu et al., 2022), and the Large Magellanic Clusters (H. E. S. S. Collaboration et al., 2015). Values marked by a '-' are not provided in the literature.

with studying extended sources in the  $\gamma\text{-ray}$  band.

Given the difficulty of studying single sources, we may study the contribution of multiple overlapping  $\gamma$ -ray halos arising from a population of YMSCs, to estimate whether this may result in a non-negligible contribution to the large-scale diffuse emission along the Galactic plane. The estimation of this emission and subsequent comparison with the data is a key element that can be used to constrain YMSCs as CR factories. In addition, calculating the emission from a population of YMSCs is also useful in view of the next generation of gamma telescopes, since it can be used to estimate the number of potentially observable YMSCs.

In this chapter, we will calculate the diffuse  $\gamma$ -ray emission arising from a synthetic population of Galactic YMSCs and compare it with available observations from Fermi-LAT. The structure of the chapter traces the workflow implemented to reach our objectives. First, we describe how to generate for each cluster a stellar population from known initial mass functions and how to calculate the fundamental properties of stars such as

luminosity, radius, and temperature. We also describe the recipes used to model stellar winds. Next, we discuss how to simulate a population of Galactic YMSCs, illustrating our choices for the distribution in mass, galactic position, and age.

Once we have all the pieces of the puzzle, we proceed to investigate the properties of the resulting Galactic YMSC population in terms of particle acceleration. To this goal, we estimate the expected  $\gamma$ -ray emission for each cluster, and we compute the diffuse contribution from the entire population. Finally, we compare the results with existing observations and discuss our findings. We anticipate that our results will provide a lower limit to the possible contribution to the  $\gamma$ -ray sky in that we are neglecting the role of SN exploding inside stellar clusters.

## **3.1 Modeling the star population inside YMSCs**

For any given YMSC, characterized by a specific mass and age, we need to build a stellar population that is consistent with its properties. The general approach is the following: knowing the mass and age of the cluster, we build a population of stars given the initial stellar mass function (IMF). Then, based on the cluster age, we remove all the stars that are expected to have exploded as supernovae. Afterward, we compute the intrinsic characteristics of all the stars that are left, such as luminosity, radius, and temperature. To this purpose, we do not use any stellar model, but rather the observed mass-luminosity, mass-radius, and mass-temperature relations. The usage of empirical relations is preferred to the usage of more robust stellar models as computational time is significantly decreased. Finally, for every star, we calculate the main parameters of the stellar wind, such as the wind luminosity, the wind speed, and the mass loss rate.

In the next subsection we describe in details all the ingredients implemented in the afore mentioned procedure.

## 3.1.1 Mass distribution of stars inside clusters

The number of stars formed as a function of their mass  $(M_*)$  is generally referred to as the *stellar initial mass function*  $(f_*(M_*))$ . Generally such a function is parameterized as a set of broken power laws. Above ~10 M<sub> $\odot$ </sub> there is a general consensus that the IMF follows the Salpeter law  $\propto M^{-2.3}$ . At lower mass the IMF is flatter, but its exact shape is still debated. Here we have decided to use the expression given by Kroupa (2001):

$$f_{\star}(M_{\star}) \propto \frac{dN_{\star}}{dM_{\star}} = \begin{cases} M_{\star}^{-0.3} & \text{for } M_{\star} < 0.08 \,\mathrm{M}_{\odot} \\ 0.08 M_{\star}^{-1.3} & \text{for } 0.08 \,\mathrm{M}_{\odot} \le M_{\star} \le 0.5 \,\mathrm{M}_{\odot} \\ 0.04 M_{\star}^{-2.3} & \text{for } M_{\star} > 0.5 \,\mathrm{M}_{\odot} \end{cases}$$
(3.2)

Each cluster with mass M will initially include a total number of stars equal to:

$$N_{\star}(M) = M \frac{\int_{M_{\star,min}}^{M_{\star,max}} f_{\star}(M_{\star}) dM_{\star}}{\int_{M_{\star,min}}^{M_{\star,max}} M_{\star} f_{\star}(M_{\star}) dM_{\star}}$$
(3.3)

where  $M_{\star,min}$  and  $M_{\star,max}$  are respectively the minimum and maximum stellar masses that can be generated in a cluster. We fix the value of  $M_{\star,min}$  to 0.08 M $_{\odot}$ , which is the minimum theoretical mass to support significant nuclear burning (Carroll & Ostlie, 1996). The choice of maximum mass, on the other hand, turns out to be an extremely delicate problem<sup>1</sup>. As a matter of fact,  $M_{\star,max}$  represents a crucial parameter for the purposes of this work, as massive stars are the ones that contribute the most to the YMSC wind luminosity. Clearly, this value cannot be arbitrarily high, and a first limit to  $M_{\star,max}$  is given by  $M_{\star,max} = 150 \text{ M}_{\odot}$ . This value seems to be widely recognized as a fundamental mass upper limit for stars with zero metallicity that form in clusters (Weidner & Kroupa, 2004; Figer et al., 2005; Oey & Clarke, 2005; Koen, 2006), and it is also the maximum value of stellar mass observed in our Galaxy. In general, it is reasonable to consider the existence of a relation between  $M_{\star,max}$  and M. Indeed, this must be true for low mass stellar clusters since, for example, a 100  $M_{\odot}$  cluster cannot include stars with masses equal or greater than 100  $M_{\odot}$ . However, for massive stellar clusters, the matter is presently still under debate. Weidner et al. (2010) seem to prove the existence of this relationship through a comprehensive study of the literature, showing that it is extremely unlikely to reproduce the observed stellar populations with a random sampling of the initial mass function that does not account for the clusters masses. On the other side, a parallel analysis of published data performed by Maschberger & Clarke (2008) using sophisticated unbiased selection criteria for stellar clusters shows that even low mass clusters do possess populations of massive stars. Given the importance of the parameter  $M_{\star,max}$  and its current uncertainties, we decide to run our analysis considering two different scenarios. In the first we use a constant maximum mass for all YMSCs fixed to the 150  $M_{\odot}$ limit. In the other case, we consider a maximum mass that depends on the cluster mass, following the relation provided by (Weidner & Kroupa, 2004) and reported in Fig. 3.1.

After generating the stellar population, depending on the YMSC age, we remove all

<sup>&</sup>lt;sup>1</sup>See Bastian et al. (2010b) for a comprehensive description of the topic.



Figure 3.1: Relationship between the maximum stellar mass that can be generated in a cluster and the mass of the host cluster (dashed line, lower x-axis). The graph also reports the mass of stars leaving the main sequence at a given time  $t_{TO}$  (solid line, upper x-axis), given by inverting Eq. 3.4.

those stars that exploded as supernova. We do so by considering that a star with a given mass  $M_{\star}$  will leave the main sequence (and soon after explode as a supernova) at a turn-off time ( $t_{TO}$ ) approximately given by the following relation (Buzzoni, 2002):

$$\log\left(\frac{t_{TO}}{1\,\mathrm{yr}}\right) = 0.825\log^2\left(\frac{M_{\star}}{120\,\mathrm{M}_{\odot}}\right) + 6.43\tag{3.4}$$

which is obtained fitting several of sets of theoretical models. Fig. 3.1 shows the resulting inverse relation. Note that no stars explode before  $\sim 2.8$  Myr, and that, for the maximum cluster age we are interested in (10 Myr), only stars with less than 20 M<sub> $\odot$ </sub> can survive.

### **3.1.2 Modeling stellar parameters**

For each generated star, we estimate stellar luminosity and radius using empirical relations. The choice of this method, expected to provide estimates that are less robust than those based on stellar models, comes for two needs: first, considering a larger mass interval than that usually covered by stellar models. Second, to minimize computation time. The latter turns out to be an extremely relevant aspect since, eventually, we intend to calculate multiple synthetic stellar populations to obtain an estimate of the diffuse  $\gamma$ -ray emission together with an appropriate guess of its statistical fluctuations. Note that, from now on, we will consider in our analysis only stars with masses larger than 2.75 M<sub> $\odot$ </sub>, roughly corresponding to the lower mass limit for a B-type star. We do so as the contribution to the cluster wind power of stars with masses below  $\sim 3 \, M_{\odot}$  is believed to be negligible.

#### Mass-Luminosity relation

Various mass-luminosity relationships (MLRs) have been proposed over the years. In general, several authors provide MLRs that are valid only in specific mass ranges. Eker et al. (2018), for example, model the MLR as a series of power laws valid from 0.179–31  $M_{\odot}$ . For the range of masses in which we are interested, the expression is:

$$\log\left(\frac{L_{\star}}{\mathrm{L}_{\odot}}\right) = \begin{cases} 3.967 \log\left(\frac{M_{\star}}{\mathrm{M}_{\odot}}\right) + 0.093 & \text{for } 2.4 < \frac{M_{\star}}{\mathrm{M}_{\odot}} < 7\\ 2.865 \log\left(\frac{M_{\star}}{\mathrm{M}_{\odot}}\right) + 1.105 & \text{for } 7 \le \frac{M_{\star}}{\mathrm{M}_{\odot}} \le 31 \end{cases}$$
(3.5)

In the case of very massive stars ( $M_{\star}$ >100  $M_{\odot}$ ), the MLR is provided by Yungelson et al. (2008), whose expression has been used in § 2.1.1 for the calculation of Cygnus OB2 parameters. The formal equation is obtained inverting Eq. 2.5:

$$L_{\star} = 10^{3.48} \left(\frac{M_{\star}}{M_{\odot}}\right)^{1.34} L_{\odot}$$
 (3.6)

No MLR is defined over a broad enough range of masses for our purposes, as we are interested in a wide interval ranging from 2.75 to  $150 M_{\odot}$ . To overcome this problem, we decided to merge the two recipes given by Eker et al. (2018) and Yungelson et al. (2008). We do so by implementing a set of smoothed broken power-laws, defined as:

$$L_{\star} = \begin{cases} L_{b1} \left(\frac{M_{\star}}{M_{b1}}\right)^{\alpha_{1}} \left[\frac{1}{2} + \frac{1}{2} \left(\frac{M_{\star}}{M_{b1}}\right)^{1/\Delta_{1}}\right]^{(-\alpha_{1} + \alpha_{2})\Delta_{1}} & \text{for } 2.4 \le \frac{M_{\star}}{M_{\odot}} < 12 \\ \mathcal{K}L_{b2} \left(\frac{M_{\star}}{M_{b2}}\right)^{\alpha_{2}} \left[\frac{1}{2} + \frac{1}{2} \left(\frac{M_{\star}}{M_{b2}}\right)^{1/\Delta_{2}}\right]^{(-\alpha_{2} + \alpha_{3})\Delta_{2}} & \text{for } M_{\star} \ge 12 \,\mathrm{M}_{\odot} \end{cases}$$
(3.7)

where  $L_{b1} = 3191 L_{\odot}$  and  $L_{b2} = 368874 L_{\odot}$  are the luminosity values calculated using Eq.3.5 at the mass break points  $M_{b1} = 7 M_{\odot}$  and  $M_{b2} = 36.089 M_{\odot}$  respectively. The value of  $M_{b2}$  is the intersection point between the two MLRs of Yungelson et al. (2008) and Eker et al. (2018). The power law indexes are  $\alpha_1 = 3.97$ ,  $\alpha_2 = 2.86$ , and  $\alpha_3 = 1.34$ 

respectively. The parameters  $\Delta_1$  and  $\Delta_2$  are used to smooth the junction between the power law components. We fix the two parameters to 0.01 and 0.15 respectively. Finally,  $\mathcal{K}\,=\,0.817$  is a normalization constant providing continuity at 12  $M_{\odot}$  Note that in the interval 2.4–12  $M_{\odot}$  the MLR is an adaption of Eq.3.5 in the form of a smoothed power law. We decided to use this form rather than Eq.3.5 as we found that Eq.3.5 is not continuous at  $M_{b1} = 7 \text{ M}_{\odot}$ . To check whether the extension of the MLR given by Yungelson et al. (2008) to masses less than 100  $M_{\odot}$  is valid, we compare Eq. 3.7 with archival data of massive stars where both bolometric luminosities and masses are given. For this purpose, we consider massive stars observed in different YMSCs, such as Cygus OB2 (Wright et al., 2015), R136 (Brands et al., 2022)<sup>2</sup>, and 30 Doradus (Schneider et al., 2018). We additionally include eclipsing binary stars from the DebCAT catalog (Southworth, 2015) and the sample of stars used by Eker et al. (2015) in their first work for the estimation of the MLR. Top panel of Fig. 3.2 shows the result of this comparison: the MLR given by Eq. 3.7 is in fair agreement with the observations, with the underlying implication that the relationship of Yungelson et al. (2008) is valid even at smaller masses, down to  $M_{b2}$ (eventually the relation can be considered still valid down to about  $\sim 20 \text{ M}_{\odot}$ ).

#### **Mass-Radius relation**

Mass-radius relations (MRRs) are in general less constrained and much broader than MLRs. Similarly to what we have done in § 2.1.1, we employ the relation provided by Demircan & Kahraman (1991):

$$R_{\star} = 0.85 \left(\frac{M_{\star}}{M_{\odot}}\right)^{0.67} R_{\odot} .$$
 (3.8)

Eq. 3.8 is plotted in the middle panel of Fig. 3.2, where a comparison with archival data is made showing that the MRR is able to adequately reproduce the observations. To this purpose, we used the same dataset employed for the MLR, with the exception of the Cygnus OB2 stellar cluster where the information on stellar radii is not provided. For the sake of completeness, we also report in Fig. 3.2 an additional MRR obtained from the work of Yungelson et al. (2008). We found the latter relation in strong disagreement with both Eq. 3.8 and data from the literature, although the authors claim that the relation should be valid for the mass range  $25-115 M_{\odot}$ . All stellar radii employed for the comparison are obtained using stellar models calibrated to other observable stellar parameters (see references for details). The only exception is that of the DebCAT data, which are based on direct measurements of the stellar radii using detached eclipsing binaries.

<sup>&</sup>lt;sup>2</sup>For every star Brands et al. (2022) provide different estimated properties after fitting stellar models with a different number of free parameters. The choice of the number of parameters ultimately depends on the available spectroscopy for each star (UV and optical or only optical).

#### **Mass-Temperature relation**

Even though empirical mass-temperature relations (MTRs) exist in the literature, once the MLR and MRR are given, we can readily calculate stars effective temperatures using Boltzmann law:

$$T_{\rm eff} = \left[\frac{L_{\star}(M_{\star})}{4\pi R_{\star}(M_{\star})^2 \sigma_b}\right]^{1/4} \,. \tag{3.9}$$

The bottom panel of Fig. 3.2 shows the difference between the empirical relation given by Eker et al. (2018), and the resulting MTR obtained from Eq. 3.9. For comparison, we also display the mass and temperature of massive stars. Once again, we consider the same set of stars extracted from catalogs that are used for the study of the MLR. With our choice of MLR and MRR, Eq. 3.9 is consistent with observed data, although, at masses below 10  $M_{\odot}$ , the temperature is slightly overestimated (of the order of few tens percent). Note that for  $M_{\star} \gtrsim M_{b2}$ , the effective temperature becomes independent from the mass. This is a direct consequence of the specific combination of the power index of our chosen MLR and MRR. From Eq. 3.7 we have that  $L_{\star}(M_{\star} \gg M_{b2}) \propto M_{\star}^{1.34}$ , while from Eq. 3.8,  $R_{\star} \propto M_{\star}^{0.67}$ . As from Eq. 3.9 one has that  $T_{\rm eff} \propto L_{\star}^{1/4} R_{\star}^{-1/2} \propto M_{\star}^{0}$ , for  $M_{\star} \gg M_{b2}$ , the resulting temperature is independent of the stellar mass.



Figure 3.2: *Top panel*: Mass-luminosity relationship used in this work (Eq. 3.7) (dashed black line) together with the relations provided by Eker et al. (2018) and Yungelson et al. (2008). For comparison, we also plot the well known MLR given by Salaris & Cassisi (2005). *Central panel*: Mass-radius relationship used in this work (Demircan & Kahraman, 1991) compared to the MRR provided by Yungelson et al. (2008). *Bottom panel*: Comparison of empirical mass-temperature relation given by Eker et al. (2018) with the relation obtained via Boltzmann's law (Eq. 3.9). In all panels the various data points report observed stellar properties and computed masses for several stars in different clusters, such as Cyg OB2 (Wright et al., 2015),

R136 (Brands et al., 2022), and 30 Doradus (Schneider et al., 2018). We also include stars from the DebCAT catalog (Southworth, 2015) and the previous work on MLRs of Eker et al. (2015).

## 3.1.3 Modeling stellar winds

Once the stellar parameters are known, we can readily estimate the wind speed, mass loss rate, and wind power for each star. From these we can afterward evaluate the cluster wind properties. We here use for the wind speed  $(v_{\star,\infty})$  and wind luminosity  $(L_{\star,w})$  the same expressions implemented in § 2.1.1, that are Eq. 2.2 and Eq. 2.7 respectively, which we again report below for convenience:

$$v_{\star,\infty} = C(T_{\text{eff}}) \left[\frac{2GM_{\star}(1-\Gamma)}{R_{\star}}\right]^{1/2}$$
$$L_{\star,w} = \frac{1}{2}\dot{M_{\star}}v_{\star,\infty}^{2}.$$

For what concerns the mass loss rate ( $\dot{M}_{\star}$ ), we use a different recipe from that of Vink et al. (2000) and Yungelson et al. (2008), employed for the case of Cygnus OB2. This is because both expression are expected to be accurate only for very massive star, while here we need a prescription valid for a broader range of masses. The empirical formula provided by Vink et al. (2000) works only for stars with  $27500 \text{ K} < T_{\text{eff}} < 50000 \text{ K}$ , which, considering Eq. 3.9, implies stars with masses larger than  $\sim 20 \text{ M}_{\odot}$ . Similarly, the expression provided by Yungelson et al. (2008) is in principle valid for  $M_{\star} > 60 \text{ M}_{\odot}$ , although the result obtained including also stars of mass about to  $20 \text{ M}_{\odot}$  is approximately correct within a factor of a few, as demonstrated in § 2.1.1.

Given our plan to consider also less massive stars, we use the expression provided by Nieuwenhuijzen & de Jager (1990), which reads:

$$\log\left(\frac{\dot{M}_{\star}}{\mathrm{M}_{\odot}\mathrm{yr}^{-1}}\right) = -14.02 + 1.24\log\left(\frac{L_{\star}}{\mathrm{L}_{\odot}}\right) + 0.16\log\left(\frac{M_{\star}}{\mathrm{M}_{\odot}}\right) + 0.81\left(\frac{R_{\star}}{\mathrm{R}_{\odot}}\right) \quad (3.10)$$

Eq. 3.10 is valid for stars with  $T_{\rm eff}>5000$  K, hence it is adequate for all the considered mass range.

As a final consistency check, we compare stellar wind luminosity and mass-loss rates with data available in the literature. Brands et al. (2022) provide these data for a sub-sample of stars in the YMSC R136. Similarly, Mokiem et al. (2007) provide them for a set of stars located both in the Milky Way and in the Small and Large Magellanic Clouds. Fig. 3.3 shows the result of this comparison. In this regard, it is interesting to note the following: first, the expression (Eq. 3.10) used for  $\dot{M}_{\star}$  is in good agreement with the data. Second, for masses below  $\sim 25 \text{ M}_{\odot}$  the stars wind luminosity appears to be highly overestimated, by almost 1-2 orders of magnitude on average. However, we expect that this error will not significantly affect the final result. In fact, even if heavily overestimated, the contribution to the cluster wind luminosity of tens of stars under 25 M<sub> $\odot$ </sub> is negligible



compared to the wind power of a single star of mass greater than  $\sim 30 \text{ M}_{\odot}$ .

Figure 3.3: *Top panel*: Expected stellar mass loss rates as a function of  $L_{\star}$  compared with observations (Mokiem et al., 2007; Brands et al., 2022). The Sun mass loss rate is reported by Carroll & Ostlie (1996). The continuous line shows the relation used in this work, while dashed and dotted lines are the mass loss rates provided by Yungelson et al. (2008) and Vink et al. (2000) respectively, which have been used in § 2.1.1. *Bottom panel*: Wind power as a function of stellar luminosity computed using the mass loss rate given by Nieuwenhuijzen & de Jager (1990). The solar wind luminosity is calculated considering a wind speed of 400 km s<sup>-1</sup> (Carroll & Ostlie, 1996).

## 3.2 Generating a synthetic population of YMSCs

In order to simulate a population of stellar clusters, the core ingredient from which one must start is the *cluster distribution function*:

$$\xi_{SC}(M, t, r, \theta) = \frac{dN}{dM dt dr d\theta}$$
(3.11)

defined such that the total number of clusters in the Milky Way with masses ranging in a given interval  $[M_{min}, M_{max}]$ , and age  $[t_{min}, t_{max}]$  is:

$$N_{SC} = \int_{M_{min}}^{M_{max}} \int_{t_{min}}^{t_{max}} \int_{0}^{R_{MW}} r\xi_{SC}(M, t, r) dM dt dr d\theta$$
(3.12)

where  $R_{MW}$  is the Milky Way radius. The true form of  $\xi_{SC}$  is not known, however, assuming that the cluster distribution is factorized in mass, time, and space,  $\xi_{SC}$  can be written as:

$$\xi_{SC}(M,t,r) = f(M)\psi(t)\rho(r,\theta)$$
(3.13)

where f(M),  $\psi(t)$ , and  $\rho(r, \theta)$  are the cluster initial mass function, the cluster formation rate and the cluster spatial distribution respectively. In the next subsections, we discuss each single term separately.

## 3.2.1 Cluster age and mass distribution

It is possible to infer both the cluster formation rate and cluster mass function from observations. One of the seminal works for the study of these two functions is the analysis done by Piskunov et al. (2018) on the Milky Way Star Cluster Survey (MWSCS). Piskunov et al. (2018) used 2242 stellar clusters from this survey for their analysis, all within 2.5 kpc from the Sun. Actually, the sample can be considered complete only up to 1.8 kpc. Hence the result from Piskunov et al. (2018) should be considered as local.

In their paper, Piskunov et al. (2018) model the cluster initial mass function as a broken power law:

$$f(M) = \frac{dN}{dM} = \begin{cases} k_1 M^{-(x_1+1)} & \text{for } M_{min} \le M \le M_b \\ k_2 M^{-(x_2+1)} & \text{for } M_b \le M \le M_{max} \end{cases}$$
(3.14)

where  $M_b = 100 \text{ M}_{\odot}$  is the mass at which the break occurs, and  $x_1$  and  $x_2$  are parameters used to fit the observed cluster mass distribution. Note that Eq. 3.14 coincides with the observed cluster mass distribution only in the case of young clusters. As explained in § 1.2.1, some of the clusters may not survive the initial stage of gas expulsion triggered by the appearance of stellar feedback mechanisms. This eventually ends up affecting the mass distribution of currently observed clusters, hence, in order to obtain the initial cluster mass function, one has to account for this effect. Numerical N-body simulations show that the cluster survival depends on the percentage of gas filling the cluster Roche lobe (Ernst et al., 2015). When it comes to inferring f(M), Piskunov et al. (2018) consider three different scenarios associated with underfilled, filled, and overfilled Roche lobes. Ultimately, this produces differs values for the parameters  $x_1$  and  $x_2$ . The scenario that best agrees with the observations is the underfilled case, for which  $x_1 = 0.39$  and  $x_2 =$ 0.54. Finally,  $k_1$  and  $k_2$  are two constants obtained by requiring both continuity at  $M_b$ and normalization of the distribution  $\int_{M_{min}}^{M_{max}} f(M) dM = 1$ , which are calculated as:

$$k_{2} = \left[M_{b}^{(x_{1}-x_{2})}\int_{M_{min}}^{M_{b}}M^{-(x_{1}+1)}dM - \int_{M_{b}}^{M_{max}}M^{-(x_{2}+1)}dM\right]^{-1}$$
(3.15a)

$$k_1 = k_2 M_b^{(x_1 - x_2)}$$
 (3.15b)

In the work carried out by Piskunov et al. (2018), the minimum and maximum stellar cluster mass are fixed to  $M_{min} = 2.5 \text{ M}_{\odot}$  and  $M_{max} = 6.3 \times 10^4 \text{ M}_{\odot}$  respectively. To be consistent with Piskunov et al. (2018), we also fix  $M_{min}$  to the same value, which should correspond to binary systems or small brown dwarf aggregates. However, it is crucial to stress that when we will calculate the number of Galactic massive stellar clusters using Eq. 3.12, the minimum considered mass  $M_{min}$  will be different from 2.5 M<sub>☉</sub>. The choice of  $M_{min}$  has a certain degree of arbitrariness. In § 1.1, we defined as massive all those clusters with masses greater than 1000 M<sub>☉</sub>. In general, the choice of  $M_{min}$  is made by considering a reasonably small mass to produce enough massive stars to make the existence of a collective cluster wind possible. We will check a posteriori in § 3.2.4 if this choice is reasonable.

The choice of  $M_{max}$  in Piskunov et al. (2018) is dictated by the most massive star cluster observed in the Milky Way. However, observations of Milky Way star clusters are biased by extinction. Hence, we decided to make a consistency check based on observations of other closeby galaxies. Observational pieces of evidence show how the luminosity of YMSCs in spiral and dwarf galaxies correlates with the measured value of the star formation rate (SFR) in their host galaxy (Weidner et al., 2004): the higher the SFR, the higher the luminosity of the brightest (hence the most massive) YMSCs. The inferred relation between the SFR and  $M_{max}$  is (Weidner et al., 2004):

$$M_{max} = k_{ML} \left(\frac{SFR}{1 \,\mathrm{M}_{\odot} \mathrm{yr}^{-1}}\right)^{0.75 \pm 0.03} \times 10^{6.77 \pm 0.02} \,\mathrm{M}_{\odot}$$
(3.16)

where  $k_{ML}$  is the cluster mass-to-light ratio which depends on the cluster age. For clusters with ages less than 10 Myr one has  $k_{ML} = 0.0144$  (Smith & Gallagher, 2001). Considering that the Milky Way SFR is  $\sim 2 M_{\odot} \text{ yr}^{-1}$  (Elia et al., 2022), the expected maximum mass is  $M_{max} \approx 1.4 \times 10^5 \text{ M}_{\odot}$ . Note that this value must be interpreted as an upper limit on a Galactic scale. However in different places of the Milky Way, it is reasonable to think that the maximum mass of a YMSC correlates with the total amount of gas. Eventually, the work performed by Pflamm-Altenburg & Kroupa (2008) has proven this relation. In their paper, Pflamm-Altenburg & Kroupa (2008) show how the H $\alpha$  emission in disk galaxies (tracing the population of short-lived massive stars) is characterized by a cut-off at some galactocentric distances, which can be smoothly explained by introducing a cluster mass function with  $M_{max,r}$  dependent on the total gas surface density:

$$M_{max,r}(r) = M_{max} \left[ \frac{\Sigma_{gas}(r)}{\Sigma_{gas}(r=0)} \right]^{\delta}$$
(3.17)

where  $\delta = 3/2$  that can be inferred by observations. Fig. 3.4 shows the evaluation of  $M_{max,r}(r)$ . We estimate  $M_{max,r}(r)$  considering the radial gas distribution implemented by Strong et al. (2000) in the GALPROP code (Strong et al., 2009). The radial gas profile accounts for both molecular and atomic gas phases. Interestingly, the maximum masses at different galactocentric radii are perfectly consistent with the measured mass of some of the most massive Galactic YMSCs. One can see that for radii larger than 2 kpc, the maximum mass is roughly constant, with variations by a factor of a few.

It is worth underlining that if we consider a radius-dependent maximum mass, Eq. 3.14 must necessarily be modified by replacing  $M_{max}$  with  $M_{max}(r)$ . This will ultimately produce a radial dependence of the parameters  $k_1$  and  $k_2$  which can be then obtained by requiring the normalization of f(M) at each galactocentric radius. Clearly, by doing so, f(M) will become dependent on r, making invalid the hypothesis of factorization for  $\xi_{SC}$ . Given that the maximum mass does not vary much with galactocentric distance, we will use, for the rest of the work, the approximation of a constant mass equal to the value adopted by Piskunov et al. (2018) ( $M_{max} = 6.3 \times 10^4 \, M_{\odot}$ ).

As stated at the beginning of this section, Piskunov et al. (2018) provide also an expression for the cluster formation rate  $\psi(t)$  inferred from the present population of local clusters. Comparison with data shows that several functional forms are allowed. A useful form adopted by Piskunov et al. (2018) is:

$$\psi(t) = \left[A + B \exp\left(C\frac{T_p - T_f}{T_p}\right)\right] \text{ Myr}^{-1} \text{kpc}^{-2}$$
(3.18)

where  $T_p = 4.8$  Gyr is the present time, considering as initial reference point the age of



Figure 3.4: Maximum cluster mass as a function of the galactocentric radius (dashed line). The normalization of  $M_{max,r}$  at the Galactic Center is calculated using Eq. 3.16. For comparison purposes, we show the mass of several of the most massive YMSCs, such as Cygnus OB2 (Wright et al., 2015), Westerlund 1 (Brandner et al., 2008), NGC 3603 (Rochau et al., 2010; Röllig et al., 2011a), Archer and Quintuplet (Figer, 2004). The maximum mass used by Piskunov et al. (2018) is also reported (dotted line).

the oldest observed cluster, and  $T_f$  is the time at which the cluster was formed, such that the age of the cluster can be calculated as  $t = T_p - T_f$ . Eq. 3.18 is such that the present SFR is given by  $\psi_{now} = A + B$ , while SFR at  $t = T_p$  is  $\psi_0 = A + Be^C$ . The coefficients A, B and C are obtained by fitting the observed cluster age distribution. Similarly to  $x_1$  and  $x_2$  in f(M), the coefficients A, B, and C also depend on the clusters capability to survive. Still considering the previously mentioned underfilled Roches lobe case, the values of A, B ad C are -0.55, 0.57, and 1 respectively. Using Eq. 3.18 one can readly see that the cluster formation rate is practically constant in the last 10 Myr, as the value of  $\psi(t)$  changes only by a few percent in the 1–10 Myr range:

$$\frac{\psi(T_p - 10 \,\mathrm{Myr}) - \psi(T_p)}{\psi(T_p)} \simeq 0.06$$

This is also well confirmed by observations, as shown in Fig. 1 of Piskunov et al. (2018). Hence, we can consider the cluster formation rate as constant in the last 10 Myr.

The average value of the cluster formation rate ( $\psi$ ) is a crucial parameter as in the

end it will be the one setting the normalization of  $\xi_{SC}$ . In order to determine the value of  $\bar{\psi}$ , we rely on the work by Lamers & Gieles (2006) who determined from observations the SFR in local clusters ( $SFR_{SC}$ ) for masses between  $M_{-} = 100 \text{ M}_{\odot}$  and  $M_{+} = 3 \times 10^4 \text{ M}_{\odot}$ . Such a rate is  $\sim 350 \text{ M}_{\odot} \text{ Myr}^{-1} \text{ kpc}^{-2}$ . The result of Lamers & Gieles (2006) is also compatible with a more recent estimate done by Bonatto & Bica (2011), who found a SFR in local clusters of  $790 \pm 160 \text{ M}_{\odot} \text{ Myr}^{-1} \text{ kpc}^{-2}$  after considering clusters within a mass range of  $M_{-} = 10 \text{ M}_{\odot}$  and  $M_{+} = 7.5 \times 10^4 \text{ M}_{\odot}$ . Starting from  $SFR_{SC}$ , it is possible to infer  $\bar{\psi}$  as:

$$\bar{\psi} = \frac{SFR_{SC}}{\int_{M}^{M_{+}} Mf(M)dM} \,. \tag{3.19}$$

This leads to average cluster formation rates of  $\bar{\psi}_{LG} \approx 1.3 \text{ Myr}^{-1} \text{ kpc}^{-2}$  and  $\bar{\psi}_{BB} \approx 1.8 \text{ Myr}^{-1} \text{ kpc}^{-2}$  for the works of Lamers & Gieles (2006) and Bonatto & Bica (2011) respectively. When simulating the Galactic population of YSMCs, we will consider the value  $\bar{\psi}_{BB}$  because it is the most recent one (however, given the uncertainties, the two values are consistent with one another).

### **3.2.2** Cluster spatial distribution

In the previous section we have determined the cluster formation rate in the solar neighborhood. However, the cluster formation rate is expected to vary across the Galactic disk according to the density of giant molecular clouds (GMC) from which stellar clusters originate. It is then reasonable to presume that the distribution of GMCs well traces the Galactic position of YSMCs, which in turn closely follow the spiral arm structure of the Milky Way. We spatially distribute the synthetic stellar clusters following a two-step procedure:

- 1. We start by generating stellar clusters with a galactocentric radial distribution following that of GMCs (under the assumption of an isotropic angular distribution) and assuming an exponential altitude distribution similar to the observed gas profile.
- 2. Afterward, based on its radial and angular position, we associate every synthetic YMSC to a specific Galactic structure, i. e. spiral arm, galactic bar, etc.

For the radial distribution of GMCs and the modeling of the Milky Way spiral structure, we rely on the materials and results of the work by Hou & Han (2014), who fitted spiral arms models simultaneously using the distribution of observed HII regions, masers, and GMCs.

Hou & Han (2014) supply a complete catalog of GMCs, which reports, for each cloud, the galactic position (along with kinematic distance), kinematic velocity, size and mass.

For a large portion of the GMCs, the kinematic ambiguity is resolved. We then recompute the kinematic distance of every cloud with disentangled kinematic ambiguity using the code developed by Wenger et al. (2018), which calculates distances using two possible approaches: a Montecarlo method or the conventional estimate using the Galactic rotation curve (see § 2.3.1). We adopt the classical rotation curve based method using the state of the art curve provided by Reid et al. (2019). A small fraction of GMCs in the catalog also have distances calculated using more reliable methods such as parallax. For those specific cases, the latter value is used.

Once the location is known, we can estimate the radial distribution of GMCs, which we express in terms of the surface mass density of molecular gas ( $\Sigma_{H_2}^{GMC}$ ). To do so, we average the total GMCs mass in 18 rings with constant width of 1 kpc and radii spanning in the interval 0–18 kpc, and centered on the Galactic center. The result is shown in Fig. 3.5, where the surface mass density of molecular material in GMCs is compared to several measurements of the diffuse molecular gas distribution. Note that the overall profile is remarkably compatible with the results from different works (Grabelsky et al., 1987; Bronfman et al., 1988; Digel, 1991; Nakanishi & Sofue, 2006; Pohl et al., 2008), except for two aspects. First, the normalization, which is found to be lower by a factor  $\sim 3$ . This is expected as we are considering only molecular gas within GMC, while the other results account for all the diffuse molecular gas. Second, the trend towards the Galactic Center, where data by Nakanishi & Sofue (2006) indicate an increase of the density, which in parallel seems not consistent with the findings of Bronfman et al. (1988). Nevertheless, the profile is also in good agreement with the radial distribution of far-infrared emission detected from embedded OB stars (Bronfman et al., 2000), emphasizing the robustness of the hypothesis that YMSCs and molecular gas.

At this point we can formally define the radial distribution of YMSCs as:

$$\rho(r) = \frac{\Sigma_{H_2}^{GMC}(r)}{\Sigma_{H_2}^{GMC}(r = 8.5 \text{ kpc})}$$
(3.20)

Note that by doing so, the normalization of  $\xi_{SC}$  at the sun position is given by the local observed cluster formation rate  $\bar{\psi}_{LG}$ , as  $\rho(r)$  is normalized at the Sun position.

Once the radial distribution is known, we allocate stellar clusters following the Milky Way observed morphology<sup>3</sup>. Unfortunately, the Milky Way spiral structure still remains nowadays a matter of debate. Within their work, Hou & Han (2014) model the Galaxy by considering two possible scenarios, one containing three and one containing four spiral arms. They furthermore consider two different functions for the spiral arms: logarith-

<sup>&</sup>lt;sup>3</sup>In the work presented, as we will describe in Section 3, we will only consider two regions of the Galactic plane. However, we still decided to simulate the distribution of clusters rigorously over the entire galaxy, especially in view of future more detailed work.



Figure 3.5: Comparison of surface mass density of molecular gas enclosed in GMC (thin dashed line), calculated using Hou & Han (2014) catalog, and mass density of diffuse molecular gas (see text for the references). The thick solid line is the surface mass density rescaled for comparison purposes. The dash-dotted line referring to the right axis is the FIR surface luminosity calculated from embedded massive stars (Bronfman et al., 2000).

mic and polynomial-logarithmic functions. Here we use the 4-arm model with simple logarithmic function as it is on average the case returning the best fit to the spiral arm tracers. The logarithmic spiral arm, are defined as:

$$\ln\left(\frac{r}{R_i}\right) = \left(\frac{\theta - \theta_i}{1 \operatorname{rad}}\right) \tan \Psi_i \tag{3.21}$$

where  $R_i$ ,  $\theta_i$ , and  $\Psi_i$  are parameters inferred from the fit procedure to the position of HII regions, masers and GMCs. In addition to the 4 spiral arms, we also include the Local Spur. Table 3.2 shows the numerical values of the parameters in Eq. 3.21. On top of the spiral structure, we also take into account the structure of the innermost region. Here we consider the presence of the Galactic bar and the Near 3 kpc and Far 3 kpc arms. The first is modeled as an ellipse with an aspect ratio of 10:4 (length:width) having a half-length of 3.3 kpc and an inclination angle of 70°, defined counterclockwise with respect to the positive direction of the x-axis (or equivalently, 20° clockwise from the Galactic Center - Solar System connecting line, see Fig. 3.6) (Churchwell et al., 2009). The Near 3 kpc and Far 3 kpc arms are similarly modeled using an ellipse, with a semi-major axis of

Parameters	Arm 1	Arm 2	Arm 3	Arm 4	Local Spur
$R_i$ [kpc]	3.27	4.29	3.58	3.98	8.16
$\Psi_i \left[ \circ \right]$	9.87	10.51	10.01	8.14	2.71
$\theta_i  [^\circ]$	38.5	189	215.2	320.1	50.6

Table 3.2: Parameters values used to calculate Milky Way spiral arms (Hou & Han, 2014).

4.1 kpc, and an aspect ratio of 0.54 (a semi-minor axis of 2.2 kpc). The ellipse major axis orientation is 52° counterclockwise with respect to the positive direction of the x-axis (Green et al., 2011). Fig. 3.6 shows the resulting structure composed with the observed position of different tracers used by Hou & Han (2014). The positions of HII regions are recalculated using the same approach employed for the GMCs.

Let us now describe in detail the implemented process to allocate the synthetic YM-SCs in the Galaxy. For every j-th cluster, we start by randomly generating its radial  $(R_j)$ and angular coordinates  $(\theta_j)$ . Radial distances are extracted considering the probability distribution  $\rho(r)$  given by Eq. 3.20, while the angular coordinate is chosen assuming a uniform distribution. Afterward, based on the values of  $R_j$  and  $\theta_j$ , we associate the YMSC to a specific structure, following the criteria listed in Tab. 3.3. Namely, depending on the criterion met, we randomly select one specific structure among the included ones. The minimum value of r allowed for each arm corresponds to the position where the arm connects to the inner region of the Milky Way.

Note that the angular coordinate  $\theta_j$  is used only to check whether the YMSC should be associated with the Local Spur. Once the cluster is placed in a given arm, the coordinate  $\theta_j$  is recalculated by inverting Eq 3.21:

$$\theta_j = \left(\frac{\tan \Psi_i}{1 \operatorname{rad}}\right)^{-1} \ln \left(\frac{R_j}{R_i}\right) + \theta_i \,. \tag{3.22}$$

In case there is an association with Near 3 kpc and Far 3 kpc arms or the Galactic Bar, the association method is slightly more involved. To be precise, we first check which structure to associate the cluster with, following a minimum distance criterion. After that, if the cluster is associated with the Galactic Bar we check whether its position is actually located within the Bar. If not,  $\theta_j$  is varied until its position falls within the Galactic Bar. Note that, given the specific geometry of the Near and Far arms and the shape of the Galactic Bar, there are certain areas where YMSCs end up being closer to the Bar (and thus associated with it) but there are no values of  $\theta_j$  such that the cluster can be moved

	R	θ	
		$\theta < 360^{\circ}$	
Spiral Arm 1	$R \geq 3.27 \: \rm kpc$	(or $50^\circ > \theta > 110^\circ$	
		if $7.59 \text{ kpc} < R < 9.17 \text{ kpc}$ )	
		$\theta < 360^{\circ}$	
Spiral Arm 2	$R \geq 4.29 \ \rm kpc$	(or $50^\circ > \theta > 110^\circ$	
		if $7.59 \text{ kpc} < R < 9.17 \text{ kpc}$ )	
		$\theta < 360^{\circ}$	
Spiral Arm 3	$R \geq 3.58 \ \mathrm{kpc}$	(or $50^\circ > \theta > 110^\circ$	
		if $7.59 \text{ kpc} < R < 9.17 \text{ kpc}$ )	
		$\theta < 360^{\circ}$	
Spiral Arm 4	$R \geq 3.98 \ \mathrm{kpc}$	(or $50^\circ > \theta > 110^\circ$	
		if $7.59 \text{ kpc} < R < 9.17 \text{ kpc}$ )	
Local Spur	$7.59  \mathrm{kpc} < R < 9.17  \mathrm{kpc}$	$50^\circ \le \theta \le 110^\circ$	
NF 3kpc / Bar	$R < 4.29  \mathrm{kpc}$	$\theta < 360^{\circ}$	

Table 3.3: Criteria in radius and angle implemented to choose which structure to associate a<br/>given YMSC with.



Figure 3.6: Milky Way structure used in this work. The spiral arms (thick continuous gray lines) are modeled following the work of Hou & Han (2014). The dashed black line represents the Local Spur. Yellow and green ellipses represent the Galactic Bar (Churchwell et al., 2009) and the Near and Far 3kpc arms (Green et al., 2011), respectively. Also shown in the overlay are the positions of the GMCs (blue points, with sizes proportional to the GMC mass) and HII regions (red points) from the catalog provided by Hou & Han (2014). The yellow star marks the position of the Solar System.

within the Bar. In this specific case, we associate the cluster with the Near 3 kpc and Far 3 kpc arms.

When a cluster is associated with the Near 3 kpc and Far 3 kpc arms, we change its coordinates so as to have them within the ellipse that defines these structures. We do so by replacing the cluster coordinates with the ones of the nearest point of the ellipse. It should be emphasized that this procedure induces a distortion in the starting radial distribution. However, the deviation found is totally negligible for the final result.

Once all clusters are associated we proceed to perturb their positions according to a Gaussian distribution, following the fundamental concept that both Spiral Arms and Near and Far structures possess an intrinsic thickness. For spiral arms, we extract a nonconstant radial fluctuation, which increases with the galactocentric distance such that the probability  $\mathcal{P}$  of having a certain fluctuation  $\Delta R_i$  is (Faucher-Giguère & Kaspi, 2006):

$$\mathcal{P}(\Delta R_j) = \frac{1}{2\pi (0.07R_j)} e^{-\frac{\Delta R_j^2}{2(0.07R_j)^2}}.$$
(3.23)

As for the clusters belonging to the Near and Far arms, we extract from a gaussian probability distribution the galactic x and y coordinates with a spread of  $\sigma_x = \sigma_y = 0.1667$ kpc, such that the probability at  $3\sigma$  returns a scattering compatible with the observed radial thickness of 0.5 kpc.

Finally, after having set the position of all YMSCs in the Galactic plane, we generate the vertical coordinate (*z*) following the observed gas distribution profile, i.e. an exponential distribution ( $\rho(z)$ ) with a characteristic spread of 100 pc (Strong et al., 2000):

$$\rho(z) = \exp\left(-\frac{z}{100\,\mathrm{pc}}\right) \tag{3.24}$$

## 3.2.3 Wind luminosity of star clusters

To model the wind from each YMSC, first, its stellar population must be generated following the description given in § 3.1. Knowing the wind luminosity  $(L_{\star,i})$  and mass loss rate  $(\dot{M}_{\star,i})$  of each i-th star of the cluster, then, the luminosity  $(L_w)$  and mass loss rate  $(\dot{M})$  of the collective cluster wind are readily obtained as:

$$L_w = \sum_i L_{\star,w} \tag{3.25a}$$

$$\dot{M} = \sum_{i} \dot{M}_{\star,i} \tag{3.25b}$$

With these parameters, one can also estimate the wind speed as:

$$v_w = \sqrt{\frac{2L_w}{\dot{M}}} \tag{3.26}$$

#### **3.2.4** Statistical properties of the YMSC synthetic population

With all the ingredients in our hands, we can finally generate a synthetic population of Galactic YMSCs and examine their general properties. After normalizing the cluster distribution function using Eq. 3.19 with the most updated value of the star formation rate (Bonatto & Bica, 2011), we obtain from Eq. 3.12 a total number of 747 YMSCs. Fig. 3.7 shows the resulting spatial distribution for a specific realization of the Galactic cluster population. Unless differently stated, we will use this specific synthetic population for the rest of the chapter.



Figure 3.7: Spatial distribution of a single realization of the YMSC Galactic population. Different colors correspond to different Galactic structures. The size of the dots is directly proportional to the mass of the clusters. The yellow star marks the Sun position.

The main parameters that determine the effectiveness of a cluster as a particle accelerator are the wind luminosity, along with the mass loss rate and the cluster wind speed. Fig. 3.8 shows the distribution of such values considering both the case when the maximum stellar mass depends on the cluster mass and the case when it is fixed to 150 M<sub> $\odot$ </sub>. Noticeably, although the latter case produces, as expected, distributions with more pronounced high-value tails, the parameters average values are almost unchanged and equals to  $\dot{M} \approx 10^{-6} \text{ M}_{\odot} \text{ yr}^{-1}$ ,  $\overline{v_w} \approx 2800 \text{ km s}^{-1}$  and  $\overline{L_w} \approx 3 \times 10^{36} \text{ erg s}^{-1}$ .



Figure 3.8: Distribution of the mass loss rate (left panel), wind luminosity (central panel), and collective cluster wind speed (right panel) for the cases of constant (step histograms) and non-constant (filled histograms) maximum stellar masses. Continuous and dashed lines represent the average values for the two different cases (see plot legend).

For the sake of completeness, we also show the trend of wind luminosity and mass loss rate as a function of mass (Fig .3.9) and cluster age (Fig .3.10). As can be easily guessed, both wind power and mass loss rate are roughly proportional to the cluster mass, as the content of massive stars increases with the latter. Similarly, after  $\sim 2$  Myr, the two values start to decrease with increasing cluster age, since the number of massive stars decreases as they start exploding as supernovae.

An interesting point worth outlining is the difference between the two cases of high stellar mass cutoff under analysis. A constant maximum stellar mass of 150  $M_{\odot}$  is able to produce clusters characterized by higher wind powers even for low cluster masses.



Figure 3.9: Wind power (top row) and mass loss rate (bottom row) as a function of the cluster mass for the synthetic population shown in Fig. 3.7. The left column shows the case where the maximum stellar mass  $(M_{\star,max})$  is fixed to  $150 \text{ M}_{\odot}$ , while the right column is for a maximum stellar mass dependent on the cluster mass. In all the plots, the marker size is proportional to the number of stars with  $M_{\star} > 20 \text{ M}_{\odot}$ . For comparison purposes, the properties of Cygnus OB2 along with their uncertainties obtained in § 2.1.1 are indicated with a black cross(Wright et al., 2015).



Figure 3.10: Same as Fig. 3.9, but the wind luminosity and mass loss rate are plotted as a function of the cluster age. Grey continuous line show for comparison the turn-off time of stars with masses of 100, 50, and 30  $M_{\odot}$ .

Along with the physical properties of the stellar clusters, it is worth investigating the dimensions of the associated wind-blown bubbles. To calculate the size of the TS ( $R_{TS}$ ) and the forward shock ( $R_b$ ), we need to estimate the interstellar medium density surrounding our YMSCs (see Eq. 1.11 and Eq. 1.8 respectively). As we stated in § 1.2.1, YM-SCs are likely to be still partially (or completely) embedded in their parent GMC, so the average medium density ( $\rho_0$ ) is expected to be higher than the average density of the interstellar medium.

To compute a rough estimate of  $\rho_0$ , we can use again the GMC catalog provided by Hou & Han (2014), which additionally reports the mass and the angular size of the clouds. Considering the cloud distances calculated in § 3.2.2, from the angular size we can estimate an average cloud physical radius, from which, assuming spherical geometry, we infer the average gas density. Fig. 3.11 shows the distribution of the particle number density for the clouds in the catalog of Hou & Han (2014), characterized by an average value of  $\sim 8.76 \text{ cm}^{-3}$ , with a spread of  $^{+31.62}_{-9.96} \text{ cm}^{-3}$ .



Figure 3.11: Particle density distribution in GMCs derived from the catalog provided by Hou & Han (2014). The solid line shows the distribution average value, while the dotted and dash-dotted lines are the particle density obtained for clusters with masses of  $10^3$  and  $10^4$  M<sub> $\odot$ </sub> respectively (see text).

There is a second less direct way to infer the clouds density, that we can use as double check. In principle, the environmental density surrounding a given cluster, which is linked to the GMC mass, should be correlated to the cluster mass. In § 1.2.1 we saw that for a given star formation efficiency  $\epsilon_{SFE}$  (usually of the order of a few percent), the mass of a cluster is linked to the mass of the parent cloud as:

$$M_{GMC} = M_{YMSC} \left(\frac{1}{\epsilon_{SFE}} - 1\right) \,. \tag{3.27}$$

The density of the cloud as a function of the cluster mass is then:

$$n_0 = \frac{M_{YMSC}}{\frac{4}{3}\pi R_{GMC}^3} \left(\frac{1}{\epsilon_{SFE}} - 1\right).$$
(3.28)

where  $R_{GMC}$  is the radius of the cloud, which also depends on the GMC mass. Several mass-radius relations for the clouds have been proposed over the decades. In general,

all relations are parametrized as power laws:

$$R_{GMC} = R_0 \left(\frac{M_{GMC}}{M_0}\right)^{\alpha}$$
(3.29)

with  $R_0$  and  $M_0$  as normalization constants. Fig. 3.12 shows some of the clouds massradius relations taken from the literature (Larson, 1981; Miville-Deschênes et al., 2017; Chen et al., 2020) and compared with mass and radius of the sample of GMCs from Hou & Han (2014). If we consider the relation provided by Miville-Deschênes et al. (2017)  $(M_0 = 36.7 \text{ M}_{\odot}, R_0 = 1 \text{ pc}, \alpha = 0.454)$ , which seems to be the one best reproducing the observed sizes and masses, using Eq. 3.28 (assuming  $\epsilon_{SFE} = 0.01$ ) leads to densities of  $\sim 10 \text{ cm}^{-3}$  and  $\sim 4 \text{ cm}^{-3}$  for  $M_{YMSC} = 10^3 \text{ M}_{\odot}$  and  $M_{YMSC} = 10^4 \text{ M}_{\odot}$  respectively. These densities are fully compatible with the estimated mean value of 10 cm<sup>-3</sup> obtained above. Taking this into account, and given that by changing the mass of the cluster by an order of magnitude the density only varies by a factor of  $\sim$ 2, we can safely consider for our purposes a constant density equal to  $\rho_0 = 10m_p \text{ cm}^{-3}$  for all YMSCs.



Figure 3.12: GMC masses vs radii for the considered sample (Hou & Han, 2014). The solid line shows the first relation obtained by Larson (1981). The dot-dashed and dashed lines are observed relations provided by Chen et al. (2020) and Miville-Deschênes et al. (2017)
 respectively. The color scale associated to the symbols represents the cloud distances from the Sun.

Once the environmental density is set, we can readily calculate the sizes of the windblown bubbles associated with our synthetic population of YMSCs. Fig. 3.13 shows the resulting distribution. The distribution peaks at  $R_{TS} = 4$  pc and  $R_b = 50$  pc and is not Gaussian but has a long tail extending towards higher values. The average values are  $\overline{R}_{TS} = 6$  pc and  $\overline{R}_b = 60$  pc. Again, changing the maximum stellar mass does not significantly affect the distributions.



Figure 3.13: Size distribution of the TS and forward shock for both cases under analysis, namely with maximum stellar mass fixes (step histogram) and dependent on the cluster mass (filled histogram). Solid and dashed lines represent the distributions average values for the filled and step histograms respectively.

## **3.3 Cosmic Ray distribution in YMSCs**

Once the YMSC distribution is determined, we have all the ingredients to compute the CR distribution. Once again, we rely on the model of hadronic CR acceleration at the wind TS developed by Morlino et al. (2021) and described in § 1.3.3.

In § 1.2.1 and § 1.3.3 we emphasized that the validity of this model is bound to the existence of a collective cluster wind. We recall that the general rule for this to happen is that the average distance between stars has to be smaller than the size of the TS. When this condition is satisfied, we call the stellar cluster "compact". In the opposite cases we speak of "loose" clusters, defined as the ones where the collective TS will not be formed. An approximate way to distinguish between compact and loose stellar clusters is by using the half mass radius, which is, however, a non trivial quantity to estimate from observations. After a comprehensive analysis of the literature, Pfalzner et al. (2016) found a power-law relation of the form:

$$R_{YMSC} = \left(\frac{M}{359 \,\mathrm{M}_{\odot}}\right)^{0.585} \,\mathrm{pc.}$$
 (3.30)

Fig. 3.14 shows Eq. 3.30 compared to the sizes of  $R_{TS}$  of our synthetic population plotted

as a function of the cluster mass. Noticeably, for a significant fraction of YMSCs the TS radius is always larger than the cluster size. This condition seems to be violated by some fraction of clusters with masses larger than  $10^4 \text{ M}_{\odot}$ . This would lead one to think that for some very massive clusters the formation of a collective wind could be suppressed. However, one still needs to take into account the effect of mass segregation, which tends to aggregate the most massive stars (which are the largest contributors to the creation of the collective wind) toward the center of the cluster. In 90% of the generated synthetic YMSCs, stars with masses greater than 10 M<sub> $\odot$ </sub> contribute  $\sim 90\%$  of the cluster wind power. It follows therefore that, instead of the half-mass radius, one should rather consider an effective radius  $R_{\text{eff}}$  defined such that it contains a significant fraction of stars above 10 M<sub> $\odot$ </sub>. If we fix this fraction to 90%, defining the distribution of stars within a cluster as a function of the mass and radius as  $\Xi(M_{\star}, r) \equiv \frac{dN_{\star}}{dM_{\star}dr}$ , the effective radius can be found by requiring:

$$\int_{10M_{\odot}}^{150M_{\odot}} \int_{0}^{R_{\rm eff}} \Xi(M_{\star}, r) dM_{\star} dr = 0.9 ; \qquad (3.31)$$

where we have assumed  $\int \int \Xi(M_{\star}, r) dM dr = 1$ . Unfortunately, the function  $\Xi(M_{\star}, r)$  turns out to be difficult to estimate. However, we can calculate  $R_{eff}$  as the containment radius of the fraction of cluster mass due to stars with  $M_{\star} > 10 \text{ M}_{\odot}$ . Such a fraction is:

$$\frac{\int_{10}^{150} M_{\odot}}{\int_{0.08}^{150} M_{\odot}} M_{\star} f(M_{\star}) dM_{\star}}{2000} \simeq 0.2$$
(3.32)

Assuming that the stellar mass distribution is  $dM_*/dr \propto r^{-\alpha}$  one has  $R_{eff} < R_{half mass}$ (unless  $\alpha \approx 3$ ). If  $\alpha$  is such that  $R_{eff} \leq 0.4R_{half mass}$  (with  $R_{eff} = 0.4R_{half mass}$  if  $\alpha = 2$ ), then, almost all stellar clusters turn out to have a TS more extended than the region containing the most massive stars.  $R_{eff}$  estimated in this way is reported in Fig. 3.14 with a dashed line. Summarizing, the conclusion of our calculation is that all YMSCs in our synthetic population can generate a TS.

Now we can proceed to the calculation of the CR distribution both in terms of morphological and spectral shape. As illustrated in Ch. 2, for an individual cluster the CR distribution will depend on two factors, namely the injection efficiency and the type of plasma turbulence that will determine the diffusion properties in the system. For particle injection at the TS, we assume a fixed spectral slope s = 4, typical of strong shocks, and acceleration efficiency ( $\epsilon_{CR}$ ) equal to 10% of the wind luminosity. Notice that the final result on the  $\gamma$ -ray emission will scale almost linearly with  $\epsilon_{CR}$ . Regarding the plasma turbulence, given the results obtained in Ch. 2, we again decide to investigate the two scenarios characterized by the Kraichnan and Bohm-like diffusion.



Figure 3.14: Termination shock radius as a function of the cluster mass. Continuous line shows the cluster mass-half mass radius relation provided by Pfalzner et al. (2016). The left panel is for the case of a synthetic Galactic cluster population realized with a fixed maximum stellar mass (150  $M_{\odot}$ ), while the right panel shows the case with maximum stellar mass dependent on the cluster mass.

A relevant aspect worth investigating is the distribution of the maximum particle energy reached in our synthetic population of CR factories. The maximum energies for the Kraichnan and Bohm cases are described by Eq. 1.86 and Eq. 1.84, respectively. We assume once more that a 10% of the clusters wind luminosity is converted into turbulent magnetic field ( $\eta_B = 0.1$ ). Note that in the case of Kraichnan turbulence, the maximum energy also depends on the injection scale of the turbulence, which is totally unknown and hard to estimate. Consequently, we decide to further split the Kraichnan scenario into two subcases, defined by an injection length scale fixed to 1 pc or to 10% of  $R_{TS}$  respectively. The same also holds for the Bohm case. However here the dependence is not on a single length scale but rather on the interval of length scales for which the turbulence power is uniformly injected. We assume in this case that turbulence is constantly injected from 10% of  $R_{TS}$  to  $10^{-5}$  pc. The results are shown in Fig. 3.15. We can see that, when the maximum stellar mass as fixed to 150 M<sub> $\odot$ </sub> the distribution of  $E_{max}$  is slightly more skewed towards high values. However, the mean values for the maximum energy are basically the same, namely  $E_{max}^{Kra}(L_{inj}=1 \text{ pc}) \approx 24 \text{ TeV}, E_{max}^{Kra}(L_{inj}=0.1R_{TS}) \approx 37$ TeV and  $E_{max}^{Bohm} \approx 36$  TeV.

Noticeably, no PeVatrons are found. This is true both when considering Kraichnan

and Bohm like turbulence, but also in all considered scenarios with different turbulence injection scales and maximum stellar masses. In fact, we found that the fraction of clusters with  $E_{max} > 100$  TeV is 13% and 7% for the Bohm and Kraichnan cases respectively. For  $E_{max} > 500$  TeV these fractions reduce to 1% for the Bohm and 0.5% for the Kraichnan case, corresponding to  $\sim 7 - 5$  clusters in total. These values do not vary a lot when considering the various scenarios under analysis, i.e. different turbulence injection scales for the Kraichnan case different maximum stellar mass. With such a small expected number of PeVatrons, the statistical fluctuations are non negligible, and multiple realizations of the Galactic population of YMSCs are needed to asses the contribution of YMSCs to CRs at energies above 100 TeV.



Figure 3.15: Distribution of maximum energies obtained in the case where the maximum stellar mass is set to 150  $M_{\odot}$  (left panel) or dependent on the mass of the host cluster (right panel). Filled and step-dashed histograms represent maximum energy in the case of Kraichnan turbulence injected at a characteristic length scale of 0.1  $R_{TS}$  and 1 pc respectively. Step-continuous histogram reports instead the maximum energy distribution for Bohm-like diffusion. Vertical lines show the distribution mean value (legend on the right panel).

## **3.4** $\gamma$ -ray emission from single sources

Now we are ready for the final step which is the estimate for the  $\gamma$ -ray emission from YM-SCs. As already stated, we will consider only the hadronic mechanism of  $\pi_0$  production and decay (see § A.1).

Since we are not interested in the analysis of individual sources, but rather in the study of the total diffuse emission of the population, we perform a simplified calculation of  $\gamma$ -ray emission from a single cluster. More precisely, we model the emission from a single source without accounting for its morphology, namely by considering only the

total flux obtained from all particles confined within the wind-blown bubble and assuming that the emission is uniformly distributed in the projected disk in the sky. Following this approach, the  $\gamma$ -ray flux coming from the i-th cluster will be:

$$\phi_{\gamma,i}(E_{\gamma}) = \frac{cn_i}{4\pi d_i^2} \int F_{CR}(E_p) \frac{d\sigma(E_p, E_{\gamma})}{dE_p} dE_p$$
(3.33)

where  $n_i$  is the target medium numerical density,  $d_i$  is the distance from the Sun of the i-th YMSC,  $\sigma$  is the cross section for  $\gamma$ -ray production from p - p interaction (Kafexhiu et al., 2014), and  $F_{CR}(E_p)$  is defined as:

$$F_{CR}(E_p) = 4\pi \int_0^{R_{b,i}} r^2 f_{CR}(r, E_p) dr$$
(3.34)

with  $f_{CR}(r, E_p)$  calculated using Eq. 2.9 after the appropriate transformation from particle momentum to energy. Note that by using Eq. 2.9, we are again accounting only for the contribution of freshly accelerated particles without considering the emission from Galactic CRs penetrating in the wind-blown bubble.

In Eq. 3.33, the density of the target medium  $n_i$  remains a difficult parameter to estimate, as it can vary depending on the evolutionary conditions of the wind bubble. In general, one could consider the density obtained from the evaporation rate of the swept-up shell (see Eq. 1.24) in addition to the material injected by the stellar winds. However, for the sake of simplicity and with the aim of proving a considerable but solid upper limit for the  $\gamma$ -ray flux<sup>4</sup>, we will consider here the simplest case of a constant target density equal to that of the surrounding environment, i.e.,  $n_0 = 10 \text{ cm}^{-3}$ .

## 3.5 Diffuse emission and comparison with data

Once the flux of all YMSCs is obtained, we calculate the diffuse  $\gamma$ -ray emission of a given area of the sky by simply summing up all the contributions of the sources included in that region. This approach is computationally convenient. However, the final result may be slightly distorted as we are not excluding the emission from those regions of the wind bubbles that do not fall in the considered sky area. The opposite situation is also not considered, namely the possibility of including the contribution of portions of bubbles that fall within the region of interest whose centroid is instead outside the area of the sky under analysis. However, we expect the two contributions to cancel each other out, so that on average the final estimation does not differ much from the true result.

We select the regions from which to extract the  $\gamma$ -ray emission considering four dif-

<sup>&</sup>lt;sup>4</sup>Under the assumption of a fixed acceleration efficiency of the particles.

ferent works. The first is the one by Yang et al. (2016), who supply the diffuse  $\gamma$ -ray spectra in several areas of the Galactic Plane using Fermi-LAT observations. Among the provided area, we select the one between  $15^{\circ} < l < 5^{\circ}$  and  $-5^{\circ} < b < 5^{\circ}$ . Fig. 3.16 shows the selected region of interest and the synthetic YMSC within it. For simplicity, we will refer from now on to this region as ROI1. The remaining three studies that we considered are Hunter et al. (1997), Bartoli et al. (2015), and Amenomori et al. (2021). These works report the diffuse  $\gamma$ -ray emission within a region spanning  $100^{\circ} < l < 25^{\circ}$  and  $-5^{\circ} < b < 5^{\circ}$ , employing data from EGRET, ARGO, and Tibet-AS $\gamma$ . Fig. 3.17 shows the latter region and the synthetic YMSCs within it. We will refer to this region from now on as ROI2.



 $log_{10}L_{\gamma}(E>1 \text{ GeV}) [erg s^{-1}]$ 

Figure 3.16: Region of interest (R0I1) for the computation of the  $\gamma$ -ray diffuse emission using the work by Yang et al. (2016). Circles represent the wind-blown bubbles from the synthetic YMSC. The color scale reports the integrated  $\gamma$ -ray luminosity above 1 GeV considering the Kraichnan scenario with maximum stellar mass as a function of the cluster mass. For comparison, we report the CTA point spread function at 10 TeV (CTA Collaboration, 2023).



Figure 3.17: Same as Fig. 3.16, but for the region of interest defined between  $100^{\circ} < l < 25^{\circ}$  and  $-5^{\circ} < b < 5^{\circ}$  (ROI2).

As already mentioned, in this work we consider just one realization of the Galactic YMSC population and calculate  $\gamma$ -ray emission on top of it. Fig. 3.18 shows the  $\gamma$ -ray spectral energy distributions compared with observations in ROI1 for all the cases under analysis. There are several things worth discussing. The most important aspect is that the flux obtained from all the cases analyzed lies below the Fermi-LAT data by a factor of  $\sim 2-3$  for the Bohm scenarios and  $\sim 10$  for the Kraichnan cases at  $E_{\gamma} > 10$  GeV. This is a remarkable result, as it implies that the way we modeled the YMSCs does not lead to a scenario inconsistent with observations and yet shows that YMSCs could be an important contributors of the diffuse emission in this band. A second result to underline is that, changing recipe for the maximum stellar mass does not lead to significant changes in the overall emission, only resulting in a normalization shift by a factor of  $\sim 1.5$ . This is due to a greater number of massive stars in the lower-mass star clusters, which produce a higher wind power and thus higher flux normalization. As we will see in short, the shift in amplitude is relatively small, as the contribution to the diffuse  $\gamma$ -ray is mainly due to the most massive stellar clusters.

Remarkable, and again not unexpected, turns out to be the difference between models using different diffusion coefficients. Bohm-like diffusion is capable of producing  $\gamma$ -ray emission up to very high energy, with a cut-off in the spectrum appearing at  $\sim 1$ TeV, contrary to the Kraichnan case where the emission start to fade at  $\sim 100$  GeV. Concerning the Kraichnan case, having a constant injection scale fixed to 1 pc produces a spectrum with a cut-off at higher energy compared to the case where the turbulence in-
jection occurs at 10% of  $R_{TS}$ , although, in the end, the overall spectra do not differ too much.

Regarding the cut-off position in the  $\gamma$ -ray spectrum, we stress once more time the fact that the energy at which it appears is less than the expected 0.1  $E_{max}$  of accelerated particles. As already discusses in § 2.5.1, the maximum energy calculated in § 1.3.3 does not report the exact location of the cutoff in the injected particles spectrum, but is usually shifted to lower energy due to the spherical geometry of the system.



Figure 3.18: Diffuse  $\gamma$ -ray emission from the synthetic population of YMSCs compared to Fermi-LAT observations in ROI1 (sky region shown in Fig. 3.16). Thin and thick lines represent the spectra after considering constant and cluster mass-dependent maximum stellar masses. Dash-dotted lines are the spectra obtained in the case of Bohm-like turbulence. Solid and dashed lines are instead the spectra calculated for Kraichnan diffusion after considering  $L_{inj} = 1$  pc and  $L_{inj} = 01R_{TS}$  respectively.

In general, it is important to note that the Krainchnan case returns an overall spectrum that is softer than the Bohm model, despite the same spectral slope at the acceleration site. This is a direct consequence of the interplay between advection and diffusion, whose relative importance differs in the two cases under analysis. To show the importance of diffusion, we can compare the full solution with the idealized case completely dominated by advection, which has  $f_{CR}$  inside the bubble equal to  $f_{TS}$ . As an example, in Fig. 3.19 we consider the case of the most massive YMSCs of our synthetic population, and assuming the scenario where the maximum stellar mass is fixed to 150 M<sub> $\odot$ </sub>. One can readily note that the particle spectrum in the Kraichnan model is much softer than in the ideal situation of pure advection, which is characterized by the expected power law  $f_{TS} \propto E_p^{-2}$ . In the case of Bohm-like diffusion, advection is expected to dominate over diffusion up to very high energy. Therefore, the CR spectrum does not differ much from the ideal situation of pure advection (except close to the cut-off energies). As a consequence, the observed  $\gamma$ -ray spectrum is harder than the one obtained considering Kraichnan-like diffusion.



Figure 3.19: CR spectra ( $F_{CR}$ ) used to compute the  $\gamma$ -ray radiation (dotted and continuous lines for Kraichnan and Bohm cases respectively) compared to hypothetical spectra where the propagation mechanism is totally advection-dominated (dash-dotted and dashed lines for Kraichnan and Bohm cases respectively). All curves reported refer to the most massive stellar cluster in our synthetic population.

Fig. 3.20 shows the diffuse  $\gamma$ -ray spectrum in ROI2 compared to EGRA, ARGO and Tibet-AS $\gamma$  observations. Similarly to ROI1, the emission at low-energy ( $E_{\gamma} < 10$  GeV), in all the considered cases under analysis, is an order of magnitude below the observed spectrum. On the contrary, above  $E_{\gamma} \gtrsim 300$  GeV the flux for the Bohm case, regardless the maximum stellar mass considered, is found to be higher than the observed spectrum by Argo by a factor  $\sim 3$  between 0.2 TeV  $\lesssim E_{\gamma} \lesssim 1$  TeV and  $\sim 6$  at  $E_{\gamma} \approx 1-2$  TeV. For the

Kraichnan case, the flux is instead below the observed flux by a factor of  $\sim 2$  at 0.2 TeV  $\lesssim E_{\gamma} \lesssim 1$  TeV, while at  $E_{\gamma} \approx 1 - 2$  TeV the expected flux matches the observations. This is true independently on the chosen turbulence injection scale and the maximum stellar mass in the cluster. This is indeed an intriguing result, indicating that the non-resolved diffuse emission from YMSCs between a few hundreds of GeV and a few TeV is likely not negligible and possible even dominant.

We would like to emphasize that the outcomes we obtained were based on the assumption of high efficiency and high target density. It is possible to adjust these two parameters to reduce the flux and achieve consistency with Argo data, even in the Bohm scenario. Fig. 3.21 shows the same spectra after assuming an efficiency of CR acceleration of  $\epsilon_{CR} = 0.01$ . Concerning the emission at very-high energy ( $E_{\gamma} > 100$  TeV), even considering the most optimistic scenario of a Bohm-like diffusion, the expected flux is significantly below the observations of Tibet-AS $\gamma$ . As already discus, at these energies however, the number of contributing cluster is so low that no conclusion can be derived based on one single realization of the galactic population of YMSCs.



Figure 3.20: Same as Fig. 3.18, but the spectral energy distributions are calculated considering R0I2 (sky region shown in Fig. 3.17). The crosses, square and triangles marks the emission observedy by EGRET Hunter et al. (1997), ARGO Bartoli et al. (2015) and Tibet-AS $\gamma$  Amenomori et al. (2021) respectively.



Figure 3.21: Same as Fig. 3.20, but the spectral energy distributions are calculated considering an efficiency of CR acceleration of  $\epsilon_{CR} = 0.1$ .

Finally, for the sake of completeness, one last crucial aspect should be discussed. One may wonder whether the total  $\gamma$ -ray emission is mainly contributed by the most numerous and less massive stellar clusters or by the less numerous and most massive ones. The answer is the latter. In fact, the total  $\gamma$ -ray luminosity differentiated with respect to the cluster mass is:

$$\frac{dL_{\gamma}}{dM} = L_{\gamma}(M) \frac{dN_{YMSC}}{dM}$$
(3.35)

where  $\frac{dN_{YMSC}}{dM} \propto M^{-1.53}$  is the cluster mass function and  $L_{\gamma}(M)$  is the  $\gamma$ -ray luminosity for a given cluster with mass M. As the  $\gamma$ -ray luminosity is dominated by the emission at low energy, one can write:

$$L_{\gamma}(M) \approx \frac{4\pi}{3} R_b^3 n_0 f_{TS} \sigma_{pp \to \gamma} \propto f_{TS} R_b^3 n_0 \tag{3.36}$$

where  $\sigma_{pp\to\gamma}$  is the integrated cross section for hadronic  $\gamma$ -ray emission. The distribution at the TS is given by Eq. 1.75, which scale as:

$$f_{TS} \propto \epsilon_{CR} n_1 v_w^2 \propto \dot{M} R_{TS}^{-2} v_w \epsilon_{CR}$$
(3.37)

where in the last step we rewrote  $n_1$  as  $n_1 = \dot{M}/4\pi R_{TS}^2 v_w$ . Knowing from Eq. 1.8 and Eq. 1.11 that:

$$R_b \propto L_w^{1/5} n_0^{-1/5} t^{3/5} \tag{3.38a}$$

$$R_{TS} \propto \dot{M}^{1/2} v_w^{1/2} L_w^{-1/5} n_0^{-3/10} t^{2/5} , \qquad (3.38b)$$

and considering Eq. 3.37, we can rewrite Eq. 3.36 as:

$$L_{\gamma}(M) \propto L_w t n_0$$
 (3.39)

The wind luminosity has a quasi-linear dependence on the cluster mass  $L_w \propto M$ , with this acknowledged and using Eq. 3.39, Eq. 3.35 finally reads:

$$\frac{dL_{\gamma}}{dM} \propto M^{-0.5} t n_0 \,. \tag{3.40}$$

If we then consider a wide interval of masses, ranging between some  $M_{min}$  and  $M_{max}$  with  $M_{min} \ll M_{max}$ , the total  $\gamma$ -ray luminosity will be:

$$L_{\gamma} \propto \int_{M_{min}}^{M_{max}} M^{-0.5} \propto M_{max}^{0.5}$$
(3.41)

hence, dominated by the most massive stellar clusters. The above estimates can be verified plotting the  $\gamma$ -ray contribution of our synthetic population by mass intervals for ROI1. This is shown in Fig. 3.22, where we have chosen three different mass intervals that contain the same number of stellar clusters, namely  $M < 1.6 \times 10^3 \text{ M}_{\odot}$ ,  $1.6 \times 10^3 < M/\text{M}_{\odot} < 3.9 \times 10^3$  and  $M > 3.9 \times 10^3 \text{ M}_{\odot}$ . Even considering the most extreme case of constant maximum stellar mass, for both Kraichnan and Bohm cases the low mass clusters do not contribute much to the total diffuse emission: the contributions of the three different mass intervals in the total  $\gamma$ -ray flux at  $\sim 10$  GeV is 9%, 21% and 70% respectively. A different way to express the same concept is that 50 % of the total  $\gamma$ -ray flux is contributed by stellar clusters with  $M > 1.2 \times 10^4 \text{ M}_{\odot}$ , which are only  $\sim$ 20% of the total ensemble (see dashed and dotted black lines in Fig. 3.22)

#### 3.6 Final remarks

In this chapter we described the method for generating a synthetic galactic population of YMSCs. The procedure is complex and requires the knowledge or the assumption of several ingredients, starting from how we model stars within each cluster and ending with the rules for generation of the population itself. The final goal was to obtain an estimate



Figure 3.22: Diffuse  $\gamma$ -ray emission from clusters in three diverse mass ranges (see legend in each panel). The dashed and dotted lines are the contribution of clusters above and below  $10^4$  M<sub> $\odot$ </sub> respectively.

of the diffuse  $\gamma$ -ray emission from the combination of multiple unresolved sources under the assumption of particle acceleration at the cluster wind TS, and to compare this emission with observations available in the literature. To this purpose, we considered two different regions of interest.

The first one, named ROI1, is defined as  $15^{\circ} < l < 5^{\circ}$  and  $|b| < 5^{\circ}$ . Here, the measurement of the  $\gamma$ -ray diffuse emission is provided by the Fermi-LAT telescope. We obtain for a single realization of the galactic YMSC population a diffuse  $\gamma$ -ray emission consistent with observations. To be more specific, the contribution from stellar clusters in ROI1 is a factor 2-3 below the data at  $E_{\gamma} \gtrsim 10$  GeV considering the Bohm scenarios. For the Kraichnan case, the expected flux is a factor  $\sim 10$  below the observed flux at  $E_{\gamma} \gtrsim 10$  GeV. These result remains unchanged when considering different maximum stellar masses in the clusters, or different injection scales of the magnetic turbulence.

In the second region analyzed, named ROI2, and defined by  $100^{\circ} < l < 25^{\circ}$  and  $|b| < 5^{\circ}$ , observations of the diffuse  $\gamma$ -ray emission are provided by EGRET, ARGO and Tibet-AS $\gamma$ . We found here that the expected emission in the range  $\sim 0.3 - 1$  TeV is comparable with the observed flux from ARGO. More precisely, the expected emission overshoot by a factor  $\sim 2 - 3$  the observations by ARGO when considering the Bohm scenario, while is consistent at 1 TeV when considering the Kraichnan case. In general, full consistency can be recovered for the Bohm case by changing the product between  $n_0$  (assumed to be  $10 \text{ cm}^{-3}$ ) and  $\epsilon_{CR}$  (assumed to be 0.1). We conclude that the emission at  $\sim 1$  TeV from YMSCs is not negligible and that the observed diffuse emission could be even dominated by stellar clusters.

Beyond results concerning the  $\gamma$ -ray emission, our study contains several novelties from the point of view of stellar clusters. In fact, the generation of a population of YMSCs

is already per se an interesting result, as it allows one to study the average properties of the Galactic clusters. For example, we found mean values of cluster wind luminosity and mass loss rate of  $\overline{L_w} \approx 3 \times 10^{36} \text{ erg s}^{-1}$  and  $\overline{\dot{M}} \approx 10^{-6} \text{ M}_{\odot} \text{ yr}^{-1}$  respectively. Furthermore, our analysis revealed that the synthetic population of YMSCs contains 157 clusters with masses exceeding  $10^4 \text{ M}_{\odot}$ , which is approximately ten times larger than the number of Galactic clusters observed within this same mass range (Portegies Zwart et al., 2010).

Although highly simplified, the modeling of stellar parameters ultimately reproduces properties of winds that are fairly consistent with observations for high-mass stars.

From the point of view of YMSCs as particle accelerators, there are two main interesting aspects to emphasize. First, the validity of the acceleration model at the wind TS. Considering the effect of mass segregation, the establishment of a collective cluster wind is a likely scenario, making the model of particle acceleration we consider a physically motivated choice. A second important finding is, the distribution of the maximum energy of freshly accelerated particles obtained for the two considered types of plasma turbulence spectra. We found mean maximum energies of  $E_{max}^{Kra} \approx 10-20$  TeV, and  $E_{max}^{Bohm} \approx 35$  TeV for Kraichnan and Bohm-like diffusion. No PeVatrons are found in this specific realization of the Galactic population, for both the considered models of particle diffusion.

The main limitation of our work is neglecting the contribution of SN explosions inside stellar clusters. For an age of  $\gtrsim 3$  Myr we know that those events occur and they will probably dominate the energetics of the bubble (Vieu et al., 2022). In this respect, our result for the  $\gamma$ -ray flux should be regarded as a lower limit in that SN explosions are likely to enhance the production of CRs. The reason why we have neglected SN explosions is because the modeling of particle acceleration ath the SNR shock propagating inside the wind bubble is not very well developed. One of the uncertainty is related to the SN shock Mach number, which is related to the particle acceleration efficiency by the shock. Let us consider, for example, the first SN exploding in the star cluster. The resulting SNR will reach the Sedov phase soon after the interaction with the wind TS. The Mach number of the shock in the hot bubble is  $\mathcal{M} = v_{sh}/c_s$ , where  $v_{sh}$  is the SN shock speed and  $c_s = \sqrt{\gamma k_B T_2/m_p}$  is the sound speed, with  $\gamma$  as the adiabatic index. Neglecting possible effects of cooling, the temperature in the bubble is determined by the TS, hence,  $k_B T_2 = 3/16(m_p v_w^2)$ , so that the Mach number is:

$$\mathcal{M} = \frac{v_{sh}}{c_s} = \frac{v_{sh}}{\sqrt{\gamma \frac{3}{16} v_w^2}} = \sqrt{\frac{16}{5}} \frac{v_{sh}}{v_w} \simeq 4 \left(\frac{v_{sh}}{5000 \,\mathrm{kms^{-1}}}\right) \left(\frac{v_w}{2000 \,\mathrm{kms^{-1}}}\right)^{-1}$$
(3.42)

Hence we see that standard values for  $v_{sh}$  and  $v_w$  give  $\mathcal{M} \simeq a f e w$ , which is not enough to generate efficient particle acceleration. The effect of cooling in the bubble could change

the conclusion. However, subsequent SNRs will expand inside a material heated by the fist SN, hence again, the value of the Mach number should be estimated carefully.

#### **3.6.1** Future prospects

The results presented so far are the outcome of work still in progress, for which some are to be improved, possibly leading to additional and more significant conclusions.

Some of these aspects are listed below:

- From the point of view of stellar physics, rather than using empirical relations like we did here, we could include robust stellar models. In principle, we could generate tables of stars with different masses and ages, which can then be used to populate stellar clusters according to their age and masses. The upside of this approach is that stellar evolution would be taken into account, eventually accounting for evolutionary stages outside the main sequence, which are not currently considered. The downside is the computation time.
- 2. Concerning the modeling of the star cluster population, to confirm the robustness of our stellar cluster population simulation method, lower-mass stellar clusters could be generated so that we could have a comparison with cluster surveys obtained from Gaia. In general, the number of locally generated clusters should be consistent with those observed. In addition, comparison with the cluster population of other Milky Way like galaxies can be performed.
- 3. In terms of modeling particle acceleration, several additional aspects could be considered. First, we could account for different injection spectral indexes after estimating for each cluster the termination shock Mach number. At lower energies, the contribution of second-order Fermi acceleration may prove to be non-negligible and should accordingly be considered. We do expect second order Fermi acceleration to affect particles up to several GeV if the magnetic turbulence is strongly enhanced.
- 4. For the calculation of the  $\gamma$ -ray emission, the estimation of the average target density remains a largely unbound parameter. For this purpose, a comparison with results obtained by considering a density dependent on the characteristics of the wind-blown bubble would be interesting. In addition, the inclusion of leptons and the subsequent calculation of their contribution to the diffuse  $\gamma$ -ray emission could prove to be and interesting exercise, possible leading to increasing constrains on the electron to proton fraction.
- 5. An additional check that should be made is with the observed diffuse neutrino flux.

6. Finally, a relevant aspect which we want to investigate is the estimate of the number of detectable sources for the new generation of  $\gamma$ -ray telescopes, such as the Cherenkov Telescope Array, the ASTRI Mini Array, and the Southern Wide-field Gamma-ray Observatory. This specific task is non-trivial, and it will be covered in a future work. The main difficulty relies on the fact that stellar clusters are extended sources. This can be readily verified by looking at Fig. 3.23, where the projected sizes of the cluster bubbles are compared to the point spread functions of the future  $\gamma$ -ray telescopes. It should be noted that a considerable part of the YMSCs is characterized by an extension of 2–3 times the telescope point spread functions, and the analysis of extended sources, especially when located in crowded regions of the sky, is a particularly difficult and challenging task. In fact, we will need instrument response functions (IRFs) for extended sources, which at the moment are not available.



Figure 3.23: Projected bubble radii compared to the point spread function of new generation  $\gamma$ -ray telescopes: Cherenkov Telescope Array (dashed line), Astri Mini Array (solid line), and the Southern Wide-field Gamma-ray Observatory (dot-dashed line). Filled and step histograms report the distribution of the projected radii for constant and nonconstant maximum stellar mass respectively.

## CHAPTER 4\_\_\_\_\_

## IONIZATION OF MOLECULAR CLOUDS CLOSE TO YMSCS

LONG with high-energy particles, YMSCs also produce a sub-GeV population of CRs. The existence of these low energy particles is of fundamental interest for two reasons. From an observational point of view, the ability of low-energy particles to penetrate deep within the dense core of molecular clumps, inducing ionization of the dense material and triggering the generation of specific complex molecular compounds, makes them the perfect tool to probe the presence of freshly accelerated particles, in parallel with  $\gamma$ -ray observations. From the point of view of fundamental physics process, the induced ionization of low-energy CRs is a possible feedback mechanism that regulates the collapse of molecular clumps, making star formation less efficient. In fact, the molecular cloud interior can only be ionized by CRs, as UV and X-rays photons are efficiently absorbed at the cloud periphery (Phan et al., 2020). Moreover, it has been shown that ionization is mainly produced by CRs with energy 1 MeV < E  $\leq$  100MeV (Padovani et al., 2009). The larger the ionization degree, the stronger the coupling of the plasma with the magnetic field, whose pressure works against the gravitational collapse.

Very young star clusters are often found to be close to or surrounded by the fragmented material of the parental GMC. As the cluster develops a wind-blown bubble, it is natural to expect some gas clumps to be encompassed within the expanding hot gas bubble. For these clumps, the ionization rate can be significantly different from the one induced by the low-energy population of galactic CRs. This is because at these energies, galactic CRs are swept away from the expanding motion of the hot bubble, and the continuos advection prevents any attempts of penetration. This effect is indeed similar to the shielding effect of solar wind on galactic CRs for energies  $\leq 1$  GeV. Ionization is hence only provided by the population of freshly accelerated particles from the YMSC, whose spectrum can differ from the galactic standard one. In this chapter, we present the estimation of the ionization rate for molecular clumps found within a YMSC wind-blown bubble. Having in mind the result obtained in Ch. 3 we compute the ionization rate for an average Galactic YMSC. Afterward, we apply our calculation to the specific case of DR21, a molecular cloud observed in the close proximity of Cygnus OB2. For this case, the employed CR spectra used for the calculation of the ionization rate will be the ones found in Ch. 2 that best reproduce the observed  $\gamma$ -ray emission. We will, hence, again consider the two extreme propagation cases, and we will discuss whether the measurement of the ionization rate can be used to constrain the propagation models, and under what circumstances.

The chapter is structured as follows: first, we describe the method employed to estimate the CR spectrum inside a molecular cloud accounting for the propagation of lowenergy particles. Afterward, we apply this method to a synthetic YMSC considering the mean properties obtained in Ch. 3. Finally, we calculate the ionization for parameters representative of DR21, and we compare the obtained result with archival data related to  $HCO^+$  observations.

#### 4.1 Penetration of low-energy CRs in a molecular cloud

As CRs penetrate within a molecular cloud (MC), their spectrum is modified by the combination of propagation and energy loss effects. To obtain the final particle spectrum in a MC, we use the simple analytic prescription proposed by Morlino & Gabici (2015). Let us, hence, consider a MC encompassed within a wind-blown bubble produced by a YMSC. The cloud is characterized by a size  $L_c$  and a density  $n_c$ . Let us furthermore assume that the MC is completely permeated by the bubble magnetic field  $B_2$ , which we consider as spatially constant and aligned along the x-axis, such that the cloud can be schematized as one-dimensional (see Fig. 4.1). The MC density is higher with respect to the bubble density  $n_2$ . We then assume that the transition between the dense cloud environment and the hot, fully ionized, bubble material happens within a thin layer of thickness  $x_c$ . According to this picture, the system can be divided into three main regions:

(A) A zone far away from the MC that we identify as the whole hot bubble downstream of the cluster wind TS. Here, the CR distribution ( $f_2$ ) is provided by Eq. 1.70b and is unaffected by the presence of the cloud<sup>1</sup>. In this one-dimensional problem, the downstream region is set at x < 0 and  $x > L_c + x_c$ .

(B) A transition zone positioned at the cloud borders between  $0 < x < x_c$  and  $L_c + x_c < x < L_c + 2x_c$  where the distribution of CRs is affected by the presence of the MC.

<sup>&</sup>lt;sup>1</sup>This statement is true unless a large fraction of the bubble volume is filled with MCs.



(C) The cloud itself, defined between  $x_c < x < L_c + x_c$ .

Figure 4.1: Scheme of the simplified one-dimensional model employed to describe the geometry of the cloud.

To obtain the particle spectra within the MC, we assume that CRs propagate along the magnetic field lines crossing the cloud, hence, we neglect perpendicular diffusion, assuming that is far slower than parallel (Morlino & Gabici, 2015). As explained in § A.2, low-energy particles in a dense environment undergo severe energy losses due to ionization. Consequently, particles loose energy and migrate to lower energy, such that at a given momentum p, one has  $f_2(p) > f_c(p)$ , where we have defined  $f_c(p)$  as the average CR spectrum within the cloud. Eventually, a negative spatial gradient will appear in the region (B), which in turn triggers the onset of streaming instabilities. The grown of streaming instabilities generates Alfvén waves with speed  $v_A = B_2/\sqrt{4\pi m_p n_2}$ , that propagate towards the cloud direction. This will cause CRs escaping from the MC to be advected back into the cloud from region (B).

The particle distribution in region (B) can be obtained by solving the transport equation (see Eq. 1.41). In the specific case of a steady state system, under the assumption of a spatially constant diffusion coefficient, the transport equation in (B) reads:

$$v_A \frac{\partial f}{\partial x} = D_B \frac{\partial^2 f}{\partial^2 x} \tag{4.1}$$

where  $D_B$  is the diffusion coefficient in region (B). Note that we are not considering energy losses. This is reasonable as the bubble density is expected to be relatively low.

If propagation in this region is mediated by the scattering with Alfvén waves, then simple dimensional analysis implies  $D_B \sim v_A x_c$ . As a consequence, for  $x < x_c$  one has  $x < D_B/v_A$ , so that in Eq. 4.1 the diffusion term dominates. Under these circumstances, the solution is:

$$f = -[f_2 - f(x_c^-)]\frac{x}{x_c} + f_2$$
(4.2)

where  $f(x_c^-)$  is the value of the distribution calculated immediately outside of the MC. The former approximate solution can be then used to calculate the flux of CRs penetrating in the MC. This can be done by integrating Eq 4.1 between  $-\infty$  and  $r_c$ :

$$-2D_B \frac{\partial f}{\partial x}\Big|_{x_c^-} + 2f(x_c^-)v_A = 2f_2 v_A , \qquad (4.3)$$

where the factor 2 accounts for both sides of the cloud. As emphasized by Morlino & Gabici (2015), the validity of the previous equation does not require the presence of the streaming instability, but rather applies to the more general cases of Alfvén waves propagating towards the cloud. This is a common situation, as Alfvén waves cannot come from the cloud itself due to the damping induced by the dense and largely neutral environment. Note that the absence of Alfvén waves within the cloud implies that inside it particles move balistically along the magnetic field lines, so that the CR distribution can be considered approximately constant in space.

Knowing the incoming particle flux, we can readily estimate the spectrum of CRs within the cloud by imposing an equilibrium between the incoming flux and the particle removal rate due to energy losses:

$$2f_2(p)v_A = \frac{L_c}{p^2} \frac{\partial}{\partial p} [\dot{p}p^2 f_c(p)]$$
(4.4)

Noticeably, if energy losses are negligible, Eq. 4.4 leads to the trivial solution  $f_c(p) = f_0(p)$ . Using once again dimensional analysis, the condition for energy losses to be neglected is:

$$u(p) \equiv \frac{v_A \tau_{loss}(p)}{L_c/2} \le 1 \tag{4.5}$$

where  $\tau_{loss}(p) = -p/\dot{p}$  is the energy loss time scale, which can be approximated as a power law in momentum (Morlino & Gabici, 2015):

$$\tau_{loss}(p) = 1.46 \times 10^5 \left(\frac{p}{0.1m_p c}\right)^{2.58} \left(\frac{n_c}{1 \text{ cm}^{-3}}\right)^{-1} \text{ yr.}$$
(4.6)

The condition reported in Eq. 4.5 can be phenomenologically interpreted as follow: as the particle propagation within the cloud is ballistic, the crossing time for a given CR can be simply defined as  $\tau_{cross} = L_c/v_p$ , where  $v_p$  is the velocity of the particle with momentum p. The condition can be then rewritten as  $\tau_{loss}/\tau_{cross} < v_p/v_A$ , which means that energy losses are only important if the particle crosses several times the molecular cloud (of the order of  $v_p/v_A$ ).

Using Eq. 4.6 in Eq. 4.5, we can estimate the energy above which ionization losses are negligible, which is:

$$E_{br} \simeq 70 \left(\frac{v_A}{100 \,\mathrm{kms}^{-1}}\right)^{-0.78} \left(\frac{N_c}{3 \times 10^{21} \,\mathrm{cm}^{-2}}\right)^{0.78} \,\mathrm{MeV}$$
 (4.7)

where  $N_c$  is the column density of the MC. At energies above  $E_{br}$ , the CR spectrum in the cloud is the same as the one in region (A). On the other hand, below  $E_{br}$ , ionization losses start to be important, and the CR spectrum is modified. In this regime, the spectrum is obtained by integrating Eq. 4.4. The final spectrum of particles within the MC can then be written as:

$$f_{c}(p) = \begin{cases} f_{2}(p) & \text{for } E \ge E_{br} \\ f_{2}(p_{br}) \left(\frac{p}{p_{br}}\right)^{-0.42} \left\{ 1 - \frac{\iota(p)}{s-3} \left(\frac{p}{p_{br}}\right)^{-2.58} \left[ 1 - \left(\frac{p}{p_{br}}\right)^{3-s} \right] \right\} & \text{for } E < E_{br} . \end{cases}$$
(4.8)

where  $f_2(p)$  is given by Eq. 1.70b. We further assume that the particle spectrum in the bubble is well represented by a power-law,  $f_2 \propto p^{-s}$ . The latter assumption is fairly well motivated for the following reason: for the energy ranges we are interested in (E<10 GeV), particle propagation in the bubble is totally advective, hence, the spectrum is the same as the one at the injection site, which is a power-law for  $p \ll p_{max}$ . Note that Eq. 4.8 is a simplified analytical approximation of the true solution, which is also provided by Morlino & Gabici (2015). However, it remains a good approximation for the purpose of this work.

For the sake of completeness, we underline that, in principle, the distribution of particles in the bubble has a radial dependence (see § 2.2). However, we stress again that, since we are interested in the low-energy regime, advection is the primary propagation process, and this produces a constant distribution in radius. Thus, ultimately,  $f_2$  can be considered constant within the bubble to a good approximation.

## 4.2 Ionization of molecular clouds embedded in the windblown bubble of a YMSC

As the wind-blown bubble develops around a newly born YMSC, it may happen that a molecular clump, formed due to the fragmentation of the parental GMC, is embedded within the hot shocked plasma. Let us then consider the case of a compact dense clump, with size  $L_c = 1$  pc and column density of  $N_c = 10^{23}$  cm<sup>-2</sup>, close to an average galactic

YMSC. We assume the cluster younger than  $t_{age} \sim 3$  Myr, so that no supernova should have exploded within it yet. We furthermore fix the wind luminosity and mass loss rate to  $L_w = 3 \times 10^{36}$  erg s<sup>-1</sup> and  $\dot{M} = 10^{-6}$  M<sub> $\odot$ </sub> yr<sup>-1</sup> respectively. These values correspond to the average wind luminosity and mass loss rate of an average Galactic YMSC, as esitmated in § 3.2.2.

We can compute the CR spectrum within the clump using Eq. 4.8. To obtain the particle spectrum in the bubble,  $f_2$ , we assume the scenario of Kraichnan-like propagation, as it represents an intermediate case between the two most extreme regimes of particle propagation due to Kolmogorov or Bohm-like turbulence. Once again we assume that a fraction  $\eta_B = 0.1$  of the wind power is converted into a turbulent magnetic field, which is injected at a characteristic length scale of  $L_{inj} = 0.1R_{TS}$ . Under these conditions, the magnetic field in the bubble is  $B_2 = 12 \ \mu$ G.

Furthermore, we calculate the density in the bubble as:

$$n_2 = \frac{M_{shell} t_{age}}{\frac{4\pi}{3} R_b^3} \approx 0.05 \,\mathrm{cm}^{-3} \tag{4.9}$$

where M is the mass of the shell evaporating in the bubble, which we estimate using Eq. 1.24. Fig. 4.2 shows the CR spectrum in the clump compared with the spectrum of the injected particles and the Galactic CR sea as measured by the Voyager spacecraft (Cummings et al., 2016). It can easily be seen that below  $E_{br}$ , the particle spectrum is significantly different from that of the bubble, which, due to advection, is practically the same as that at the TS. The spectrum also differs from that of the CR sea. Therefore, as anticipated, the ionization rate will also be different.

For simplicity, to estimate the ionization rate  $\zeta_{H_2}$ , we only consider the effect of ionization induced by primary protons. This can be calculated using Eq. 13, that we report here for convenience:

$$\zeta_{H2} = \sum_{k} \int_{I(H_2)}^{E_{max}} cf_k(E_k) [1 + \phi_k(E_k)] \sigma_k^{\text{ion}}(E_k) dE_k + \int_0^{E_{max}} cf_p(E_p) \sigma_p^{\text{e.c.}}(E_p) dE_p .$$
(4.10)

We ignore the effect of primary electrons (Eq. 15), as the content of accelerated electron inside stellar clusters has not been established yet. The obtained ionization rate is  $\zeta_{H_2} \approx 2.2 \times 10^{-18} \text{ s}^{-1}$ , which is a factor  $\sim 5$  lower than the Spitzer value  $\zeta_{Spitzer} \approx 10^{-17} \text{ s}^{-1}$  (Spitzer & Tomasko, 1968).

A crucial aspect to emphasize is that the obtained ionization rate is strongly affected by the position of the break, namely by the energy at which ionization losses start to dominate. As described in § 4.1, this depends on the number of times a particle crosses the molecular clump under consideration, which is related in turn to the value of the



Figure 4.2: Comparison between the CR spectrum in a molecular clump embedded within the wind-blown bubble (dashed line) and the particle spectrum in the bubble (dot-dashed line). The spectrum of the Galactic CR sea obtained from Voyager measurement is also reported (solid line). The vertical dotted line denotes the value of  $E_{br}$  at which ionization-induced energy losses in the clump start to dominate.

Alfvén velocity, which depends on the bubble magnetic field and density, both of which can easily be different from the expected values. When estimating the bubble density we assumed that the material within it is composed of the evaporated cold shell medium<sup>2</sup>. However, shell fragmentation and turbulent mixing with the external interstellar medium can increase this value to a higher density. Concerning the magnetic field in the bubble, we are assuming conversion of the wind power into a turbulent magnetic field everywhere in the bubble. This is an approximate assumption, and the magnetic field fluctuation could have an additional spatial dependence. Following these concerns, we show in Fig. 4.3 how the ionization rate changes as a function of the magnetic field and bubble density assuming that they are free variables. Noticeably, the value is always lower than  $\zeta_{Sptizer}$  except for very low density and high magnetic fields.

For the sake of completeness, we recall that the Spitzer value should be considered only as a reference value, as it is calculated without accounting for the physics of CR propagation within the cloud. Phan et al. (2018) provide a better estimation using a refined version of the approach proposed by Morlino & Gabici (2015). They found a ion-

<sup>&</sup>lt;sup>2</sup>The contribution of the wind material is negligible.

ization rate that is lower than  $\zeta_{Spitzer}$  (see solid black line in Fig. 4.3), and closer to, but still higher than, our results, where the density of the bubble and the magnetic field are fixed to  $n_2 = 0.05 \text{ cm}^{-3}$  and  $B_2 = 12 \,\mu\text{G}$ , implying  $\zeta_{H_2} \approx 9.2 \times 10^{-18}$ .



Figure 4.3: Ionization rate induced by primary protons as a function of bubble density and magnetic field for a clump with  $N_c = 10^{23}$  cm<sup>-2</sup>. The dashed line indicates the standard Spitzer value, while the cross marks our results obtained for values of  $B_2$  and  $n_2$  consistent with the YMSC properties (see text). The solid line indicates the ioniazation rate as estimated by Phan et al. (2018).

Finally, we additionally calculate the ionization rates in the case of a molecular clump with  $N_c = 10^{22}$  cm<sup>-2</sup>. In this case, the ionization rate estimated by Phan et al. (2018) is much closer to the standard Spitzer value, and our result for fixed  $n_2$  and  $B_2$  lies between the two numbers (see Fig. 4.4).



Figure 4.4: Same as Fig. 4.3, but for a MC with column density  $N_c = 10^{22} \text{ cm}^{-3}$ .

### 4.3 Comparison with reality: The case of DR21

Because Cygnus OB2 (Cyg OB2) lies within the Cygnus-X star formation complex, it represents the perfect test bed for studying the ionization rate in molecular clouds near a YMSC. Indeed, there are plenty of MCs that in projection are located close to the cluster, although the uncertainties in the position along the line of sight prevent a conclusive association. Nevertheless, a relevant fraction of these clouds is likely in close proximity of the cluster, as they show to be bright at 8  $\mu$ m.



119

Radiation at 8  $\mu$ m should trace the emission from hot dust in photon dominated regions (Röllig et al., 2011b), heated by the ionizing light of massive stars, suggesting then the vicinity to an ionizing source (see Fig. 4.5). In addition, some clouds, such as DR18 and DR20NW, are characterized by an elongated shape pointing toward the direction of Cyg OB2 (Schneider et al., 2006), suggesting a direct interaction with the wind. More precisely, the vicinity of the DR20 complex is also confirmed by precise distance measurement with maser parallaxes (Rygl et al., 2012): the distance is estimated to be 1.46 kpc, very close to the estimated distance of Cyg OB2 (see § 2.1). In addition to DR20, again through maser distance measurements, two massive MCs are observed in the vicinity of Cyg OB2: DR21 and W75N.

The MC DR21, positioned at 1.5 kpc, is of great interest for the scope of this work as the ionization rate has been measured by Hezareh et al. (2008) based on HCO<sup>+</sup> observations (see § A.2). Noticeably, the estimated ionization rate,  $\zeta_{H_2}^{DR21} \approx 3.1 \times 10^{18} \text{ s}^{-1}$ , is found to be lower than the one expected for clouds with the same column density (Padovani et al., 2009). This fact makes the case of DR21 even more interesting, given the result obtained in § 4.2 where a reduced ionization rate was shown for molecular clumps in interaction with a cluster wind-blown bubble.

Having in mind the results obtained in Ch. 2, we can estimate the ionization rate in DR21 by using the particle spectra that best reproduce the observed  $\gamma$ -ray emission, considering both the cases of Kraichnan and Bohm-like propagation. Following the same approach used in § 4.2 for the case of an average Galactic YMSC, we readily obtain the ionization rates of  $\zeta_{H_2} \approx 1.88 \times 10^{-18} \text{ s}^{-1}$  and  $\zeta_{H_2} \approx 1.08 \times 10^{-18} \text{ s}^{-1}$  for the Kraichnan and Bohm cases respectively. Noticeably, both values are in fairly good agreement with the measured ionization rate of DR21.

Note that the particle spectra we are using are calibrated on  $\gamma$ -ray emission under the assumption of a constant particle density in the bubble equal to  $n_2 \approx 8 \text{ cm}^{-3}$ . As we have already pointed out in § 2.5.1, this density is perhaps too high (which we recall, causes a low acceleration efficiency). As mentioned in the previous section, the density in the bubble for estimating the ionization rate turns out to be a crucial parameter, so an erroneous estimate of the density is likely to produce a wrong result. In terms of the  $\gamma$ ray spectrum, the efficiency and the density in bubble are totally degenerate parameters. This means that the same  $\gamma$ -ray spectrum can be re-obtained by keeping constant the product  $n_2 \epsilon_{CR}$  while varying both. Following this reasoning, we additionally consider the case where the CR efficiency is fixed to  $\epsilon_{CR} = 0.1$ , and the bubble density is changed accordingly. All other parameters are fixed to their previous values. The resulting ionization rates are  $\zeta_{H_2} \approx 5.98 \times 10^{-17} \text{ s}^{-1}$  and  $\zeta_{H_2} \approx 6.27 \times 10^{-18} \text{ s}^{-1}$  for the Kraichnan and Bohm cases respectively.

Fig. 4.6 summarized the results obtained so far. Hezareh et al. (2008) state that their



Figure 4.6: Ionization rates estimated for DR21. Cross and X points report the ionization rates for the Kraichnan case for fixed  $n_2$  and  $\epsilon_{CR}$  respectively. The same holds for the Bohm scenario, denoted by square and diamond markers respectively. The solid line indicates the ionization rate measured by Hezareh et al. (2008). Dashed and dotted lines report the uncertainty bars for a factor of 2 and 5 respectively.

measured ionization rate is correct within a factor "of a few", we hence report in Fig. 4.6 also their value with uncertainty bars of factors 2 and 5. Interestingly, all values are roughly in agreement with the exception of the Kraichnan case with  $\epsilon_{CR} = 0.1$ , which returns a ionization rate of more than one order of magnitude higher than the measured one. This is a remarkable result, showing that the measurement of the ionization rate of closeby MCs, combined with  $\gamma$ -ray observations, can be used to constrain (and in some cases even exclude) the models of CR diffusion in these sources.

#### 4.4 Preliminary conclusions and future prospectives

The study of the ionization rate in clouds close to a YMSC is a relevant aspect to understand the physics of CRs in these sources. First, the measurement of the ionization rate, coupled with  $\gamma$ -ray observations, provides a unique method to constrain particle propagation in the vicinity of a stellar cluster, returning in additions valuable information on the CR spectrum at low energy. Secondly, understanding how YMSCs can affect the population of CRs in their neighborhood is a crucial aspect to better comprehend the process of star formation. Eventually, the enhancement (the suppression) of the content of CRs within the wind-blown bubble can induce negative (positive) feedback for star formation.

The work presented in this chapter is split into two distinct parts. In the first half of the chapter, we estimated the ionization rate for a molecular cloud in close proximity of a typical Galactic YMSC. We fixed the wind luminosity and mass loss rate of the stellar cluster to the average values obtained in § 3.2.2 during the study of the Galactic population of stellar clusters. We considered two possible column densities for the molecular cloud:  $N_c = 10^{23}$  cm<sup>-2</sup> and  $N_c = 10^{22}$  cm<sup>-2</sup>, and we found a ionization rate always lower than the Spitzer value. We additionally compared our ionization rate with the more accurate estimates given by Phan et al. (2018), who calculated the ionization rate induced by the population of Galactic CRs using a robust modelization of particle propagation within the cloud. We found our ionization rates in good agreement with the value predicted by Phan et al. (2018), for both considered column densities. Finally, we provided predictions for the ionization rate considering both the magnetic field and particle density in the bubble as free variables ranging between 1–50  $\mu$ G and 10<sup>-3</sup>–1 cm<sup>-3</sup> respectively.

In the second part of the chapter, we focused on the specific case of the molecular cloud DR21, which is located close to the YMSC Cygnus OB2. To infer the ionization rate of DR21, we considered the particle distributions obtained in Ch. 2 that best fit the observed  $\gamma$ -ray emission. More precisely, we used the particle distributions obtained assuming Kraichnan and Bohm-like diffusion. We then compared the resulting ionization rate with archival measurements obtained through observations of HCO<sup>+</sup> and found our result consistent with observations. However, we noticed that our estimates are probably biased.

In fact, the employed particle distributions are obtained after fitting the  $\gamma$ -ray observations with the underlying assumption that the density in the bubble is 8 cm<sup>-3</sup>, which is the average density in a  $\pm 400$  pc region around Cygnus OB2. This number is too high and difficult to justify unless invoking extreme scenarios of material mixing and intense mass evaporation rate from the cold shell. We then recomputed the ionization rate using a lower particle density in the bubble, which we fix to  $n_2 = 0.58$  cm<sup>-3</sup> and  $n_2 = 1.77$  cm<sup>-3</sup> for the Kraichnan and Bohm cases respectively. The particle distributions are accordingly changed by setting the efficiency for CR acceleration to  $\epsilon_{CR} = 0.1$  such that the  $\gamma$ -ray spectrum remains unchanged and so does the consistency with observations. The resulting new ionization rates are in agreement with observations only when considering a Bohm-like diffusion. This would tents to favor a scenario where particle diffusion is provided by a diffusion coefficient with sharper energy dependence than for the Kraichnan case.

The entire work carried out in this chapter is still in its early stage, and there is room

for significant improvements. At present, there are two main limitations to the presented work. The first is the absence of the contribution to the ionization from primary electrons. Currently, the electron distribution within the bubble remains unconstrained. Consequently, no reliable guesses can be made for their contribution. The second is the uncertainty in the diffusion coefficient in the bubble. This prevents us from making a robust modelization and systematic estimation of the ionization rate in clouds close to YMSCs. Therefore, it is currently not possible to make a conclusive statement regarding this particular feedback on star formation by YMSCs.

To overcome these limitations, systematic radio and X-ray observations of YMSCs could be used to constrain the magnetic field intensity within the wind-blown bubbles. This would lead to better estimates of the ionization rates, in parallel to a better modelization of the particle diffusion in these systems. X-ray and radio observations could also provide precious constrains on the electrons distribution. We aim to investigate these aspects with the upcoming data from the MeerKAT radio telescope and the Galactic Plane survey carried out by eROSITA.

# CHAPTER 5\_

### CONCLUSIONS

ound massive stellar cluster (YMSCs) represent at present one of the most intriguing sources in the panorama of high-energy astrophysics. The environment shaped by the winds of multiple young massive stars provides ideal conditions for particle acceleration, making these objects potential Galactic CR accelerators. In this manuscript, we analyzed the capability of YMSCs to produce CRs assuming the scenario where particle acceleration exclusively take place at the termination shock of the cluster wind. We performed the investigation by means of a comprehensive modelization of the hadronic  $\gamma$ -ray emission and by estimating the ionization rate in nearby molecular clouds induced by the injected low-energy particles. To model the CR distribution in the scenario we considered, we employed the model developed by Morlino et al. (2021), which describes both particle acceleration at the wind termination shock and particle propagation within the wind-blown bubble of the stellar cluster.

In the first part of the work, we focused on the specific case of the YMSC Cygnus OB2. We tried to interpret the diffuse  $\gamma$ -ray emission detected by the Fermi-LAT, HAWC, and ARGO experiments in terms of hadronic emission generated by a population of freshly accelerated particles, whose distribution is obtained following the model of Morlino et al. (2021). When modeling the particle distribution within the wind-blown bubble, we considered three possible scenarios corresponding to three different diffusion regimes: Kolmogorov, Kraichnan and Bohm like. We found that a Kolmogorov like diffusion in the system cannot reproduce the observed  $\gamma$ -ray spectrum, as the wind luminosity required to fit the  $\gamma$ -ray flux at very-high energy is more than one order of magnitude higher than the one inferred from the observed population of massive stars ( $L_w \simeq 1.5 - 3 \times 10^{38}$  erg s<sup>-1</sup>). Similarly, also Kraichnan like diffusion requires a wind luminosity higher than the one inferred from the stellar population. However, differently from the Kolmogorov

case, the discrepancy is reduced and amount to a factor of  $\sim 4$ . When considering Bohm like diffusion, we found that the  $\gamma$ -ray spectrum can be explained with a wind luminosity consistent with our estimations, if  $\sim 30\%$  of the wind power is converted into turbulent magnetic field. As the wind luminosity required to reproduce the  $\gamma$ -ray emission decreases when considering increasingly harder energy dependencies for the diffusion coefficient (at the cost of increasing by a factor of a few the fraction of wind power in magnetic turbulence), we concluded that, if acceleration proceeds according to our model, the most plausible scenario is the one where the turbulence spectrum has a spectral index that lies between the Kraichnan and Bohm cases.

In addition to the  $\gamma$ -ray spectrum, the observations of Fermi-LAT and HAWC provide also the radial profile of the emission in a circular region of 2.2° (corresponding to  $\sim 54$ pc for an assumed distance of 1.4 kpc) centered on Cygnus OB2. We then compared the expected radial profile for our best-fit cases to the observed morphology. We limited this analysis only to the Kraichnan and Bohm scenarios. We found that both cases return a flat morphological profile in the same region. This profile is consistent with HAWC observations, but not with the Fermi-LAT ones, which are characterized by a centrally peaked morphology. From the point of view of our model, a flat morphology is the result of advection as the dominant propagation mechanism. Diffusive transport in a region of 2.2° (corresponding to a physical radius of  $\sim 54$  pc) is expected to start dominating only at energies above  $\sim 100$  TeV. The reason for the peaked morphology observed by Fermi-LAT remains not fully understood. One possibility is that the increasing  $\gamma$ -ray luminosity towards the center of the system is due to inverse Compton emission by a population of accelerated leptons confined in a thin shell around the termination shock. This scenario will be investigated in a forthcoming paper, where a robust modelization of electrons acceleration and diffusion in stellar clusters, along with an estimate of their  $\gamma$ -ray emission, will be provided.

Given the obtained results, we conclude that the model of particle acceleration at the wind termination shock of Cygnus OB2 is capable to reproduce the observed  $\gamma$ -ray emission, and that particle diffusion in the system is likely the result of a scenario in which the power spectrum of the magnetic turbulence has a spectral index in between Kraichnan and Bohm predictions. In the near future, it will be possible to confirm or reject these conclusions through a number of different tests, among which is the analysis of the  $\gamma$ -ray spectrum at different distances from Cygnus OB2, and the study of the  $\gamma$ -ray emission from nearby molecular clouds. In this regard, joint observations with the MAGIC and LST-1 telescopes are currently ongoing for the molecular clouds DR21 and W75N.

In the second part of the manuscript, we analyzed the  $\gamma$ -ray emission expected from

a simulated population of Galactic YMSCs. Given that, within the model of particle acceleration and propagation in YMSCs we assume, the  $\gamma$ -ray radiation is expected to be mostly produced by particles confined within the wind-blown bubble, the emission of single clusters is expected to be on average extended ( $\geq 0.1^{\circ}$ ) and possibly difficult to distinguish from the diffuse. Therefore, we focused on the diffuse emission produced by the collective contribution of multiple sources, rather than examining the properties of each source individually.

To do so, we simulated a population of Galactic YMSCs following a cluster distribution function tuned on several observational parameters, such as mass and age distribution of local clusters, observed star forming rate in clusters, and the Galactic radial distribution of giant molecular clouds. To estimate the cluster wind properties, we simulated for each cluster a synthetic population of stars. When simulating the cluster stellar population, we considered two potential scenarios. In the first one, we assumed that the maximum mass of stars generated by each cluster is fixed as 150  $M_{\odot}$ . In the second scenario, we assumed that the maximum stellar mass is a function of the cluster mass. After generating the star population, for each star we modeled the associated wind using again empirical relations based on the stellar parameters, such as bolometric luminosity, radius and temperature. We estimated these parameters starting from the stellar mass and using empirical relations. In addition to our main goal of estimating the diffuse  $\gamma$ -ray emission, the simulation of a synthetic Galactic population of YMSCs provides already per se an interesting results. We found that the average wind mass loss rate and wind power of stellar clusters are  $\overline{\dot{M}} \approx 10^{-6} \,\mathrm{M_{\odot} \, yr^{-1}}$  and  $\overline{L_w} \approx 3 \times 10^{36} \,\mathrm{erg \, s^{-1}}$  respectively. The average size of the wind-blown bubble is found to be  $\overline{R}_b \approx 60$  pc, so that the mean projected size of the  $\gamma$ -ray emission is  $\approx 0.25^{\circ}$ . These average values do not vary much when we considered different maximum stellar masses.

When calculating the CR distribution in each cluster, we considered two different diffusion regimes, Kraichnan and Bohm like. We further consider subcases for the Kraichnan regime, assuming two possible injection scales of the magnetic turbulence, 1 pc and 10% of the termination shock radius. In all scenarios, and independently of the considered maximum stellar mass, we found no cluster able to produce maximum particle energies above 1 PeV. The fraction of YMSCs accelerating particles above 100 TeV is found to be 7 - 13%, while above 500 TeV it reduces to  $\sim 0.5 - 1\%$ , which corresponds to 3 - 6clusters. With such a small number of clusters statistical fluctuations are not negligible, and multiple realizations of the galactic population of YMSCs are required to obtain a robust estimate of the expected number of PeVatron clusters.

After modeling the CR distribution in all YMSCs, we calculated the diffuse  $\gamma$ -ray emission in two specific regions of the Galactic plane, the first one defined by  $15^{\circ} < l < 5^{\circ}$  and  $|b| < 5^{\circ}$  (named ROI1) and the second one by  $100^{\circ} < l < 25^{\circ}$  and  $|b| < 5^{\circ}$  (named

R012). The diffuse  $\gamma$ -ray spectrum in R011 is available from Fermi-LAT observations, while in R012, data from EGRET, Argo and Tibet-AS $\gamma$  are considered. We found that in R011 the computed  $\gamma$ -ray spectrum is below the data by a factor of  $\sim 2-3$  for the Bohm scenario and  $\sim 10$  for the Kraichnan case at  $E_{\gamma} > 10$  GeV. In R012, we obtain a diffuse emission that overshoots the Argo observations by a factor  $\sim 3$  between 0.2 TeV  $\lesssim E_{\gamma} \lesssim 1$  TeV and  $\sim 6$  at  $E_{\gamma} \approx 1-2$  TeV when considering the Bohm scenario. The spectrum in the Kraichnan case is instead below the data by a factor of  $\sim 2$  at 0.2 TeV  $\lesssim E_{\gamma} \lesssim 1$  TeV, while accounting for most of the observed flux at  $E_{\gamma} \approx 1-2$  TeV. With these numbers, we conclude that the observed diffuse emission at  $\sim$  TeV energies could be largely provided by YMSCs. We also note that full consistency with observations in the Bohm scenario can be obtained by reducing the average target density (assumed to be 10 cm<sup>-3</sup>) or the efficiency in CR acceleration (assumed to be 10%).

In general, the obtained  $\gamma$ -ray spectra do not vary much when considering different maximum stellar masses in clusters, and different turbulence injection scales for the Kraichnan case. However, the spectra obtained when considering a Kraichnan and Bohm-like diffusion are significantly different. Using a Kraichnan diffusion coefficient returns a diffuse  $\gamma$ -ray emission characterized by a cut-off at energies of  $\sim 100$  GeV, while the emission in the Bohm scenario starts to fade at energies above 1 TeV. At lower energies the spectrum for the Bohm case is found to be harder than for Kraichnan. This is a direct consequence of the interplay between advection and diffusion, whose relative importance differs in the two cases under analysis. In the Kraichnan case, diffusion starts to dominate at energies lower than in the Bohm case. This produces a decrease of the number of emitting particles in the wind blown bubble which causes a softening of the  $\gamma$ -ray spectrum. Lastly, we explored how the diffuse  $\gamma$ -ray emission varies with the cluster mass. Our analysis revealed that the less numerous and most massive clusters are the primary contributors to the diffuse emission.

Several additions and improvements to the current status of the work are possible and foreseen. At present, the mains missing ingredients in the second part of the presented work are the contribution to particle acceleration by supernova explosions and the contribution to  $\gamma$ -ray emission from accelerated leptons. Both aspect are still poorly understood and lack a firmly established theoretical framework. Regarding the simulation of the galactic population of YMSCs, our future plans include conducting a comprehensive consistency check of the generated population by comparing it to the observed population of clusters in Milky Way-like galaxies. Furthermore, multiple realizations of the Galactic population will have to be considered.

In the third and last part of the manuscript, we moved our analysis to lower energies, and we investigated the ionization rate induced by sub-GeV particles in molecular clouds close to YMSCs. Young clusters are expected to be surrounded by dense molecular clumps, which can be either remains of the cluster parent molecular cloud or generated by the fragmentation of the swept-up shell created by the expansion of the wind bubble. We considered a situation in which one of these clumps finds itself embedded within the wind blown bubble. We modeled the propagation of low energy CRs following the approach of Morlino & Gabici (2015), where particles in the cloud move balistically along the magnetic field while experiencing energy losses due to direct ionization. Following this method, the particle spectrum in the cloud is equal to the one in the wind blown bubble for energies larger than a certain value  $E_{br}$ , which describes the energy below which losses start to be relevant. Energy losses are relevant when particles cross several times the molecular cloud, and this condition is related to the Alfvèn velocity in the bubble, hence, it depends on the value of plasma density and magnetic field strength in the bubble. For  $E < E_{br}$ , the spectrum is harder as it is modified by energy losses.

As a first trial, we considered a molecular clump of size  $L_c = 1$  pc with two possible values of the column density  $N_c = 10^{22}$  cm<sup>-2</sup> and  $N_c = 10^{23}$  cm<sup>-2</sup>. We assumed the case in which the clump is close to a YMSC of age less than 3 Myr whose wind luminosity and mass loss rate are equal to the average values estimated from our analysis of the Galactic population of clusters, i.e.  $\overline{L_w} \approx 3 \times 10^{36}$  erg s<sup>-1</sup> and  $\overline{\dot{M}} \approx 10^{-6}$  M<sub> $\odot$ </sub> yr<sup>-1</sup>. When calculating the CR distribution in the wind blown bubble, we considered the scenario where particle propagation is governed by Kraichnan like diffusion. Assuming for the magnetic field in the bubble a value such that the magnetic turbulence power is 10% of the wind luminosity, and considering a density in the bubble consistent with the mass evaporated from the swept up shell, we found a ionization rate of  $\zeta_{H_2} \approx 2.2 \times 10^{-18}$  s<sup>-1</sup> for  $N_c = 10^{23}$  cm<sup>-2</sup>. This is lower than the Spitzer value by almost one order of magnitude. Even when compared with more recent estimates, presented in Phan et al. (2018), our ionization rate is still lower by a factor of a few.

When considering a lower column density of the cloud, we found a ionization rate of  $\zeta_{H_2} \approx 9.2 \times 10^{-18}$ , which is consistent within a factor of a few with both the Spitzer value and the estimates given by Phan et al. (2018). We noted that the ionization rate obtained can substantially differ by varying the energy at which the losses become important. As this value depends on the magnetic field and density inside the bubble, which are highly uncertain parameters, we also made predictions for the ionization rate by taking into account a broad range of magnetic field and density values. In general, low (high) magnetic field and high (low) density produced lower (higher) ionization rates than the Spitzer value.

After studying the case of a generic cluster, we focused on the specific case of Cygnus OB2 and estimated the expected ionization rate for the nearby molecular cloud DR21. To this end, we used the CR distributions in Cygnus OB2 calculated for a Kraichnan and

Bohm like diffusion that best fit the observed  $\gamma$ -ray emission. We found the ionization rates to be  $\zeta_{H_2, K} \approx 1.88 \times 10^{-18} \text{ s}^{-1}$  and  $\zeta_{H_2, B} \approx 1.08 \times 10^{-18} \text{ s}^{-1}$  for the Kraichnan and Bohm cases respectively. These values are consistent within a factor of a few with the ionization rate estimated by observations of the HCO<sup>+</sup> and equal to  $\zeta_{H_2}^{DR21} \approx 3.1 \times 10^{-18} \text{ s}^{-1}$  (Hezareh et al., 2008).

These ionization rates are obtained considering a density in the bubble of 8 cm<sup>-3</sup>, which is the average density observed near Cygnus OB2 and inferred from HI and CO observations. This value of density is considerably high. Given the significant variation observed in the ionization rate by varying the density within the bubble, we also investigated a scenario where the density is kept constant at  $0.58 \text{ cm}^{-3}$  and  $1.77 \text{ cm}^{-3}$  for the Kraichnan and Bohm scenarios respectively. These values are chosen so that the  $\gamma$ -ray spectrum remains unchanged if the efficiency for CR acceleration is fixed to 10% of the wind luminosity. The chosen densities are in better agreement with the expected value assuming that the material inside the bubble is composed of the mass evaporated from the swept-up shell. We found for this new scenario ionization rates of  $\zeta_{H_2} \approx 5.98 \times 10^{-17} \text{ s}^{-1}$  and  $\zeta_{H_2} \approx 6.27 \times 10^{-18} \text{ s}^{-1}$  for the Kraichnan and Bohm cases respectively. The ionization rate for the Kraichnan case is more than one order of magnitude higher than the value inferred from observations. This seems to favor the Bohm case, or in general, a diffusion coefficient in the system with an energy dependence stronger than the Kraichnan case.

The work presented in this last part is still at an early stage of development. Nevertheless, the results obtained are promising and underline the importance of having a parallel modelization of the  $\gamma$ -ray emission together with the ionization rate induced by low-energy CRs. Indeed, the combination of these two pieces of information can provide a powerful consistency check for any model of CR acceleration and propagation. Furthermore, the capability of understanding the ionization rate of clouds close to YM-SCs is crucial to assess the relative importance of the feedback channels that govern the star formation process.

At present, the main limitation of this work is given by the absence of the contribution to the ionization rate induced by primary leptons. As a future step, we plan to utilize X-ray and radio data from Galactic surveys by eROSITA and MeerKAT to conduct a systematic analysis of the environment surrounding YMSCs. This will enable us to obtain a more reliable estimation of the magnetic field within the wind-blown bubbles of stellar clusters, which will lead to even more solid estimates of the ionization rate.

Now more than ever the modelization of YMSCs as CR accelerators and  $\gamma$ -ray sources is of primary importance. In the coming years, thanks to the new generation of  $\gamma$ -ray observatories, such as CTA, ASTRI Mini Array, and SWGO, the importance of YMSCs as high-energy sources is bound to increase. The CTA and ASTRI observatories will perform deep surveys of the Galactic plane, which, combined with extended surveys in other bands, such as those of MeerKAT and eROSITA, will provide a valuable set of information for the systematic study of these sources. Eventually, multi-band analyses will be the key to study particle acceleration in YMSCs and evaluate their contribution to the generation of Galactic CRs. All these studies will hopefully soon bring us closer to the solution of the century-long enigma of the origin of CRs. In parallel, a campaign of observations and investigation of the ionization rate in molecular clouds in close proximity to YMSCs would provide a significant cross-check for the presence of accelerated particles. These observations could also provide insights into the star formation mechanism and improve our understanding of feedback mechanisms associated with the presence of freshly accelerated CRs in the vicinity of clusters.

### APPENDIX

#### A.1 $\gamma$ -ray emission as a tracer of CRs

The detection of high-energy radiation from a given object in the sky is direct proof of the presence of energetic particles. CRs may produce  $\gamma$ -rays through different possible radiation processes that depend on the type of emitting particles. Energetic hadrons interacting with the ISM matter may undergo the following nuclear interactions:

$$p + p \rightarrow p + p + \pi^{0}$$
$$p + p \rightarrow p + n + \pi^{+}$$
$$p + p \rightarrow p + p + \pi^{+} + \pi^{-}$$

where p are protons and  $\pi$  are pions<sup>1</sup>. The production of neutral pions is the main channel to generate hadronic  $\gamma$ -ray emission, as these subsequent decays according to  $\pi^0 \rightarrow \gamma + \gamma$ . Clearly, the creation of neutral pions is only possible if the energy of a proton is sufficiently high. The proton energy threshold ( $E_{th}$ ) can be obtained through a few straightforward calculations of particle kinematic (Rybicki & Lightman, 1986), and is:

$$E_{th} = 2m_{\pi}c^2 + \frac{m_{\pi}^2}{2m_p}c^2 \approx 280 \text{ MeV}$$
(1)

where  $m_{\pi}$  is the pion mass.

<sup>&</sup>lt;sup>1</sup>It is worth noting that charged pions are also created as a result of nuclear interactions. These mesons can then decay, producing neutrinos that propagate straight to Earth. Neutrino astronomy can therefore be considered as an additional probe of the presence of CRs (Spiering, 2012), with the fundamental and important difference from  $\gamma$ -ray astronomy that neutrinos can only be created via a hadronic interaction channel, so the detection of neutrinos is a smoking gun for the presence of hadronic CRs. Unfortunately, given the low interaction cross-section of neutrinos and the low fluxes expected at high-energy, this branch of astronomy remains affected by low statistics and hence particularly challenging, requiring massive detectors and very long exposure times.

Given a distribution of CRs ( $f_{CR}$ ), it is possible to calculate the expected  $\gamma$ -ray flux ( $\phi_{\gamma}$ ) from  $\pi^0$  production within a certain volume V as:

$$\phi_{\gamma}(E_{\gamma}) = \frac{1}{4\pi d^2} \int \int_V cf_{CR}(E_p, r)n(r) \frac{d\sigma(E_p, E_{\gamma})}{dE_p} dE_p dV,$$
(2)

where *d* is the distance of the considered volume,  $\sigma$  is the cross-section for  $\gamma$ -ray production from  $\pi_0$  decay, *n* is the number density of the target medium, and  $E_p$  with  $E_{\gamma}$  are the CR kinetic energy and the  $\gamma$ -ray energy respectively. Kafexhiu et al. (2014) provides an analytical prescription for the cross-section, based on empirical fits of published results from p-p interactions at particle colliders and outcomes of Montecarlo predictions based on the Standard Model. The full expression results rather cumbersome, but it can be briefly described as the product of two contributing terms:

$$\frac{d\sigma}{dE_{\gamma}}(E_p, E_{\gamma}) = A(E_p) \times F(E_p, E_{\gamma})$$
(3)

where  $A(E_p) = \max(d\sigma_{\pi}/dE_p)$  is the maximum value of the pion production cross section, while  $F(E_p, E_{\gamma})$  is a term describing the spectrum of produced  $\gamma$ -ray as a function of the proton energy. Fig. 1 shows the cross section as a function of  $E_{\gamma}$  for different values of the parent proton energy. Notice that a proton is able to produce a  $\gamma$ -ray with a maximum energy of  $\sim 0.1E_p$ . This is particularly important as the detection of  $\gamma$ -rays with a certain energy  $E'_{\gamma}$  implies the presence of hadrons with energies of at least  $\sim 10E'_{\gamma}$ .

Leptons produce  $\gamma$ -rays through the Inverse Compton (IC) process. This mechanism consists in the scattering of high energy electron with a low energy photon, resulting from one side in an average energy gain for the photon, and an average energy loss for the electron. If in the electron rest frame the photon energy is  $h\nu \ll m_e c^2$ , with  $h, \nu$  and  $m_e$  as the Planck constant, photon frequency and electron mass respectively, then, the process is mediated by Thomson cross section ( $\sigma_T$ ). If, instead,  $h\nu \simeq m_e c^2$  the crosssection decrease and is described by the Klein-Nishina formula (Klein & Nishina, 1929).

In case of an electron colliding with an isotropic monochromatic radiation field with frequency  $\nu_0$ , under the assumption of Thomson regime, the average frequency ( $\nu_c$ ) of the upscattered photons and the radiative power emitted by the electron are (Rybicki & Lightman, 1986):

$$\nu_c = \frac{4}{3}\gamma^2\nu_0\tag{4}$$

and

$$\frac{dE_{IC}}{dt} = \frac{4}{3}\sigma_T \beta^2 \gamma^2 U_{ph}^2 \tag{5}$$

where  $\gamma$  is the electron relativistic Lorentz gamma factor, and  $U_{ph}$  is the radiation field energy density. In general, the outgoing photon will not have a single characteristic fre-



Figure 1: Differential cross section for  $\gamma$ -ray production for different protons energy of 1 GeV, 10 GeV, 100 GeV, 1 TeV, 10 TeV, and 1 PeV. The cross-section is calculated using the recipe of Kafexhiu et al. (2014), considering the specific case where the trend at energies higher than 100 GeV is obtained using the Montecarlo SIBYLL.

quency, but rather a possible spectrum of frequencies, which makes the calculation of the IC spectrum non-trivial. However, one can with good accuracy assume that all photons are upscattered with the specific frequency given by Eq. 4, in which case the IC spectrum from a given volume V can be written as:

$$\phi_{IC}(E_{\gamma}) = \frac{1}{4\pi d^2} \int \int_V \frac{dE_{IC}}{dt} f_e(E_e, r) \delta(\nu - \nu_c) dE_e dV$$
(6)

where  $f_e$  is the distribution of emitting electrons and  $E_e$  is the energy of the electrons. In the case of the Klein-Nishina regime, Eq. 5 cannot be used, and the previous expression is no longer valid. The complete treatment becomes then quite involved, with the full solution in this regime given by Blumenthal & Gould (1970). Yet, in the specific case where the target radiative field is represented by a black body, a simple analytical approximation is still possible in any scattering regime. This solution has been developed by Khangulyan et al. (2014) and consists of approximating the energy losses as:

$$\frac{dE_{IC}}{dt} = q_1(T_{rad})G_{iso}(E_e, T_{rad})$$
(7)

with

$$G_{iso}(E_e, T_{rad}) = \left(\frac{m_e c^2}{k_B T_{rad}}\right)^2 \frac{c_{iso} Y(E_e, T_{rad})}{1 + \frac{12}{\pi^2} c_{iso} Y(E_e, T_{rad})} \ln\left(1 + \frac{\pi^4 Y(E_e, T_{rad})}{135c_{iso}}\right) , \quad (8)$$

$$q_1(T_{rad}) = \frac{4}{3} \sigma_T c U_{ph} \frac{135}{16\pi^4} , \qquad (9)$$

and

$$Y(E_e, T_{rad}) = \frac{4E_e k_B T_{rad}}{(m_e c^2)^2}$$
(10)

where  $T_{rad}$  is the black body temperature,  $k_B$  is the Boltzmann constant, and  $c_{iso} = 4.62$ . The IC flux is then easily obtained by using Eq.6 with the new expression for the emitted power.

#### A.2 Ionization of molecular clouds by low energy CRs

Differently from ionizing radiation, low-energy CRs can penetrate deep within the core of gas clumps, providing ionization of clouds, possibly heating the dense cold gas (Galli & Padovani, 2015), and indirectly inducing chemical reactions in the ISM, generating complex molecular compounds (Dalgarno, 2006). As YMSCs are expected to be surrounded by the dense molecular envelope of the parent GMC, we will focus through this section on the ionization of molecular clouds (MC). The ionization processes in MCs have been exhaustively investigated by Padovani et al. (2009). There is a plethora of interactions that can lead to the ionization of  $H_2^2$ , namely, proton-induced ionization<sup>3</sup>:

$$p_{CR} + H_2 \to p_{CR} + H_2^+ + e$$
 (11a)

$$p_{CR} + H_2 \to H + H_2^+ \tag{11b}$$

$$p_{CR} + H_2 \to p_{CR} + H + H^+ + e$$
 (11c)

$$p_{CR} + H_2 \to p_{CR} + 2H^+ + 2e$$
 (11d)

or ionization induced by CR electrons:

$$e_{CR} + H_2 \to e_{CR} + H_2^+ + e$$
 (12a)

$$e_{CR} + H_2 \to e_{CR} + H + H^+ + e$$
 (12b)

$$e_{CR} + H_2 \to e_{CR} + 2H^+ + 2e$$
. (12c)

<sup>3</sup>Also nuclei may induce ionization but, for the sake of simplicity, as hadron induced process, we will only consider ionization generated by protons.

<sup>&</sup>lt;sup>2</sup>MCs also harbor a fraction of helium, which can be likewise ionized (see Padovani et al. (2009))

The molecular hydrogen ion  $(H_2^+)$  production rate from the sole contribution of protons and electrons through *direct ionization* (Eq. 11a and Eq. 12a) and *electron capture* (Eq. 11b) processes can be calculated as:

$$\zeta_{H2} = \sum_{k} \int_{I(H_2)}^{E_{max}} cf_k(E_k) [1 + \phi_k(E_k)] \sigma_k^{\text{ion}}(E_k) dE_k + \int_0^{E_{max}} cf_p(E_p) \sigma_p^{\text{e.c.}}(E_p) dE_p \quad (13)$$

where the index k account for the considered CR species (electrons or protons),  $I(H_2) = 15.603$  eV is the ionization potential of H<sub>2</sub>, and  $\sigma^{\text{ion}}$  and  $\sigma^{\text{e.c}}$  are the direct ionization and electron capture cross sections respectively (see Padovani et al. (2009) and references therein). The quantity  $\phi_k(E_k)$  is a correction factor accounting for the ionization induced by a population of secondary electrons created by direct ionization, and can be calculated as:

$$\phi_k(E_k) = \frac{1}{\sigma_k^{\text{ion}}(E_k)} \int_{I(H_2)}^{E_{max}} \mathcal{P}(E_k, E'_e) \sigma_e^{\text{ion}}(E'_e) dE'_e \tag{14}$$

with the term  $\mathcal{P}(E_k, E'_e)$  describing the probability that a secondary electron with energy E'<sub>e</sub> is created during a primary ionization by a particle with energy E<sub>k</sub>. To estimate, instead, the electron production rate, one also needs to account for the contribution of *dissociative ionization* (Eq. 11c and Eq. 12b) and *double ionization* (Eq. 11d and Eq. 12c) processes:

$$\zeta_{e} = \sum_{k} \int_{I(H_{2})}^{E_{max}} cf_{k}(E_{k})[1 + \phi_{k}(E_{k})]\sigma_{k}^{\text{ion}}(E_{k})dE_{k} + \sum_{k} \int_{E_{\text{diss.ion.}}}^{E_{max}} cf_{k}(E_{k})[1 + \phi_{k}(E_{k})]\sigma_{k}^{\text{diss.ion.}}(E_{k})dE_{k}$$

$$+ 2\sum_{k} \int_{E_{\text{doub.ion.}}}^{E_{max}} cf_{k}(E_{k})[1 + \phi_{k}(E_{k})]\sigma_{k}^{\text{doub.ion.}}(E_{k})dE_{k}$$
(15)

where  $\sigma^{\text{diss.ion.}}$  and  $\sigma^{\text{doub.ion.}}$  are the cross sections for the dissociative and double ionization process. While we mention them for completeness, generally, these two ionization processes can be safely neglected in comparison to the direct ionization, as the cross section are a factor 10–100 below  $\sigma^{\text{ion.}}$ .

Creation of  $H_2^+$  ions in a dense environment can trigger an intricate chain of chemical gas phase based reactions with the formation of complex molecules, such as for example, DCO<sup>+</sup> and HCO<sup>+</sup> (*D* stands for the deuterium isotope). The detection of these species through molecular radio emission lines can be used to assess the value of  $\zeta_{H_2}$ (Caselli et al., 1998; Vaupré et al., 2014). In a steady-state regime, the abundances of DCO<sup>+</sup> and HCO<sup>+</sup> are set by the following main reactions (Guelin et al., 1977; Caselli et al., 1998):

$$k_{CR} + H_2 \xrightarrow{\zeta_{H2}} k_{CR} + H_2^+ + e \tag{16a}$$

$$H_2^+ + H_2 \xrightarrow{\kappa_{H_2^+}} H_3^+ + H \tag{16b}$$

$$H_3^+ + CO \xrightarrow{\kappa_H} HCO^+ + H_2$$
 (16c)

$$HCO^+ + e \xrightarrow{\beta'} CO + H$$
 (16d)

$$H_3^+ + e \xrightarrow{\beta} 3H \text{ (or } H_2 + H)$$
 (16e)

$$H + H \xrightarrow{\kappa'} H_2$$
 (16f)

$$H_3^+ + HD \underbrace{\stackrel{\kappa_f}{\overleftarrow{\qquad}}}_{\kappa_f^{-1}} H_2 D^+ + H_2 \tag{16g}$$

$$H_2D^+ + CO \xrightarrow{\kappa_D} DCO^+H_2$$
 (16h)

$$DCO^+ + e \xrightarrow{\beta'} CO + D$$
 (16i)

$$H_2D^+ + e \xrightarrow{\kappa_e} 2H + D \text{ (or } H_2 + D \text{ or } HD + H)$$
 (16j)

$$H + D \xrightarrow{\kappa''} HD$$
 (16k)

$$H_2D^+ + CO \xrightarrow{\kappa'_D} HCO^+ + H_2$$
 (161)

$$H_3^+ + D \xleftarrow[\kappa_f'^{-1}]{\kappa_f'^{-1}} H_2 D^+ + H$$
(16m)

$$CO^+ + HD \xrightarrow{\kappa_{CO^+}} DCO^+ + H$$
 (16n)

where the parameters appearing over the arrows denotes the creation (or destruction) rates for each chemical compound (see Vaupré et al. (2014)). The ionization rate  $\zeta_{H2}$  can then be analytically expressed in terms of the abundance ratios  $R_D = [DCO^+]/[HCO^+]$  and  $R_H = [HCO^+]/[CO]$  (Wootten et al., 1979; Guelin et al., 1982):

$$R_D = \frac{[DCO^+]}{[HCO^+]} \simeq \frac{1}{3} \frac{x(H_2D^+)}{x(H_3^+)} \simeq \frac{1}{3} \frac{\kappa_f x(HD)}{\kappa_e + x(e) + \delta + \kappa_f^{-1}/2}$$
(17)

$$R_H = \frac{[HCO^+]}{[CO]} = \frac{\kappa_H x(H_3^+)}{\beta' x(e)} \simeq \frac{\kappa_H}{x(e)[2\beta x(e) + \delta]\beta'} \frac{\zeta_{H2}}{n_H}$$
(18)

where  $\kappa_f$ ,  $\kappa_e$ ,  $\kappa_H$ ,  $\beta$ ,  $\beta'$ , are the reaction rates<sup>4</sup> occurring in the chemical networks of Eq.16,  $\delta \approx \delta_{H_3^+} \approx \delta_{H_2D^+}$  is the total destruction rate of  $H_3^+$  or  $H_2D^+$  due to reactions with neutral species such as CO and O, and  $x(\mathcal{X}) = n_{\mathcal{X}}/n_H$  denotes the fractional abundance of a given specie  $\mathcal{X}$  with number density  $n_{\mathcal{X}}$ . Inversion of Eq. 18 gives the expression for

<sup>&</sup>lt;sup>4</sup>Note that the reaction coefficients actually mediate collisional processes, and are consequently (some of them) dependent on the kinetic temperature of the gas (Caselli et al., 1998).
$\zeta_{H2}$  (Caselli et al., 1998):

$$\zeta_{H2} = [2\beta x(e) + \delta] \frac{R_H x(e)\beta' n_H}{\kappa_H} \simeq \left[ 7.5 \times 10^{-4} x(e) + \frac{4.6 \times 10^{-10}}{f_D} \right] x(e) n_H R_H \quad (19)$$

where  $f_D$  is the depletion factor of C and O, defined such that  $1/f_D$  is the fraction of C and O in the gas phase, and

$$x(e) = \left[\frac{\kappa_f x(HD)}{3R_D} - \delta\right] \simeq \frac{2.7 \times 10^{-8}}{R_D} - \frac{1.2 \times 10^{-6}}{f_D}$$
(20)

Assuming we know the depletion factor,  $\zeta_{H2}$  is easily obtained as  $R_D$  and  $R_H$  can be readily estimated from observation through the ratios of molecular rotational lines.

As a final remark, it is essential to emphasize that the value of  $\zeta_{H2}$  does not return any information on the spectral shape of the CR distribution since the spectral information is lost in the integral in Eq. 13. However, one could use low-energy  $\gamma$ -ray emission to retrieve the CR spectrum, in this way, the value of the ionization rate can be employed as a cross-check to have a comprehensively self-consistent estimate of the CR distribution.

## BIBLIOGRAPHY

- Abbasi R. U., et al., 2008, Phys. Rev. Lett., 100, 101101
- Abbott D. C., 1979, Mass Loss and Evolution of O-Type Stars, 83, 237
- Abdollahi S., et al., 2020, ApJS, 247, 33
- Abeysekara A. U., et al., 2021a, Nature Astronomy, 5, 465
- Abeysekara A. U., et al., 2021b, Nature Astronomy, 5, 465
- Abramowski A., et al., 2012, A&A, 537, A114
- Ackermann M., et al., 2011a, Science, 334, 1103
- Ackermann M., et al., 2011b, Science, 334, 1103
- Adamo A., et al., 2020, Space Sci. Rev., 216, 69
- Aguilar M., et al., 2015, PRL, 114, 171103
- Aharonian F., Yang R., de Oña Wilhelmi E., 2019, Nature Astronomy, 3, 561
- Aharonian F., et al., 2022, arXiv e-prints, p. arXiv:2207.10921
- Albert A., et al., 2021, ApJ, 907, L30
- Amato E., 2014, International Journal of Modern Physics D, 23, 1430013
- Amato E., Blasi P., 2018, Advances in Space Research, 62, 2731
- Amenomori M., et al., 2021, Phys. Rev. Lett., 126, 141101
- Apel W. D., et al., 2013, Phys. Rev. D, 87, 081101

- Astiasarain X., Tibaldo. L., Martin P., Knödlseder J., Remy Q., 2023, arXiv e-prints, p. arXiv:2301.04504
- Baade W., Zwicky F., 1934, Physical Review, 46, 76
- Bakalová A., 2022, in 31st Texas Symposium on Relativistic Astrophysics. p. 8
- Bally J., 2016, ARA&A, 54, 491
- Bartoli B., et al., 2014a, ApJ, 790, 152
- Bartoli B., et al., 2014b, ApJ, 790, 152
- Bartoli B., et al., 2015, ApJ, 806, 20
- Bastian N., Covey K. R., Meyer M. R., 2010a, ARA&A, 48, 339
- Bastian N., Covey K. R., Meyer M. R., 2010b, ARA&A, 48, 339
- Bell A. R., 1978, MNRAS, 182, 147
- Bell A. R., 2004, MNRAS, 353, 550
- Bell A. R., Schure K. M., Reville B., Giacinti G., 2013, MNRAS, 431, 415
- Berezhko E. G., Krymskii G. F., 1988, Soviet Physics Uspekhi, 31, 27
- Berlanas S. R., Wright N. J., Herrero A., Drew J. E., Lennon D. J., 2019, MNRAS, 484, 1838
- Bica E., Bonatto C., Barbuy B., Ortolani S., 2006, A&A, 450, 105
- Binns W. R., et al., 2008, New Astron. Rev., 52, 427
- Bird D. J., et al., 1994, ApJ, 424, 491
- Blandford R., Eichler D., 1987, Phys. Rep., 154, 1
- Blandford R. D., Ostriker J. P., 1978, ApJ, 221, L29
- Blasi P., 2013, A&ARv, 21, 70
- Blumenthal G. R., Gould R. J., 1970, Reviews of Modern Physics, 42, 237
- Bonatto C., Bica E., 2011, MNRAS, 415, 2827
- Brandner W., Clark J. S., Stolte A., Waters R., Negueruela I., Goodwin S. P., 2008, A&A, 478, 137

- Brands S. A., et al., 2022, A&A, 663, A36
- Bronfman L., Cohen R. S., Alvarez H., May J., Thaddeus P., 1988, ApJ, 324, 248
- Bronfman L., Casassus S., May J., Nyman L. Å., 2000, A&A, 358, 521
- Bucciantini N., Amato E., Bandiera R., Blondin J. M., Del Zanna L., 2004, A&A, 423, 253
- Buzzoni A., 2002, AJ, 123, 1188
- Bykov A. M., Toptygin I. N., 2001, Astronomy Letters, 27, 625
- Bykov A. M., Marcowith A., Amato E., Kalyashova M. E., Kruijssen J. M. D., Waxman E., 2020, Space Sci. Rev., 216, 42
- CTA Collaboration 2023, CTA performances, https://www.cta-observatory.org/ science/ctao-performance/
- Cao Z., et al., 2021, Nature, 594, 33
- Cardillo M., Amato E., Blasi P., 2015, Astroparticle Physics, 69, 1
- Carroll B. W., Ostlie D. A., 1996, An Introduction to Modern Astrophysics
- Caselli P., Walmsley C. M., Terzieva R., Herbst E., 1998, ApJ, 499, 234
- Cash W., Charles P., Bowyer S., Walter F., Garmire G., Riegler G., 1980, ApJ, 238, L71
- Casse M., Paul J. A., 1980, ApJ, 237, 236
- Cassinelli J. P., 1979, ARA&A, 17, 275
- Castor J., McCray R., Weaver R., 1975, ApJ, 200, L107
- Cesarsky C. J., Montmerle T., 1983, Space Sci. Rev., 36, 173
- Chen B. Q., et al., 2020, MNRAS, 493, 351
- Cherenkov Telescope Array Consortium et al., 2019, Science with the Cherenkov Telescope Array, doi:10.1142/10986.
- Churchwell E., et al., 2009, PASP, 121, 213
- Comerón F., Pasquali A., 2012, A&A, 543, A101
- Connell J. J., 1998, ApJ, 501, L59
- Cristofari P., 2021, Universe, 7, 324

- Crowther P. A., 2007, ARA&A, 45, 177
- Cummings A. C., et al., 2016, ApJ, 831, 18
- D'Aì A., et al., 2022, Journal of High Energy Astrophysics, 35, 139
- Dalgarno A., 2006, Proceedings of the National Academy of Science, 103, 12269
- Dame T. M., Hartmann D., Thaddeus P., 2001, ApJ, 547, 792
- Demircan O., Kahraman G., 1991, Ap&SS, 181, 313
- Dickel H. R., Wendker H., Bieritz J. H., 1969, A&A, 1, 270
- Digel S. W., 1991, PhD thesis, Harvard University, Massachusetts
- Dobbs C. L., Burkert A., Pringle J. E., 2011, MNRAS, 413, 2935
- Draine B. T., 2011, Physics of the Interstellar and Intergalactic Medium
- Drew J. E., Greimel R., Irwin M. J., Sale S. E., 2008, MNRAS, 386, 1761
- Drury L. O., 1983, Reports on Progress in Physics, 46, 973
- Drury L. O., Downes T. P., 2012, MNRAS, 427, 2308
- Duerr R., Imhoff C. L., Lada C. J., 1982, ApJ, 261, 135
- Eker Z., et al., 2015, AJ, 149, 131
- Eker Z., et al., 2018, MNRAS, 479, 5491
- Elia D., et al., 2022, ApJ, 941, 162
- Elmegreen B. G., Efremov Y. N., Larsen S., 2000, ApJ, 535, 748
- Ernst A., Berczik P., Just A., Noel T., 2015, Astronomische Nachrichten, 336, 577
- Evoli C., 2020, The Cosmic-Ray Energy Spectrum, doi:10.5281/zenodo.4396125, https: //doi.org/10.5281/zenodo.4396125
- Fall S. M., Krumholz M. R., Matzner C. D., 2010, ApJ, 710, L142
- Falle S. A. E. G., 1975, A&A, 43, 323
- Faucher-Giguère C.-A., Kaspi V. M., 2006, ApJ, 643, 332
- Fermi E., 1949, Physical Review, 75, 1169

- Figer D. F., 2004, in Lamers H. J. G. L. M., Smith L. J., Nota A., eds, Astronomical Society of the Pacific Conference Series Vol. 322, The Formation and Evolution of Massive Young Star Clusters. p. 49 (arXiv:astro-ph/0403088), doi:10.48550/arXiv.astroph/0403088
- Figer D. F., Najarro F., Geballe T. R., Blum R. D., Kudritzki R. P., 2005, ApJ, 622, L49
- Gabici S., Montmerle T., 2015, in 34th International Cosmic Ray Conference (ICRC2015). p. 29 (arXiv:1510.02102), doi:10.22323/1.236.0029
- Galli D., Padovani M., 2015, arXiv e-prints, p. arXiv:1502.03380
- Gies D. R., 1987, ApJS, 64, 545
- Ginzburg V. L., Syrovatskii S. I., 1963, The Origin of Cosmic Rays
- Glassgold A. E., Langer W. D., 1974, ApJ, 193, 73
- Gleeson L. J., Axford W. I., 1968, ApJ, 154, 1011
- Grabelsky D. A., Cohen R. S., Bronfman L., Thaddeus P., May J., 1987, ApJ, 315, 122
- Green J. A., et al., 2011, ApJ, 733, 27
- Green G. M., Schlafly E., Zucker C., Speagle J. S., Finkbeiner D., 2019, ApJ, 887, 93
- Greisen K., 1966, Phys. Rev. Lett., 16, 748
- Guelin M., Langer W. D., Snell R. L., Wootten H. A., 1977, ApJ, 217, L165
- Guelin M., Langer W. D., Wilson R. W., 1982, A&A, 107, 107
- Guo F., Li S., Li H., Giacalone J., Jokipii J. R., Li D., 2012, ApJ, 747, 98
- Gupta S., Nath B. B., Sharma P., Eichler D., 2020, MNRAS, 493, 3159
- H. E. S. S. Collaboration et al., 2015, Science, 347, 406
- Hams T., et al., 2004, ApJ, 611, 892
- Hanson M. M., 2003, ApJ, 597, 957
- Heinke C. O., 2010, in Kalogera V., van der Sluys M., eds, American Institute of Physics Conference Series Vol. 1314, International Conference on Binaries: in celebration of Ron Webbink's 65th Birthday. pp 135–142 (arXiv:1101.5356), doi:10.1063/1.3536355

- Helder E. A., Vink J., Bykov A. M., Ohira Y., Raymond J. C., Terrier R., 2012, Space Sci. Rev., 173, 369
- Hezareh T., Houde M., McCoey C., Vastel C., Peng R., 2008, ApJ, 684, 1221
- Hillenbrand L. A., Hartmann L. W., 1998, ApJ, 492, 540
- Höorandel J. R., 2006, in Journal of Physics Conference Series. pp 41–50 (arXiv:astro-ph/0508014), doi:10.1088/1742-6596/47/1/005
- Hou L. G., Han J. L., 2014, A&A, 569, A125
- Hunter S. D., et al., 1997, ApJ, 481, 205
- Jiang Z., et al., 2002, ApJ, 577, 245
- Johnson H. L., Morgan W. W., 1954, ApJ, 119, 344
- Kafexhiu E., Aharonian F., Taylor A. M., Vila G. S., 2014, Physical Review D, 90, 123014
- Kalirai J. S., Richer H. B., 2010, Philosophical Transactions of the Royal Society of London Series A, 368, 755
- Karam J., Sills A., 2022, MNRAS, 513, 6095
- Khangulyan D., Aharonian F. A., Kelner S. R., 2014, ApJ, 783, 100
- Kiminki D. C., Kobulnicky H. A., Vargas Álvarez C. A., Alexander M. J., Lundquist M. J., 2015, ApJ, 811, 85
- Klein O., Nishina T., 1929, Zeitschrift fur Physik, 52, 853
- Klepach E. G., Ptuskin V. S., Zirakashvili V. N., 2000, Astroparticle Physics, 13, 161
- Knödlseder J., 2000, A&A, 360, 539
- Koen C., 2006, MNRAS, 365, 590
- Kolmogorov A., 1941, Akademiia Nauk SSSR Doklady, 30, 301
- Koyama K., Petre R., Gotthelf E. V., Hwang U., Matsuura M., Ozaki M., Holt S. S., 1995, Nature, 378, 255
- Kraichnan R. H., 1965, Physics of Fluids, 8, 1385
- Kramer C., Stutzki J., Rohrig R., Corneliussen U., 1998, A&A, 329, 249

Krause M., Fierlinger K., Diehl R., Burkert A., Voss R., Ziegler U., 2013, A&A, 550, A49

Kroupa P., 2001, MNRAS, 322, 231

Krumholz M. R., McKee C. F., 2005, ApJ, 630, 250

- Krumholz M. R., et al., 2014, in Beuther H., Klessen R. S., Dullemond C. P., Henning T., eds, Protostars and Planets VI. p. 243 (arXiv:1401.2473), doi:10.2458/azu\_uapress\_9780816531240-ch011
- Krumholz M. R., McKee C. F., Bland-Hawthorn J., 2019, ARA&A, 57, 227
- Kudritzki R.-P., Puls J., 2000, ARA&A, 38, 613
- Lada C. J., Lada E. A., 1991, in Janes K., ed., Astronomical Society of the Pacific Conference Series Vol. 13, The Formation and Evolution of Star Clusters. pp 3–22
- Lada C. J., Lada E. A., 2003, ARA&A, 41, 57
- Lamers H. J. G. L. M., Gieles M., 2006, A&A, 455, L17
- Lancaster L., Ostriker E. C., Kim J.-G., Kim C.-G., 2021a, ApJ, 914, 89
- Lancaster L., Ostriker E. C., Kim J.-G., Kim C.-G., 2021b, ApJ, 914, 90
- Larson R. B., 1981, MNRAS, 194, 809
- Lemoine M., 2021, Phys. Rev. D, 104, 063020
- Liu B., Yang R.-z., Chen Z., 2022, MNRAS, 513, 4747
- Lombardi S., Antonelli L. A., Bigongiari C., Cardillo M., Gallozzi S., Green J. G., Lucarelli F., Saturni F. G., 2022, in 37th International Cosmic Ray Conference. p. 884, doi:10.22323/1.395.0884
- Lopez L. A., Krumholz M. R., Bolatto A. D., Prochaska J. X., Ramirez-Ruiz E., Castro D., 2014, ApJ, 795, 121
- Lozinskaya T. A., Pravdikova V. V., Finoguenov A. V., 2002, Astronomy Letters, 28, 223
- Maccarone T. J., Kundu A., Zepf S. E., Rhode K. L., 2007, Nature, 445, 183
- Maeder A., Meynet G., 1993, A&A, 278, 406
- Maschberger T., Clarke C. J., 2008, MNRAS, 391, 711
- Massey P., Thompson A. B., 1991, AJ, 101, 1408

- Mathis J. S., Mezger P. G., Panagia N., 1983, A&A, 128, 212
- McKee C. F., 1989, ApJ, 345, 782
- McKee C. F., Ostriker E. C., 2007, ARA&A, 45, 565
- Miville-Deschênes M.-A., Murray N., Lee E. J., 2017, ApJ, 834, 57
- Mokiem M. R., et al., 2007, A&A, 473, 603
- Morlino G., Gabici S., 2015, MNRAS, 451, L100
- Morlino G., Blasi P., Peretti E., Cristofari P., 2021, MNRAS, 504, 6096
- Nakanishi H., Sofue Y., 2006, PASJ, 58, 847
- Nakanishi H., Sofue Y., 2016, PASJ, 68, 5
- Nieuwenhuijzen H., de Jager C., 1990, A&A, 231, 134
- Oey M. S., Clarke C. J., 2005, ApJ, 620, L43
- Padovani M., Galli D., Glassgold A. E., 2009, A&A, 501, 619
- Parker R. J., Goodwin S. P., 2007, MNRAS, 380, 1271
- Pfalzner S., 2009, A&A, 498, L37
- Pfalzner S., Kirk H., Sills A., Urquhart J. S., Kauffmann J., Kuhn M. A., Bhandare A., Menten K. M., 2016, A&A, 586, A68
- Pflamm-Altenburg J., Kroupa P., 2008, Nature, 455, 641
- Phan V. H. M., Morlino G., Gabici S., 2018, MNRAS, 480, 5167
- Phan V. H. M., Gabici S., Morlino G., Terrier R., Vink J., Krause J., Menu M., 2020, A&A, 635, A40
- Piskunov A. E., Just A., Kharchenko N. V., Berczik P., Scholz R. D., Reffert S., Yen S. X., 2018, A&A, 614, A22
- Pohl M., Englmaier P., Bissantz N., 2008, ApJ, 677, 283
- Portegies Zwart S. F., McMillan S. L. W., Gieles M., 2010, ARA&A, 48, 431
- Potgieter M. S., 2013, Living Reviews in Solar Physics, 10, 3
- Prantzos N., Doom C., Arnould M., de Loore C., 1986, ApJ, 304, 695

- Ransom S. M., 2008, Dynamical Evolution of Dense Stellar Systems, 246, 291
- Reddish V. C., Lawrence L. C., Pratt N. M., 1966, Publications of the Royal Observatory of Edinburgh, 5, 111
- Reid M. J., et al., 2019, ApJ, 885, 131
- Reimer A., Pohl M., Reimer O., 2006, The Astrophysical Journal, 644, 1118
- Renzo M., Ott C. D., Shore S. N., de Mink S. E., 2017, A&A, 603, A118
- Reynolds S. P., 2008, ARA&A, 46, 89
- Rochau B., Brandner W., Stolte A., Gennaro M., Gouliermis D., Da Rio N., Dzyurkevich N., Henning T., 2010, ApJ, 716, L90
- Rogers H., Pittard J. M., 2013, MNRAS, 431, 1337
- Röllig M., et al., 2011a, A&A, 525, A8
- Röllig M., et al., 2011b, A&A, 525, A8
- Roman-Duval J., Jackson J. M., Heyer M., Johnson A., Rathborne J., Shah R., Simon R., 2009, ApJ, 699, 1153
- Rosen A. L., Lopez L. A., Krumholz M. R., Ramirez-Ruiz E., 2014, MNRAS, 442, 2701
- Rybicki G. B., Lightman A. P., 1986, Radiative Processes in Astrophysics
- Rygl K. L. J., et al., 2012, A&A, 539, A79
- Saha L., Domínguez A., Tibaldo L., Marchesi S., Ajello M., Lemoine-Goumard M., López M., 2020, ApJ, 897, 131
- Salaris M., Cassisi S., 2005, Evolution of Stars and Stellar Populations
- Salpeter E. E., 1955, ApJ, 121, 161
- Schneider N., Bontemps S., Simon R., Jakob H., Motte F., Miller M., Kramer C., Stutzki J., 2006, A&A, 458, 855
- Schneider F. R. N., et al., 2018, A&A, 618, A73
- Shapley H., 1918, PASP, 30, 42
- Skilling J., 1975, MNRAS, 172, 557

Smith L. J., Gallagher J. S., 2001, MNRAS, 326, 1027

Southworth J., 2015, in Rucinski S. M., Torres G., Zejda M., eds, Astronomical Society of the Pacific Conference Series Vol. 496, Living Together: Planets, Host Stars and Binaries. p. 164 (arXiv:1411.1219), doi:10.48550/arXiv.1411.1219

Spiering C., 2012, European Physical Journal H, 37, 515

- Spitzer L., 1962, Physics of Fully Ionized Gases
- Spitzer L., 1978, Physical processes in the interstellar medium, doi:10.1002/9783527617722.
- Spitzer Lyman J., Tomasko M. G., 1968, ApJ, 152, 971
- Strong A. W., Moskalenko I. V., Reimer O., 2000, ApJ, 537, 763
- Strong A. W., Moskalenko I. V., Ptuskin V. S., 2007, Annual Review of Nuclear and Particle Science, 57, 285
- Strong A. W., Moskalenko I. V., Porter T. A., Jóhannesson G., Orlando E., Digel S. W., 2009, arXiv e-prints, p. arXiv:0907.0559
- Sun X.-N., Yang R.-Z., Wang X.-Y., 2020a, MNRAS, 494, 3405
- Sun X.-N., Yang R.-Z., Liang Y.-F., Peng F.-K., Zhang H.-M., Wang X.-Y., Aharonian F., 2020b, A&A, 639, A80
- Sun X.-N., Yang R.-Z., Liang E.-W., 2022, A&A, 659, A83
- Takekoshi T., et al., 2019, ApJ, 883, 156
- Taylor A. R., et al., 2003, AJ, 125, 3145
- Thompson T. A., Fabian A. C., Quataert E., Murray N., 2015, MNRAS, 449, 147
- Thornbury A., Drury L. O., 2014, MNRAS, 442, 3010
- Tibaldo L., Gaggero D., Martin P., 2021, Universe, 7, 141
- Torres-Dodgen A. V., Tapia M., Carroll M., 1991, MNRAS, 249, 1
- Uyanıker B., Fürst E., Reich W., Aschenbach B., Wielebinski R., 2001, A&A, 371, 675
- Vaupré S., Hily-Blant P., Ceccarelli C., Dubus G., Gabici S., Montmerle T., 2014, A&A, 568, A50

Vercellone S., et al., 2022, Journal of High Energy Astrophysics, 35, 1

- Verzi V., 2019, in 36th International Cosmic Ray Conference (ICRC2019). p. 450, doi:10.22323/1.358.0450
- Vesperini E., 2010, Philosophical Transactions of the Royal Society of London Series A, 368, 829
- Vieu T., Gabici S., Tatischeff V., Ravikularaman S., 2022, MNRAS, 512, 1275
- Vink J., 2012, A&ARv, 20, 49
- Vink J., 2020, Physics and Evolution of Supernova Remnants, doi:10.1007/978-3-030-55231-2.
- Vink J. S., de Koter A., Lamers H. J. G. L. M., 2000, A&A, 362, 295
- Vishniac E. T., 1983, ApJ, 274, 152
- Weaver R., McCray R., Castor J., Shapiro P., Moore R., 1977, ApJ, 218, 377
- Webber W. R., 1998, ApJ, 506, 329
- Weidner C., Kroupa P., 2004, MNRAS, 348, 187
- Weidner C., Kroupa P., Larsen S. S., 2004, MNRAS, 350, 1503
- Weidner C., Kroupa P., Bonnell I. A. D., 2010, MNRAS, 401, 275
- Wenger T. V., Balser D. S., Anderson L. D., Bania T. M., 2018, ApJ, 856, 52
- Wiedenbeck M. E., Greiner D. E., 1981, Phys. Rev. Lett., 46, 682
- Wilking B. A., Lada C. J., 1983, ApJ, 274, 698
- Williams J. P., McKee C. F., 1997, ApJ, 476, 166
- Williams J. P., Blitz L., McKee C. F., 2000, in Mannings V., Boss A. P., Russell S. S., eds, Protostars and Planets IV. p. 97 (arXiv:astro-ph/9902246)
- Wilson T. L., Rohlfs K., Hüttemeister S., 2009, Tools of Radio Astronomy, doi:10.1007/978-3-540-85122-6.
- Wootten A., Snell R., Glassgold A. E., 1979, ApJ, 234, 876
- Workman R. L., et al., 2022, Progress of Theoretical and Experimental Physics, 2022, 083C01

Wright N. J., Drake J. J., Drew J. E., Vink J. S., 2010, ApJ, 713, 871

Wright N. J., Drew J. E., Mohr-Smith M., 2015, MNRAS, 449, 741

Wright N. J., Goodwin S., Jeffries R. D., Kounkel M., Zari E., 2022, arXiv e-prints, p. arXiv:2203.10007

Xu W. F., Gao X. Y., Han J. L., Liu F. S., 2013, A&A, 559, A81

Yang R., Aharonian F., Evoli C., 2016, Phys. Rev. D, 93, 123007

Yang R.-z., de Oña Wilhelmi E., Aharonian F., 2018, A&A, 611, A77

- Yungelson L. R., van den Heuvel E. P. J., Vink J. S., Portegies Zwart S. F., de Koter A., 2008, A&A, 477, 223
- Zel'Dovich Y. B., Pikel'Ner S. B., 1969, Soviet Journal of Experimental and Theoretical Physics, 29, 170

de Wit W. J., Testi L., Palla F., Zinnecker H., 2005, A&A, 437, 247